



37564	6
ADA.1 2	FINAL REPORT APPLICATIONS OF CLUSTER ANALYSES AND INTEGER PROGRAMMING TO MULTIPLE TARGET TRACKING
	21 March 1983
	DTIC FEB6 USA
UTC FILE COPY	Tracor Applied Sciences Tracor Inc. 6500 Tracor Lane Austin Texas 7872 84 02 6 095.

Š Š, 5 S 1. N. N.

1.1.1.1.4

Contract No. N00014-82-C-0044 Tracor Project 034-010 Document No. T83-AU-9552-U

FINAL REPORT

APPLICATIONS OF CLUSTER ANALYSES AND INTEGER PROGRAMMING TO MULTIPLE TARGET TRACKING

21 March 1983

Submitted to

Scientific Officer Statistics and Probability Program Of'ice of Naval Research 800 North Quincy Street Arlington, Virginia 22217

Attention: Dr. Ed Wegman



Submitted

oper m Don Cooper

6 lean

Glenn Corser

Approved

Haves

Frank C. Harris **Department** Director System Modeling Department

Tracer Applied Sciences

Tracor, Inc. 6500 Tracor Lane Austin, Texas 78721 Telephone 512 926-2800

REPORT DOCUMENTATION PA	GE READ INSTRUCTIONS
1. AEPORY MUNBER 183-AU-9552-U	AI 37564
• TITLE (and Addition) Final Report: Applications of Clust	er Analyses
and Integer Programming to Multiple Tracking	Target I April 1982 - 31 Mar PERFORMING ONG. REPORT NUMB T83-AU-9552-U
7 AUTHOR(+)	8. CONTRACT OR GRANT NUMBER
Don Cooper, Glenn Corser, Frank Harr	is N00014-82-C-0044
PERFORMING ORGANIZATION NAME AND ADDRESS	10, PROGRAM ELEMENT, PROJECT, " AREA & WORK UNIT NUMBERS
Applied Sciences Group 6500 Tracor Lane Austin, TX 7	65152N, RO145-TW, NR 274-299
1. CONTROLLING OFFICE NAME AND ADDRESS	12. REPORT DATE
Office of Naval Research	21 March 1983
800 North Quincy St. Arlington,	VA 22217 153
14. MONITORING AGENCY NAME & ADDRESSOI different fra	m Controlling Office) 18. SECURITY CLASS. (of this report)
N/A	Unclassified
	ISA DECLASSIFICATION DOWNGRAD
Approved for Public Release: Distri 7. DISTRIBUTION STATEMENT (of the about an interest of the But in But i	bution Unlimited
Approved for Public Release: Distri 7. Distribution Statement (of the abovest encored in Bi N/A	bution Unlimited
Approved for Public Release: Distri 7. Distribution Statement (of the electric entered in Bi N/A	bution Unlimited
Approved for Public Release: Distri Approved for Public Release: Distri 7. Distribution Statement (of the eboweet envirod in Bi N/A N/A N/A	bution Unlimited
Approved for Public Release: Distri Approved for Public Release: Distri 7. DISTRIBUTION STATEMENT (of the abovest entered in Bi N/A N/A	bution Unlimited
Approved for Public Release: Distri Approved for Public Release: Distri Distribution Statement (of the electronic entered in Bind N/A Supplementary works N/A Key works (Continue on reverse aldo if necessary and id Target Tracking Clust	bution Unlimited
Approved for Public Release: Distri Approved for Public Release: Distri 7. DISTRIBUTION STATEMENT (of the electronic entered in Binned N/A	heck 20, if different from Report)
Approved for Public Release: Distri Approved for Public Release: Distri 7. DISTRIBUTION STATEMENT (of the ebolyant envirod in Binnet N/A	bution Unlimited
Approved for Public Release: Distri Approved for Public Release: Distri 7. DISTRIBUTION STATEMENT (of the aboves) entered in Bi N/A • SUPPLEMENTARY WOTES N/A • XEY TOROS (Continue on reverse olds if necessary and id Target Tracking Clust Localization Integ Multi-Target Tracking Automatic Data Sorting • ABSTRACT (Continue on reverse olds if necessary and ide See attachment	bution Unlimited
Approved for Public Release: Distri Approved for Public Release: Distri 7. DISTRIBUTION STATEMENT (of the electronic entered in Bi N/A • SUPPLEMENTARY WOTES N/A • KEY WORKS (Continue on reverse olds if necessary and the Target Tracking Clust Localization Integ Multi-Target Tracking Automatic Data Sorting • AGSTRACT (Continue on reverse olds if necessary and the See attachment	bution Unlimited
Approved for Public Release: Distri Approved for Public Release: Distri 7. DISTRIBUTION STATEMENT (of the abeliast entered in B N/A • SUPPLEMENTARY WOTES N/A • KEY WORDS (Continue on reverse olds if necessary and la Target Tracking Clust Localization Integ Multi-Target Tracking Automatic Data Sorting • ABSTRACT (Continue on reverse olds if necessary and la See attachment	bution Unlimited

6.2.2

20. ABSTRACT

A Multiple-Target Tracking Algorithm has been designed and tested for use with DIFAR sonobuoys. The algorithm is implemented as a functionally modular computer program composed of five main subroutines. Operating in near real time and in less than 64K words of memory, the algorithm is capable of separating raw measurements into data sets corresponding to individual targets at the sensor level (using cluster analysis), correlating target measurements across sensors (using physical constraint and statistical tests), and selecting the most likely track scenario among the various potential scenarios (using integer programming). The process requires no operator decision and no a priori information about the number or initial conditions of the targets. The algorithm is self initializing from the raw buoy data. Using synthetic data, the algorithm was tested on several multitarget scenarios with excellent results.

Situation tinding and ranges)

3

Ŗ

STATES STATES

TABLE OF CONTENTS

Section

	EXECUTIVE SUMMARY	1
1.0	INTRODUCTION	6
1.1	Summary of Past Work	6
1.2	Results from Current Investigation	9
1.2.1	Selection of Clustering Algorithm	10
1.2.2	Decision Rules for Automatic Cluster	12
	Extraction	
1.2.3	Initial Guess Procedure	13
1.2.4	Physical and Statistical Constraints on	14
	the Initial Guess Estimates	
1.2.5	Integer Programming and Optimal Track	16
	Scenario	
1.2.6	Scenario Simulation Results	16
1.3	Report Organization	17
2.0	DATA SORTING BY CLUSTER ANALYSIS	19
2.1	Description of Multi-Target Scenario Data	19
2.2	Moody and Jardine's B _k Overlapping	20
	Single Linkage Clustering Algorithm	
2.3	Selection of Ling's (1,r) Clustering	22
	Algorithm	
2.4	Data Normalization	24
3.0	AUTOMATIC CLUSTER EXTRACTION	27
3.1	Single Linkage Clustering and Cluster Formations	27
3.2	The Isolation Index	30
3.3	Isolation Index Distribution - Survival Function	31
3.4	Polarization Test and Cluster Extraction	32
3.5	Internal Consistency and Cluster	40
	Extraction	

÷.

TABLE OF CONTENTS (CONTINUED)

Section

Page

3.5.1	Regression and Internal Consistency	42
3.5.2	Weighting Scheme	47
3.6	Automatic Cluster Extraction Summary	48
4.0	INITIAL GUESS PROCEDURE	50
4.1	Crossed-Bearing, Crossed-Frequency Initial	50
	Guess Procedure	
4.2	Measurement Weighting Equations	53
5.0	PHYSICAL AND STATISTICAL CONSTRAINTS ON	56
	THE INITIAL GUESS ESTIMATES	
5.1	Physical Constraints Test	56
5.2	Two Sensor Statistical Compatability Test	59
5.3	Three Sensor Statistical Compatibility	63
	Test	
6.0	INTEGER PROGRAMMING AND SCENARIO SELECTION	70
6.1	Set Partitioning Formulation	71
6.2	Applying Set Partitioning to the Scenario	73
	Selection - Constraints	
6.3	Applying Set Partitioning to Scenario	75
	Selection - Cost Function	
6.4	Integer Programming Summary	78
7.0	RESULTS	79
7.1	Introduction	79
7.2	Scenario Data Generation	80
7.2.1	Sensor Locations	80
7.2.2	Target Trajectories	81
7.2.3	Simulated DIFAR Multi-Target Data	81
7.3	Three Target Scenario, Good Geometry,	86
	Strong Signal	
7.3.1	Introduction	86
7.3.2	Data Clustering Results - Buoys 1 and 3	87

TABLE OF CONTENTS (CONCLUDED)

Section

addarada dagaaraa kaala

S.

355

X

3

RXA

Page

7.3.3	Cluster Extraction Results Buoy 3 and	111
	Buoy 1	
7.3.4	Initial Guess Results	111
7.3.5	Integer Programming Results	117
7.3.6	Conclusion	117
7.4	Three Target Scenario with Bad Geometry	120
	Low Noise, 0 dB threshold in 1 Hz Band	
7.5	Three Target Scenario with Good Geometry,	134
	Moderate Noise, No Threshold	
7.6	Two Target Scenario with Good Geometry, Low	138
	Noise, and O dB in a 1 Hz Band Threshold	
8.0	SUMMARY AND RECOMMENDATIONS	142
8.1	Project Summary	142
8.2	Recommendations for Future Work	142
9.0	REFERENCES	147

LIST OF FIGURES

Figure

4

1.1	Top-Down Flowchart of Tracor's Multi-Target	11
	Tracking Algorithm (MTTA)	
2.1	Example of Two-overlapping Clusters	20
3.1	Clustering Example	29
3.2	Bearing/Frequency Plot Examples	34
3.3	Plots to Determine Polarization Test Cut Off Point	39
3.4	Flowchart of the Internal Consistency Test	41
5.1	Plot of the Line-of-Sight Bearing Inter-	57
	sections Resulting from the Crossed-	
	Bearings Procedure for Scenario 1	
5.2	Initial Guess Procedure's 3-Sensor Combination Tests	65
7.1	Plot of the MTTA's Estimated Tracks vs.	88
	the True Tracks for the 3-Target,	
	Strong Signal and Good Geometry Scenario	
7.2	Integer Programming Equation for Scenario 1	118
7.3	Plot of the MTTA's Estimated Tracks vs.	121
	the True Tracks for the 3-Target Strong	
	Signal, and Poor Geometry Scenario	
7.4	Plot of the MTTA's Estimated Tracks vs.	135
	the True Tracks for the 3-Target, Weak	
	Signal and Good Geometry Scenario	
7.5	Plot of the MTTA's Estimated Tracks vs.	139
	the True Tracks for the 2-Target,	
	Strong Signal and Good Geometry Scenario	
8.1	Logical Flowchart of Tracor's Proposed	145
	Final MTTA	

2

6

ст. С

Same and

C. M. S. S. S.

LIST OF TABLES

TABLE		Page
7.1	DESCRIPTION OF THE 3 TARGET TRACKS FOR SCENARIO 1	82
7.11	DESCRIPTION OF THE 3 TARGET TRACKS FOR SCENARIO 2	83
7.III	DESCRIPTION OF THE 2 TARGET TRACK FOR SCENARIO 4	84
7.IV	SIMULATED MULTI-TARGET DIFAR DATA FOR BUOY 1 OF SCENARIO 1	89
	SIMULATED MULTI-TARGET DIFAR DATA FOR BUOY 3 OF SCENARIO 1	91
7.VI	CLUSTERING OUTPUT FOR BOUY 1 OF SCENARIO 1	94
7.VII	CLUSTERING OUTPUT FOR BUOY 3 OF SCENARIO 1	98
7.VIII	SURVIVAL FUNCTION AND ISOLATION INDEX VALUES FOR CLUSTERS FROM BUOY 1 OF SCENARIO 1	103
7. IX	SURVIVAL FUNCTION AND ISOLATION INDEX VALUES FOR CLUSTERS FROM BUOY 3 OF SCENARIO 1	107
7 .X	BUOY, CLUSTER, AND TARGET NUMBER CORRE- SPONDENCE TABLE	112
7.XI	INITIAL GUESS AND CONSTRAINTS RESULTS FOR SCENARIO 1	114
7.XII	SIMULATED MULTI-TARGET DIFAR DATA FOR BUOY 1 OF SCENARIO 2	122
7 .XIII	CLUSTERING OUTPUT FOR BUOY 1 OF SCENARIO 1	124
7.XIV	SIMULATED MULTI-TARGET DIFAR DATA FOR BUOY 3 OF SCENARIO 2	128
7 . XV	CLUSTERING OUTPUT FOR BUOY 3 OF SCENARIO 2	130

EXECUTIVE SUMMARY

During the past year Tracor created and tested a Multiple-Target Tracking Algorithm (MTTA) designed for use with DIFAR sonobuoys. Operating in near real-time and in less than 64 K words of memory, the algorithm is capable of separating raw measurements into data sets corresponding to individual targets at the sensor level, correlating target measurements across sensors, and selecting the most likely track scenario among the various potential scenarios. The process requires no operator decision and no a priori information about the number or initial conditions of the targets. The algorithm is self initializing from the raw buoy data. Using synthetic data, the algorithm was tested on several multi-target scenarios with excellent results. This executive summary presents an overview of work accomplished on the MTTA during the past year. Included is a brief description of the algorithm's structure, conclusions of the study, and recommendations for further work.

The MTTA is implemented as a functionally modular computer program composed of five main subroutines. At present all processing is serial; however, with little effort the algorithm could be implemented in a parallel processing scheme.

Sorting Data at Sensor Level-Cluster Analysis.

When plots of frequency versus bearing measurements from a DIFAR sensor are constructed, the noise free data lie along either straight or curvilinear lines. For targets that are moderately separated in bearing or frequency, the associated lines will be quite distinct. Based on these observations, experiments were conducted to test the usefulness of cluster analytic techniques to separate target data at the sensor level. They proved to be successful and Ling's (1,r) algorithm

(equivalent to a single-linkage clustering algorithm) was used to separate data at the sensor level into potential target data sets.

and substant and all all and a land a land a land a land a land a land

Automatic Cluster Extraction. While cluster analysis worked well in separating observations into proper target data sets, the structure of the algorithm made it difficult to distinguish these sets from the others created by the algorithm. This is a standard and much discussed problem in cluster analysis. Tracor was able to develop a three stage attack on the problem which has proven to be quite successful. The first step is to use Ling's isolation index to select clusters which are well isolated from the rest of the data. Next, the clusters are checked for randomness and only those that are distinctly nonrandom are selected. Lastly, a qualitative regression procedure is used to check clusters with subsets for internal consistency. Only those clusters which are isolated, nonrandom, and internally consistent are selected as target data sets.

Intersensor Correlation-Initial Guess Procedure

and Constraint Rules. To correlate data from more than one sensor, a crossed-bearing, crossed-frequency, batch type algorithm is used to provide initial state vector estimates from the data contained in the various two-sensor and three-sensor cluster combinations. These estimates are then subjected to a series of physical and statistical based tests to determine their validity. The physical tests include checks on reasonableness of position and velocity estimates. The statistical tests use qualitative regression procedures to test the two sensor estimates for compatability, and Gallant's non-linear regression test to compare a three sensor estimate

4

and its associated two sensor estimates for consistency. Intersensor cluster combinations which pass all tests are then considered as potential target tracks.

Scenario Construction - Integer Programming. Once the potential target track set has been delineated, all that remains is to construct the set of tracks which provide the most likely scenario description. To solve this problem, Tracor has used a 0-1 integer programming set partitioning procedure to sift through the large number of potential scenarios to select the most likely one. To devise a cost function, each track is fed into Tracor's hybrid tracking algorithm and the value of the likelihood function for the associated measurement model residual stream is computed. The likelihood function for a scenario is based on the likelihood function associated with each track, and finding the optimal scenario is equivalent to maximizing the likelihood function over the set of all possible scenarios.

The MTTA was tested on four scenarios of varying difficulty; three of the scenarios contained three targets while one of the scenarios contained only two targets. For three of the scenarios, all targets were detected, sorted, and good track estimates were generated. In the remaining scenario, two of the three targets were detected and tracked, but the third target was only partially detected and no track was generated. The following conclusions were drawn from this study:

> Observation geometry was the single most important variable for successful detection and tracking of the targets by the MTTA,

- Strong signal-to-noise ratio is important, but not crucial to the successful operation of the MTTA, and
- 3) Execution time for the MTTA varies from scenario to scenario based on the complexity of the trajectories. All scenarios were of six minute duration, and the MTTA took from seven to nine minutes to run through them. Minimal parallel processing could turn the MTTA into a faster than real time batch processing scheme.

The recommendations for future work fall into two categories -- refinement of clustering and extraction procedures and MTTA extensions. Under the first category, the following topics were identified as algorithm deficiencies that need to be investigated:

- Examination of alternative clustering approaches (CLASSY, Anderberg's overlapping algorithm).
- 2) Using quadratic terms in the cluster extraction regression to model CPA.
- 3) Addition of a clustering attribute based on power or SNR to help sort targets.

In the second category, all multi-target schemes must deal with the question of whether a measurement belongs to currently existing tracks or is it part of a track that has yet to be initialized. Tracor's proposed implementation of a complete

like seconds. Beside second success and an and a success the second success.

-

NY.

2

• •

ļ

3

THE REAL PROPERTY

multi-target tracking algorithm would address this question as soon as a data point was acquired. If it was determined that the data point was not part of a current track, then it would be put into a data pool and at designated times the MTTA would be activated to search for and initialize new tracks. The extension to the MTTA would consist of the decision rules required to ask whether a point belongs to a track or not and then the procedures required to update a track. An additional task would be to investigate whether smoothing of raw data would help both in tracking and updating.

In summary. Tracor has developed a self initializing multi-target tracking algorithm that has performed well with simulated DIFAR data. It runs in nearly real time and requires less than 64 K words of memory. Although in this study the MTTA used Tracor's Hybrid Tracking Algorithm, any single target tracking algorithm could be used to generate scores for each potential track. What has been created is a superstructure which sorts and makes decisions independently of the particular target tracker used. Planned refinements and extensions to the algorithm would give it the ability to run for long periods of time providing a real-time picture of the current tracking scenario. It is felt that the MTTA represents a promising step towards a fully automated DIFAR based, multi-target tracking algorithm.

1.0

N

Ŋ

Ŋ

N.

INTRODUCTION

For many years, the U.S. Navy has been concerned with the problems of detecting, classifying, and tracking submerged targets encountered in anti-submarine warfare (ASW) engagements. One of the problems which has been extensively investigated at Tracor is underwater target tracking with data gathered by deployed, passive sonobuoy detection systems. Past efforts have concentrated on the problem of tracking a single target. In recent years, however, greater emphasis has been placed on developing techniques which can use passive data to track multiple targets simultaneously. no a priori When information is available concerning numbers target or trajectories, the multiple target tracking problem becomes very difficult to solve. This report contains the findings from the past year of Tracor's efforts to address the problem of tracking multiple targets with passive data when no a priori information is available. Under the current contract a multi-target tracking algorithm was developed which performed very well on simulated multi-target DIFAR data. The algorithm was able to properly sort data at the sensor level, correlate the and reconstruct tracks ACTOSS sensors under data consideration. It is felt that a positive step has been taken in creating an effective DIFAR based multi-target tracking algorithm.

1.1 Summary of Past Work

In the past, Tracor has studied both the single and the multiple target tracking problems. Initial emphasis was placed on developing a quick and accurate single target tracking algorithm. This work led first to Tracor's development of a batch-type tracker known as the Maximum Likelihood Estimator (MLE, Reference 1). The MLE was designed

to initialize tracking solutions when no <u>a priori</u> information was provided and to continue tracking a single target until all the data have been processed. However, due to the batch nature of the tracker, the MLE was forced to iterate through four different motion models to account for possible target maneuvers and to use sophisticated statistical techniques to automatically choose the most appropriate motion model. Unfortunately, this process was found to be both cumbersome and time consuming.

In an effort to develop a quicker and more efficient tracker, Tracor then developed an Extended Kalman Filter (EKF, References 2 and 3) for single target tracking applications. Since the EKF was a sequential tracker, a single, less sophisticated motion model was used with process noise incorporated into the tracker to compensate for any unmodeled trajectory changes. This single motion model substantially reduced the complexity of the tracker and helped to greatly speed up its execution time. Furthermore. if reasonably good a priori information was available for the target's state, the EKF was not only much quicker, but it was at least as accurate as the MLE. Unfortunately, when no a priori state information was available, the EKF frequently failed to converge upon an adequate track of the target's trajectory, and thus could not be counted on to track targets under all conditions.

Since a quick but accurate tracker was desired for single target-tracking applications, Tracor also developed a tracker which utilized the best characteristics of both the MLE and the EKF. This led to the development of Tracor's Hybrid Tracking Algorithm (HTA, Reference 2 and 3). The HTA utilized a batch type starter to initialize the tracker and then, after initialization had been successfully completed,

7

Ē

2

Ś

1

Ş

automatically switched to a sequential tracker to continue updating the target's trajectory. This tracker was found to provide the accuracy needed to initialize a trajectory when no <u>a priori</u> information was available, and to track as quickly and as accurately as the EKF after initialization was completed. The HTA was then chosen as Tracor's final product for single target tracking applications.

After completing the development of the HTA as a single target tracker, Tracor became involved in the development of a multiple target tracking algorithm (MTTA). This algorithm was to be developed to handle the very difficult problem of tracking multiple targets when no a priori available concerning information was target numbers or trajectories. Furthermore, only passive frequency and bearing data (the data types commonly available from deployed sonobuoys) were to be used to track the targets. In the past, several authors have developed multiple target trackers which used a priori information or active data types such as range measurements to initialize the tracker, but very few have examined the more common and much more difficult problem of tracking multiple targets with only passive data and no a priori information.

Last year, (Reference 3) the development of the MTTA was broken up into two phases. The first phase was designed to enhance the HTA's performance so it could be readily modified for further use in the MTTA. As envisioned, the HTA would serve as the core tracking algorithm which would be used to track the various targets present in a given scenario.

Once the improvements to the HTA were completed. an attack on the second, more difficult, problem of data sorting was begun. For the multiple target tracking problem, some means was needed to sort the data received from individual sensors into sets of individual target data. Without any a priori information concerning the targets that were present, none of the traditional "gating" or "nearest neighborhood" approaches could be used to sort the data. After studying frequency versus bearing time histories of the data, it was decided to try to use cluster analysis to sort the multiple target data into single target data sets. After several trials, single-linkage cluster analysis was found to be fairly successful in sorting the data when the time tag, estimated frequency, and the cosine and sine of the bearing estimates were used as object attributes. However, the results of the cluster analysis program were output as tree diagrams or dendrograms which required user interpretation to correctly sort the data. For the cases studied, all the multiple target data were generated by user dictated simulations, so it was known what type of behavior to look for in the data. With this advantage, it was possible to interpret the tree diagram outputs such that individual target data sets could be readily found. Without the benefit of this knowledge, it is doubtful that the tree diagrams could have been used to find the correct clusters of data. From these results it was decided that cluster analysis showed good potential for sorting multiple target data into individual target data sets, but a suitable cluster extraction scheme had to be developed.

1.2

Results from Current Investigation

Tracor has continued its multiple target tracking investigation through the current year. During this time,

3 .1

2

5

3

ž

ļ

Tracor's MTTA has evolved into a large scale program containing five separate modules. A top-down flowchart of these modules, and a brief description of each one is given in Figure 1.1. The following sections summarize the work that has been performed this year in each of the module areas.

1.2.1 Selection of Clustering Algorithm

During this year's investigation three different clustering algorithms were examined: the single-linkage clustering algorithm from CLUSTAR, Ling's generalized (k,r) Moody Jardine's algorithm, and and single-linkage, non-hierarchical overlapping B algorithm. Euclidean distances were used in all algorithms to generate the necessary resemblance matrix of dissimilarity coefficients. The overlapping algorithm allows different clusters to contain one, two or several points in common while the clusters maintain their own separate identity. On the other hand, the non-overlapping algorithm requires that the data points that created the overlap in the B_{μ} algorithm be in one cluster or the other or the non-overlapping technique will force the two clusters to merge into one cluster at these points.

Initially, it had been felt that the B_k overlapping algorithm might provide some flexibility that would be useful for solving the data sorting problem. However, in the cases studied, the non-overlapping techniques have been found to be adequate for sorting the data. Furthermore, both the non-overlapping algorithms were found to be substantially faster than the overlapping algorithm. It was therefore decided to remain with the single-linkage, non-overlapping



.

े -

2

ल

Figure 1.1. Top-Down Flowchart of Tracor's Multi-Target Tracking Algorithm (MTTA)

33

3

ANN NUMBER

an interes | the states in

- 534

approach that had been used previously. Of the two remaining algorithms that were under consideration, it was decided to use Ling's (1,r) algorithm rather than his generalized (k,r) algorithm or CLUSTAR's single linkage algorithm because:

> With k=1, Ling's (k,r) algorithm was equivalent to a single-linkage algorithm and produced the same results as CLUSTAR's algorithm,

- 2) Its output was more accessible and easier to use than the tree diagrams from CLUSTAR, and
- 3) Most importantly, Ling has developed a quantitative measure for the (1,r) algorithm that was found to be useful in implementing a procedure for automatic cluster extraction.

1.2.2 Decision Rules for Automatic Cluster Extraction

One of the problems seen last year with the single-linkage clustering algorithm was that it continued to link or chain all the data and the clusters together until ultimately, all the data were chained together in one large, all-inclusive cluster. Naturally, the sorted data for the individual targets were usually found in smaller, subclusters contained as subsets of the large final cluster. In last year's report, available knowledge concerning the simulated scenarios was used to aid in picking the correct smaller clusters from the tree diagram outputs. This year it was sought to automate this cluster extraction process. In doing

1.2.46.66

so three different decision rules have been established to automatically extract the appropriate clusters. The three rules or tests adopted include:

- Ling's isolation index is used to identify potentially useful clusters. It identifies clusters which are well isolated from other clusters, implying that the data in the isolated cluster have little similarity with the data in other clusters,
- 2) A polarization test for data randomness based on work done by Alam and Mitra (Reference 4) which seeks to detect and eliminate noisy clusters, i.e. those clusters containing a high percentage of points corresponding to background noise and not signal measurements, and
- 3) Lastly, a regression based test is used to make the final identifications of those isolated clusters corresponding to likely target data sets.

1.2.3 Initial Guess Procedure

After the multiple target data from individual sensors have been sorted by cluster analysis into individual target data sets, these data clusters must be properly matched with clusters from other sensors to construct tracks for the individual targets. With only passive frequency and bearing measurements, it is usually impossible to track a target with data from just one sensor. Typically, data from two or three sensors with other observation geometries must be used to track

などのない。

Management - -----

R

R

a given target. For the multi-target case, several clusters from each of the sonobuoys will be available to try to track a Ultimately, one must find the right combination of target. clusters from the different sensors that is needed to track each of the individual targets. After the individual clusters from each sensor have been established, the next step is to get a rough idea of which combination of clusters can potentially be used to build tracks. For this purpose, a crossed-bearing, crossed-frequency least squares procedure has been employed to generate an estimate of the initial target state for each possible combination of matched, intersensor clusters. This procedure has been used previously (Reference 3) to generate a starting point for the HTA's initializer for use in tracking single targets. It has been found that with some slight modifications, this same crossed-bearing, crossed-frequency procedure may be used to provide a cursory look at how well the intersensor combination of data clusters go together to provide estimates of potential initial conditions for the targets. As will be described next, the rough estimates produced here can be examined to see if they meet certain physical and statistical constraints. For those estimates that fail the constraints tests, their corresponding intersensor from combination be eliminated cluster can further consideration. This then helps to reduce the scope of the problem under consideration before it passes on to the next MTTA module.

1.2.4 <u>Physical and Statistical Constraints on the</u> Initial Guess Estimates

As stated, the initial guess estimates must meet certain constraints before they are passed for further consideration by Tracor's MTTA. The first test is to see if

the position estimate falls within the physical limits of the sonobuoy's observation range and if the target's velocity estimate falls within the range of allowable values for target trajectories. For instance, if the range for the day of a given sonobuoy is set at 10,000 meters, one could safely reject any initial guess estimates that would place a target 30,000 meters from the sonobuoy. Similarly, if the initial guess of target velocity falls well outside the known range of values that are possible for target trajectories, one could safely eliminate the intersensor cluster combination that yielded that For the physical constraints routine used estimate. in Tracor's MTTA, the user inputs the maximum sonobuoy detection range and the maximum target speed, and the program rejects all combinations whose estimates are greater than 1.5 times the allowable maximum values. This allows a 50% error to be present in initial guess estimates while still accepting estimates from the upper range of allowed values. The initial guess procedure has been found to be more accurate than this 50% error tolerance, but this tolerance permits a safety cushion which prevents possible intersensor cluster combinations from being rejected prematurely. The second set of constraints applied to the initial guess estimator consists of two statistical tests that measure the consistency of the estimates produced by two or three sensors. The first test uses indicator variables (References 5 and 6) to determine if both sensors in a two sensor initial guess estimate are providing statistically compatible information. Once **a**11 two-sensor combinations have been examined, those three-sensor combinations which contain at least two "good" two-sensor combinations are tested. A nonlinear regression test developed by Gallant (Reference 7) is applied to the data to determine if the three-sensor initial guess estimate is equivalent to the

ù

3

とうないない

AN ACCUS

ないいない

various possible two-sensor estimates. At least two of the three possible two-sensor estimates must be equivalent to the three-sensor estimate for the three-sensor inital guess estimate to pass the test.

1.2.5 Integer Programming and Optimal Track Scenario

Once the initial guess and constraints procedures have selected a set of potential target data sets P, the only remaining problem is to determine the subset of P that provides the "optimum" track scenario. In the MTTA, the optimum scenario is the one which maximize the value of the likelihood function over the set of potential scenarios. Clearly, this is a combinatorial problem that, even for moderate size problems, possesses a large number of possible solutions. In a fashion analagous to Morefield (Reference 8), Tracor has chosen to pose this problem as a 0-1 set partitioning problem using integer programming techniques to sift through the various possible track scenarios to find the optimal set of tracks.

1.2.6 Scenario Simulation Results

With exception, Tracor's MTTA one has successfully tracked all the targets in the multi-target scenarios studied in this investigation. Efforts to automate the cluster extraction process have been largely successful, and the initial guess and integer programming procedures have worked quite well at finding the right set of trajectories needed to track the multiple targets. Four different simulated scenarios have been used in this investigation, and the MTTA successfully tracked all of the targets in three of the four scenarios. For the one scenario that failed, three targets were present but the MTTA only tracked two of them. Careful

examination of the results indicated that for this particular scenario poor observation geometry caused the clustering algorithm to become confused and to merge the data from two targets into one cluster. This confusion was caused when the observational geometries for two different sonobuoys forced two different targets to have nearly identical frequency and bearing measurements during the same time period, and the non-overlapping clustering algorithm could not sort the data into individual target sets. In spite of this problem, the MTTA was still able to track two of the three targets. One of the other scenarios tested the MTTA's capability to track multiple targets with data gathered from a weak signal source that caused random noise to be included in the measurement set. For this scenario, the MTTA effectively sorted the true signals from the noise and was then able to track all the targets found in the scenario. The other two scenarios presented good geometries with strong signals, so the MTTA was able to track all the targets very well. The results from these four scenarios are felt to offer representative examples of how well the MTTA can work in multi-target scenarios when no a priori information is available to initialize the tracker.

1.3 Report Organization

12222000

Section 2 of this report includes detailed descriptions of the cluster analysis work performed under this contract. The next section describes the techniques that have been used to automate the cluster extraction process needed to sort multiple target data. In the fourth section, the initial guess procedure is described in detail. Section 5 describes the physical and statistical constraints that were used to eliminate unlikely intersensor cluster combinations. The integer programming procedure used by Tracor's MTTA is

described in the next section, which also includes details of the optimization constraints and the HTA cost function that were used by the optimization algorithm. Section 7 contains a summary of the results obtained for the various simulated scenarios that were examined during this investigation. Finally, Section 8 contains conclusions from this year's research as well as recommendations for future investigations.

2.0

DATA SORTING BY CLUSTER ANALYSIS

Cluster analysis is a field in numerical taxonomy that uses attributes which describe a set of objects to group or sort those objects according to the degree of similarity between them. For the data sorting problem, measurements from an individual target should exhibit strong similarities in their frequency and bearing estimates whereas measurements from different targets should exhibit strong dissimilarities between them. Last year's research established that single-linkage cluster analysis showed good potential for sorting multiple target data into single target data sets (Reference 3). It was shown that frequency versus bearing time histories of the measurements produced a rough, chain-like curve for each target when the data were plotted in three dimensions. Because of the chain-like behavior found in the data. single-linkage clustering algorithms were chosen because they are known to chain data together in forming clusters. Last years results also showed that Euclidean distances between object attributes were the most useful for generating the resemblance matrix of dissimilarity coefficients needed to cluster the data. This section contains the results of current studies which sought to find ways to improve the usefulness of the single-linkage clustering concept.

2.1 Description of Multi-Target Scenario Data

Before evaluating any clustering algorithm's performance, some multi-target data had to be obtained to perform the data sorting tests. For this study, simulated multiple target data were created because no real data were readily available. Frequency and bearing measurements for each target were output at fixed time intervals, and these data were then merged by time tags to simulate a multiple target data set. This procedure was performed for each of the three sonobuoys found in a scenario, so that at the conclusion of the simulation, a separate multi-target data set was generated for each sonobuoy. The method used to generate this multiple target data was identical to the method used in last year's report to generate the multiple-linetracker data (Reference 3).

2.2 <u>Moody and Jardine's B_k Overlapping Single</u> Linkage Clustering Algorithm

Three cluster analysis algorithms were evaluated during the current investigation. The first, Moody and Jardine's B_k algorithm (Reference 9), is classified as an overlapping single-linkage algorithm. Overlapping means that two clusters share some points by overlapping their boundaries at these points, but they also maintain separate boundaries away from these points. An illustration of two overlapping clusters is presented in Figure 2.1.



Figure 2.1 - Example of two-overlapping clusters.

Ì

•

Ň

Ç,

Sec. Sec. So

24202355

1111111111 A606465

Note how the boundaries for clusters 1 and 2 overlap at the apex of the data points in Figure 2.1; they are still classified as separate clusters, but share five points in common. For non-overlapping techniques, the five ambiguous data points in question may have been in either cluster 1 or cluster 2, they may have been grouped in a cluster by themselves, or they may have served as a link to join all the data into one new cluster. However, the points in question could not be contained in two separate clusters as was illustrated in Figure 2.1.

It was originally felt when the B_{L} algorithm was programmed that such an overlapping scheme might be useful for cases when a non-overlapping clustering algorithm could not clearly differentiate the data from two different targets. Certain geometries are possible where a given sensor may not be able to clearly spot the differences between frequency and bearing estimates for two different targets, and it was feared that a non-overlapping algorithm would chain the two different data sets into one cluster at the point of intersection between the two data sets. Clearly if cluster analysis was to sort multiple target data into clusters for each separate target, one would not like the two data sets to be combined into one cluster and then try to use this cluster as data from a single target. It was hoped that this non-overlapping technique would be useful for preventing these ambiguities from confusing the cluster analysis program. However, after closely examining the performance of the B_k algorithm, it was decided to continue usage of non-overlapping, single linkage clustering algorithms for the following reasons:

- The B_k algorithm was found to be much slower computationally than the nonoverlapping algorithms,
- 2) It was felt that if sufficient automatic cluster extraction rules could be developed for a non-overlapping algorithm, then data from the two clusters could be sorted before the data sets merged together. This would prevent the need for an overlapping algorithm, and
- 3) Finally, it was felt that automatic cluster extraction rules could be perfected for at least one of the clustering algorithms. Since Ling had already looked at some aspects of this problem for his (l,r) algorithm, it was decided to expand upon this work and concentrate on using the (l,r) algorithm to automate the cluster extraction process.

2.3 <u>Selection of Ling's (1,r) Clustering Algorithm</u>

After eliminating Moody and Jardine's B_k overlapping clustering algorithm from further consideration, two non-overlapping, single linkage clustering algorithms were left. One of the candidate algorithms was the CLUSTAR single linkage algorithm (Reference 10) that was used to produce the clustering tree diagrams found in last year's report. This algorithm was found to perform well, but the dendrogram outputs were not considered to be very useful for automatic cluster extractions. Having had the advantage of knowing in advance

1

. .

> , ,

3

2. 2. 2. what data clusters to look for, it was possible to find the desired clusters in the tree diagram outputs. Without this <u>a</u> <u>priori</u> knowledge, it was doubtful that the correct data clusters could have been correctly picked from the output. Clearly, either CLUSTAR's output had to be modified to make it easier to obtain the clustered points, or another algorithm had to be found which performed as well but whose output could be easier to use in the cluster extraction process.

candidate The second clustering algorithm available was Ling's (k,r) algorithm (References 11 and 12). This algorithm requires at least k points to be within some distance r in similarity before the data points can be grouped together into a cluster. For k = 1, Ling points out that his (1,r) algorithm is identical to a single-linkage clustering Since CLUSTAR's single-linkage algorithm algorithm. had performed so well, Ling's (1,r) algorithm was compared to it to see if the (1,r) algorithm could be used. Indeed, Ling's (1,r) algorithm output the same results, but it had a few advantages which made it preferable to CLUSTAR's algorithm. First, the clustering results output by the algorithm provided a set of data points for each new cluster, not clustering tree diagrams as were seen before. Second, Ling had proposed two different decision making rules that could be used with his (1,r)algorithm to aid in the automatic extraction of isolated clusters. Since we have sought to automate the cluster extraction process, it was hoped that these decision rules beneficial in picking isolated clusters. would be the Furthermore, both algorithms took about the same amount of processing time to sort the data, so neither could be penalized due to their respective computing cost. Since both algorithms produce the same results for about the same cost, it was

-

Tracer Applied Sciences

decided to use Ling's (1,r) algorithm because the output was easier to use and because some decision making rules were available to aid in automating the cluster extraction process.

2.4 Data Normalization

It should be recalled from last year's report that the raw attribute data for each measurement had to be normalized in some fashion to give each attribute roughly the same range of numerical values before the resemblance matrix could be generated. For instance, 360 second trajectories have been used to generate the multi-target data, so the raw time units vary from 0 to 360. Over such a short time interval and for the center frequency values and geometries used in our Doppler shifted frequency measurements simulations, the typically vary by only 0.1 to 0.2 Hz. The bearings typically change by at most 20° to 30°, so the cosine and sine of the bearing measurements vary by only a few tenths. Unfortunately, this large variation in time values as compared to the change in the frequency and the cosine and sine of the bearing estimates caused the clustering algorithm severe problems when the raw attributes are used. When Euclidean distances were used to generate a resemblance matrix for the data, the large in time units dominated the difference dissimilarity coefficient while differences in the other attributes were virtually ignored. Because of this problem, the raw attribute measurements must somehow be scaled to provide the same order of magnitue of differences between sample measurements for each of the attributes.

Last year, experiments were run to test the five different data standardization techniques available in CLUSTER to find the single one that best fit our problem. The one which was finally picked scaled each of the attributes so that their respective range of values fell between 0 and 1. Specifically, in last year's report, each of the attributes were scaled by the following transformation:

$$\overline{\mathbf{X}}_{ij} = \frac{\mathbf{X}_{ij} - \min_{j} (\mathbf{X}_{ij})}{\max_{j} (\mathbf{X}_{ij}) - \min_{j} (\mathbf{X}_{ij})}$$

where:

.

10222222

XXXXXX

This normalization process worked adequately for our investigation, but one big problem was encountered with this The maximum and minimum values used for technique. normalization varied with each data set, so no two data sets were normalized in the same fashion. This was felt to cause large fluctuations in the values for the dissimilarity coefficients for the different data sets found in last years clustering tree diagrams. Due at least in part to this variation in the data normalization process, the cutoff clustering for halting the process varied threshold substantially from one data set to another. Because of these standard data normalization procedure was problems, а instituted in this year's investigation. The following procedure has been used to normalize the input attribute data for each measurement set:

> 1) $\overline{t} = t/120$ 2) $\overline{\cos\beta} = 5 \times \cos\beta$
3) $\sin\beta = 5 \times \sin\beta$ 4) $f = 10 \times f$

This procedure yields a consistent technique for scaling all the attributes for every problem. Furthermore, for the scenarios investigated here, this normalization procedure allows each attribute to vary by about two or three units. Tests have shown the data normalization procedure described above works very well, so this technique was used to normalize all data simulated in this study.

3.0

AUTOMATIC CLUSTER EXTRACTION

After Ling's (1,r) algorithm sorts the data into clusters, some way is needed to pick the best clusters out of the set. Last year when the tree diagrams were used, the a priori knowledge of how the data should be properly sorted was used to pick the best set of clusters. However, with Ling's (1,r) algorithm, we have sought to automate the process by which the best clusters are chosen. Thus, we are not forced to rely on user interpretation to find the best results and can make the procedure more usable for real applications. In this report, the process of automating the procedure for picking the best clusters is referred to as automatic cluster extraction. First isolated clusters are identified, and then two tests are cull both noise clusters employed to and internally inconsistent clusters. Following is a detailed discussion of this automatic cluster extraction process.

3.1 Single Linkage Clustering and Cluster Formations

A clustering procedure is said to be hierarchical if, for any given pair of clusters produced by the procedure, either one cluster is a subset of the other or they are disjoint. Thus, a hierarchical clustering procedure produces a nested structure that can be represented by a tree diagram or a dendrogram.

Single-linkage clustering is a hierarchical procedure that is based on the "nearest neighbor" concept. A brief description of single linkage clustering follows. Given a set of objects

 $s_{o} = \{ 0_{1}, 0_{2}, \dots, 0_{n} \}$

and a distance d that is defined for each object pair $\alpha(0_i, 0_j)$, the single linkage-clustering procedure first determines the closest pair of objects in the set and then joins them to form a cluster C. By defining the distance between cluster C and any object $0_{\rm L}$ to be

$$d(C, O_k) = \min \left\{ d(O_i, O_k): O_i \in C \right\}$$

and forming a new object set S that consists of C and all points of S_o not contained in C (i.e. $S_1 = \{C\} \cup \{S_o/C\}\}$), the above process can be repeated using S_1 in place of S_o. If the distance between two clusters is defined to be

 $d(c', c'') = \min \{ d(0_i, 0_j): 0_i \in C', 0_j \in C'' \}$

then the above procedure can be repeated for sets S₂, S₃, --- until all objects are grouped into one large cluster.

linkage clustering is Single a nonmetric procedure in the following sense. If the original distances $d(0_1, 0_1)$ are replaced by their rank orderings (i.e., the smallest is replaced by 1, the next smallest by 2, etc.), and the above algorithm is applied to the set S, then the sets S_2 , S_3 ,... will be unchanged (i.e., the same S₁, collection of clusters will be produced). Thus, each distance d on the object set S can be replaced by a symmetric nxn matrix, M_{d.s}, having integral entries ranging from 1 (denoting the distance between the closest pair of objects) to n(n-1)/2 (denoting the distance between the two objects that are farthest apart). Figure 3.1 illustrates the concepts discussed above. The fact that each pair (S, d) can be replaced by the pair (S_0, M_{d,s_0}) results in a great

ដ

N.V.

N.

AND DESCRIPTION REPORT REPORT AND DESCRIPTION AND DESCRIPTION OF A DESCRIPTION AND DESCRIPTION

$$S_0 = \{0_1, 0_2, 0_3\}$$

Object Combinations	Distance	Rank Ordering
$d(0_1, 0_2)$. 5	2
$d(0_1, 0_3)$	2	3
d(0 ₂ , 0 ₃)	. 25	1



 $S_2 = \{C_2\}$, $C_2 = \{0_1, 0_2, 0_3\}$

Figure 3.1. Clustering Example

simplification since for $S_0 = \{0_1, 0_2, \dots, 0_n\}$, there are only (n(n-1)/2)! distinct possible choices for M_{d,s_0} . This fact will be exploited when measures of cluster significance are considered since the output from a clustering algorithm consists of a large number of clusters, some of which are contained in, and some of which contain, the target data sets sought.

3.2 The Isolation Index

When performing cluster analysis on a set of objects, each cluster is formed at a particular clustering level. For single linkage clustering, these levels can be taken to be integers beginning with 1. Formally, the single linkage clustering level r of a cluster C is defined by

$$r = min \{k: C \in S_k\}.$$

Some authors refer to r as the "time of birth" of cluster C. If the sets S_1 , S_2 , S_3 , ... are called clustering sets, then the clustering level of a cluster C is just the index of the clustering set in which C first appears. The index k of set S_{μ} will be referred to as the clustering level k.

After a cluster has first formed at level r, it may remain unaltered for many future clustering levels. However, when C is not equal to the entire set of objects, it follows that eventually C will be augmented to form a new cluster C'. A cluster that remains unchanged for a number of levels is said to be well isolated. The following definition quantifies an isolation index measure possessed by cluster C:

$$L(C) = r' - r,$$

同

U

where r' is the clustering level of the augmented cluster C' that is described above. The clustering level r' can be thought of as the "time of death" of cluster C. In Figure 3.1, cluster C, is "born" at level 1 and "dies" at level 2. Therefore, $i(C_1) = 2 - 1 = 1$.

3.3 Isolation Index Distribution - Survival Function

Recall that the matrix M_{d.S} determines the clusters that are produced by a single-19nkage clustering By choosing a null probability model on M_{d.S}, procedure. it is possible to use probabalistic techniques to assess the significance of single-linkage clusters. The simplest and most conservative model to impose on the space of all possible random matrices M_{d.s} is to assume that each matrix is Under this assumption, the mathematics are equiprobable. tractable and the statistical theory resulting from the use of this model can be interpreted as establishing bounds for other models (i.e., a cluster that is judged to be statistically insignificant according to this model is unlikely to be significant under any other model). Under the equiprobable assumption, the isolation index of a cluster of size 1 becomes a random quantity whose distribution can be calculated. Specifically, if C is a cluster of size j with clustering level r. then

Prob [i (c) = s] =
$$\frac{\frac{1}{s} \left(j(n-j) \right) \left(N - r - j(n-j) \right)}{\binom{N-r}{s}}$$

where $N = \binom{n}{2}$. The probability of the event $\{i(C) \ge s\}$ being small implies that the chance of a cluster with this isolation index appearing simply by accident or at random is small.

Thus, a good assumption is that these data "belong together," i.e. came from the same target. Therefore, the survival function is used to help select the set of candidate clusters; a probability threshold is set and those clusters with survival function values falling below the threshold are passed on for further testing.

3.4

Polarization Test and Cluster Extraction

Once the isolation index has been used to identify clusters of data that possess greater than expected similarities, these clusters must then be analyzed to determine their randomness and internal consistency. Two primary reasons for doing this take advantage of the way in which single linkage clustering joins data together (see section 3.1):

- Random noise can appear as an isolated cluster, either when targets are present or when just random noise is clustered, and
- 2) As larger and larger clusters are formed from the data, isolated clusters can become subsets of larger isolated clusters. Eventually, there may be several isolated clusters contained in one large isolated cluster, and they may or may not be observations for the same target.

This section discusses the test for randomness employed by the MTTA. To motivate the selection of this test, imagine a cluster of data triples (time, frequency, and 22.2

1

bearing) projected onto the frequency-bearing plane, (Figure 3.2 provides several examples). Then,

- Figure 3.2.a represents random noise, frequency and bearing estimates are scattered randomly about,
- Figure 3.2.b represents data from a target showing significant bearing change but little Doppler change,
- 3) Figure 3.2.c represents data for a target that shows little bearing change but substantial frequency change, possibly approaching CPA with the sonobuoy, and
- 4) Figure 3.2.d represents data from a target with little bearing change or frequency change. This is a typical plot for a target in which a large amount of data has been gathered over a short time interval.

If a grid is imposed on the plane, then the number of observations occurring in a cell can be tallied and the number of observations occurring in a particular row or column can also be tallied. Then for clusters of random noise (as in Figure 3.2.a), the row totals should be roughly equal and the column totals should all be roughly equal while for target clusters, such as those displayed in Figures 3.2.b and c, either the row or column totals (or both) will display very multinomial equality. If a distinct departures from distribution is assumed for these row and column totals, then Alam and Mitra (Reference 4) have devised a test to determine







ملحاف

۲ ۲

1. W.

Figure 3.2. Bearing/Frequency Plot Examples

if the mass of the points is spread out evenly over the row and column totals or if it is concentrated, or polarized, in just a few of the rows and columns. A brief explanation of the test follows.

A partial ordering of the vectors $P=(p_1,\ldots,p_n)$ is defined by the relation:

 $\vec{P} > \vec{P}'$ if $\sum_{i=1}^{j} p_{(i)} \ge \sum_{i=1}^{j} p'_{(i)}$; j = 1, 2, ..., k

where $p_{(i)}$ is the ith largest value among p_1, \dots, p_k . Then, a multinomial distribution with probability vector \overrightarrow{P} is said to be less polarized than the multinomial distribution with probability vector \overrightarrow{P} if $\overrightarrow{P} * \overrightarrow{P}$. Note that

 $\left(\frac{1}{k}, \frac{1}{k}, \ldots, \frac{1}{k}\right) < \left(\frac{1}{k-1}, \frac{1}{k-1}, \ldots, \frac{1}{k-1}, 0\right) < (1, 0, 0, \ldots, 0).$

Let $\vec{R} = (r_1, \dots, r_k)$, $\sum_{i=1}^{n} r_i = n$ be the k row totals associated with a particular cluster and consider the following test:

 $H_0: \vec{R} = \vec{R}$ against the alternative $H_1: \vec{R} > \vec{R}$

where $\overline{R'}$ is the equally likely case. Alam and Mitra consider the test statistic

$$T(\vec{R}) = r_i^2/n$$

à

and show that H_0 is rejected for large values of T. The critical region of the test is

 $\{\overline{R}^{\dagger} \mid T(\overline{R}) > C\} \text{ where}$ $P\{T(\overline{R}) > C \mid \overline{R} = \overline{R}^{\dagger}\} = \alpha.$

For $\vec{R}^{T} = (1/k, 1/k, \dots, 1/k)$ Alam and Mitra show that

 $T^* = kT(\vec{R}) - n$

is asymptotically distributed as χ^2 k-1. A similar test can also be conducted for the column totals.

In applying this test, the grid size selected was 25 by 25, with bearing running from 0 to 2π radians and frequency from 148.8 to 151.2 Hz. In their paper, Alam and Mitra briefly discuss the convergence properties of the test and show that n must be fairly large before the χ^2 k-l approximation is very good. This was observed in the current study, for large clusters $(n \ge 60)$ testing at the .995 percentage point of distribution was the X 54 very effective at distinguishing noise clusters from data clusters. However, for clusters of size less than forty, the test still discriminated very well but the χ^2_{24} distribution did not provide a good critical value. Clusters composed mostly of observational data had high values of T* while clusters composed mostly of noise had much lower values of T*. However, the values of T* found for noise clusters were higher than the cut-off points supplied by the χ^2_{24} distribution.

-L

ì

•

2

3

At this point there were two basic choices:

- 1) Compute the conditional distribution function CDF of $T(\overline{R})$ directly using the recursion formula presented by Alam and Mitra, or
- 2) Determine if some simple, "rule-of-thumb" existed which would allow reasonable decisions to be made.

The second alternative was chosen because:

- Implementation of the Alam and Mitra CDF algorithm would have been a formidable programming task, particularly for n > 5,
- Including the calculation as a subroutine would have imposed a substantial computation penalty, and
- 3) A simple and reasonable "rule-of-thumb" value was readily available.

Primary considerations involved in selecting a cut-off value were:

1) It should not reject H_0 for very small clusters (n \leq 5) no matter how concentrated its members are,

- 2) It should reject H_0 for small clusters (5 $\leq n \leq 15$) only if most of their members are concentrated in one or two cells, and
- 3) It should reject H_o for moderate sized clusters (15 < N <40) only if a significant number (40% or better) of their members are concentrated in one or two cells.

Figure 3.3 contains an example of several graphs that were constructed for various values of T*. Simulation results had shown that for small time periods (≤ 5 minutes), clustered target bearing and frequency data would usually be confined to two or three cells, with some noise points possibly scattered about in other cells. Thus, curves were constructed for T* as a function of x_1 , x_2 , x_3 where:

- 1) x, was the number of points in cell i,
- 2) x_1 ranged from 0 to n,
- 3) x_2 ranged from 0 to $n-x_1$,
- 4) x_3 ranged from 0 to $n-x_1-x_2$, and
- 5) the remaining $n-x_1-x_2-x_3$ points were considered to be placed one to a cell.

Clearly, the value of T* is invariant to which cells contain the points, so that the choice of x_1 , x_2 , x_3 is not important.

The dashed lines of Figure 3.3 indicate the two cutoff points, the leftmost line corresponds to the .995

i j. -6.61.915 5.5 N.



Figure 3.3. Plots to Determine Polarization Test Cut off Point

percentage point of the χ^2_{24} distribution and the rightmost line corresponds to twice this number. Keeping in mind that the hypothesis being tested is

 $H_{o}: \vec{R} = \vec{R} = (1/k, \dots, 1/k) \text{ versus}$ $H_{1}: \vec{R} > \vec{R},$

it is clear that the rule-of-thumb value requires substantially higher cut-off values than does the approximation value. Additionally, the rule-of-thumb value will not reject H_0 for n < 5 and requires almost total concentration in a single cell for n near 5. For $n \ge 10$, concentrations in the 40% to 60% range are required for single cell polarization and in the 80% - 90% range for two and three cell polarizations. The comparable χ^2_{24} approximation values are at least 20% - 30% below this. Using the rightmost line as the cut-off has worked quite well in this study. In order for a cluster to be declared "random" it must accept H_0 for both the row and column totals.

3.5 Internal Consistency and Cluster Extraction

This section discusses the final stage in the cluster extraction process, the test for internal consistency. Figure 3.4 illustrates the logic flow for the procedure.

At this point in the extraction process, the clusters under consideration are assumed to be well-isolated, non-random groups of data. What remains is to examine for consistency each cluster that contains two or more isolated clusters as subsets. Basically, the idea is to determine whether the information contained in all the isolated subclusters is compatible or not.







a de la companya de l

and the second

3.5.1 <u>Regression and Internal Consistency</u>

One of the primary difficulties in finding an internal consistency test for isolated clusters which contain other isolated clusters as subsets is that the frequency and bearing estimates are functions of time within these clusters. Typically, these isolated subclusters correspond to time segments of observations for one or more targets; they start as small groups of two or three points and then chain outward until the cluster under consideration is formed. Under these circumstances, usual measures of similarity such as means, ranges, modes, etc., can be very misleading. For a particular target, the average frequency, bearing sine, or bearing cosine from the first part of an observation period can be quite different from the average computed from the last part of the observation period. Based on these considerations, regression techniques were used to develop the internal consistency test.

-

ij

.

When frequency, bearing sine, and bearing cosine for a particular target are plotted against time for short intervals (less than six minutes), they are nearly linear functions and can be approximated fairly well by straight Measurements from two different and distinguishable lines. targets define different straight lines, and this forms the basis for the use of single-linkage clustering techniques. However, as was shown in Section 3.1, the single-linkage algorithm creates ever larger clusters by joining points or clusters to already existing clusters. If clusters are joined from two different and distinguishable targets, then the problem of internal consistency can be cast as a regression problem to determine whether a set of a data defines one or two regression lines. The lines could differ by a change in intercept, a change in slope, or by both.

Neter and Wasserman (Reference 6) give a procedure for doing this. The basic formulation of the model is:

- Let Y_i be the measurement under examination (frequency, bearing sine, or bearing cosine) and let T_i be the time of the measurement,
- 2) Let I_i be an indicator variable attached to Y_i such that $I_i = 0$ if Y_i comes from the first cluster and $I_i = 1$ if Y_i comes from the second cluster, and
- 3) The model becomes

 $Y_{i} = B_{0} + B_{1}T_{i} + B_{2}I_{i} + B_{3}T_{i}*I_{i} + \epsilon_{i}$

and the response (regression function) is

$$E(Y) = B_{A} + B_{1}T + B_{2}I + B_{3}T*I.$$

To understand the function of the indicator variable in this model, note that for measurements from the first cluster (I=0),

$$E(Y) = B_0 + B_1 T + B_2 (0) + B_3(0)$$

= B_0 + B_1 T

and for measurements from the second cluster (I=1),

$$E(Y) = B_0 + B_1T + B_2(1) + B_3(1)(T)$$

(B_0 + B_2) + (B_1 + B_3)T.

Thus, B_2 measures the difference in intercept estimates between the two lines and B_3 measures the differences in slope estimates between the two lines. Therefore, the test for equality of the two regression lines corresponds to a test of the hypothesis

> $H_{o}: B_{2} = B_{3} = 0$ versus $H_{1}: B_{2}$ or $B_{3} \neq 0$.

The test statistic is

 $\frac{SS(B_2, B_3|B_0, B_1)}{2} \div \frac{SSE}{n_1 + n_2 - 4} = F^*$

where

- 1) SS $(B_2, B_3 | B_0, B_1)$ is the reduction in the regression sum of squares due to the inclusion of I, and T*I in the model,
- 2) SSE is the error sum of squares for the model, and
- 3) n₁ is the number of objects in cluster 1 and n₂ is number of objects in cluster 2.

The above procedure outlines a method for determining whether a set of measurements coming from two different clusters is internally consistent or not. This is done for each of the three measurements available i.e. frequency, bearing sine, and bearing cosine. A weighted decision scheme is then used based on the three calculated

F-values (F_f , F_{BS} , and F_{BC}), to determine whether the two clusters contain consistent information or not. Considerations used to establish this scheme include:

- Frequency information should be weighted less than bearing sine and bearing cosine information. This is done because frequency is only measured to the nearest tenth of a Hertz, so a normal Doppler shift of only 0.1 Hz can indicate a spurious inconsistency,
- 2) The bearing sine (bearing cosine) measurements from two targets may be quite close while the bearing cosine (bearing sine) measurements are substantially different. Thus, only one of the bearing component measurements is needed to indicate inconsistency, and
- 3) When in doubt, pass a cluster. The result of passing a cluster that should not be passed is an increase in processing time for other areas of the MTTA algorithms. The result of not passing a cluster that should be passed is the possible loss of a target track.

The weighting scheme (to be discussed below) depends on a method developed by Suich and Derringer, (Reference 13) and extended by Ellerton (14) to examine the significance of subsets of the regression variables. Typically, this subset of variables describes some particular characteristic, such as trend or curvature, of the response

system. In our case, the characteristic being examined is the tendency of the data to be best described by two regression lines instead of one.

Let $Y_f(\vec{x})$ represent the full regression model and let $Y_r(\vec{x})$ represent the reduced regression model, that is, the full model minus the variable subset under consideration. The quantity to be tested, γ , is the ratio of the average, squared expected differences in prediction of the two models to the average variance of these predicted differences;

 $\gamma = \frac{\text{average expected squared prediction difference}}{\text{average variance of predicted differences}}$

The test statistic is the usual F-ratio, which in this case is,

$$\frac{SS(B_2, B_3|B_0, B_1)}{2} \div \frac{SSE}{n_1 + n_2 - 4}$$

and the hypothesis tested is

$$\begin{array}{l} H_{o}: \quad \gamma = \gamma_{o} \text{ versus} \\ H_{1}: \quad \gamma > \gamma_{o}. \end{array}$$

Acceptance of the null hypothesis indicates that either the full model should be reassessed as a predictor or that the error variation in the data is simply too large relative to the variation in predictive power of $\hat{Y}_f(\vec{x})$ and $\hat{Y}_r(\vec{x})$. Essentially, if there are large errors relative to the range covered by the data, then a subset of variables may, under the usual F-test, appear to be significant when in fact they are simply fitting the errors in the data. The purpose of this test is to detect that situation.

Ē

In the extraction algorithm γ_0 was set equal to 2, and the critical value then became about four times the F-value used to make the standard test:

3.5.2 Weighting Scheme

Using the regression analysis just described, the interval consistency test reduces to the following, letting F_u stand for the "usual" regression F critical value and F_E stand for the value generated by applying Ellerton's work, the weighting scheme used is:

1) If
$$\begin{cases} F_{f}^{\star} < F_{u} \\ F_{f}^{\star} \ge F_{u} \end{cases}$$
 then $\begin{cases} W_{f} = 0 \\ W_{f} = 1 \end{cases}$

- 2) If $\begin{pmatrix} F_{BS}^{*} < F_{u} \\ F_{u} < F_{BS}^{*} < F_{E} \\ F_{BS}^{*} \ge F_{E} \end{pmatrix}$ then $\begin{pmatrix} W_{BS} = 0 \\ W_{BS} = 1 \\ W_{BS} = 2 \end{pmatrix}$
- 3) If $\begin{cases} F_{BC}^{\star} \leq F_{u} \\ F_{u} \leq F_{BC}^{\star} \leq F_{E} \\ F_{BC}^{\star} \geq F_{E} \end{cases}$ then $\begin{cases} W_{BC} = 0 \\ W_{BC} = 1 \\ W_{BC} = 2 \end{cases}$
- 4) $W = W_f + W_{BS} + W_{BC}$, and

5) If $W \ge 2$ then reject the hypothesis that the two clusters are consistent.

Basically, this scheme says that frequency variation alone is not enough to reject the idea of consistency, instead it must reject in concert with either bearing sine or cosine. On the other hand, bearing sine or cosine measurements are enough by themselves to reject consistency if the indication of multiple cluster data is extremely strong. Middle of the road rejection values from both bearing cosine and sine measurements are also enough to reject consistency. Because a false rejection of the consistency hypothesis is much more costly than a false acceptance, all critical values were set with $\alpha = .005$.

If a particular cluster possesses exactly two isolated subclusters, then the above procedure is fine as it stands, the cluster will be accepted or rejected based on the final value of W. However, if a given isolated cluster contains more than two isolated clusters as subsets, then another procedure for rejecting consistency is needed. Suppose the cluster in question has k isolated subclusters, then there are $\binom{K}{2}$ cluster pairs to be examined for consistency. If it is assumed that all clusters come from the same track, then k-l of the potential pairs should link together or show consistency. Thus, for clusters with k isolated subclusters (k > 2), the consistency hypothesis will be rejected if fewer than k-1 pass the two cluster consistency test. cluster pairs Experience has shown this to be a fairly liberal criterion.

3.6 Automatic Cluster Extraction Summary

In order to be selected as a target data set a cluster must possess three characteristics:

1) Isolation from the rest of the data,

- 2) Polarization of observations in a few frequency or bearing bins, and
- 3) Internal consistency among isolated subclusters.

Cluster isolation is measured using Ling's survival function. Those clusters possessing a survival function value of less than 10⁻⁴ are considered well isolated and selected for further testing. Once a cluster is considered isolated it is subjected to the Alam and Mitra polarization test to determine if most of its members are concentrated in a few bins or not. Those clusters whose observations are concentrated in a few bins are determined to be nonrandom collections and are passed on to the final test. Clusters remaining at this point are examined for subclusters which happen to be isolated clusters also. Those clusters which contain no isolated subclusters are considered internally consistent and declared target data sets. Clusters containing isolated subclusters are given a regression test for internal consistency. Those that pass the test are passed as target data sets.

All the above tests are conducted using hypothesis tests with alpha values set high to allow marginal groups to pass through and be selected. It is felt that it is better to pick a bad cluster as a potential target data set than reject a cluster that is a target data set.

4.0

INITIAL GUESS PROCEDURE

Once the sonobuoy data have passed through the cluster formation and cluster extraction algorithms, they have been grouped into sets of clusters for each sensor that correspond to potential target data sets. The next step is to solve the intersensor correlation problem which seeks to determine which data clusters from each sensor contain observations on the same target. For this study, simultaneous observations for a target were required from either two or three sensors because, in general, a single sonobuoy cannot provide enough information with which to track a target. For even a small number of sensors and targets, the number of potential combinations can become fairly large. For example, consider a 3 target scenario observed by three sonobuoys and assume that each buoy hears each target perfectly and that the cluster formation and extraction routines perfectly separate There are then three data clusters associated with the data. each sensor, and these can be combined into 27 (i.e., 3*3 + 3*3+ 3*3) potential two-cluster intersensor combinations and 27 (3*3*3) potential three-cluster intersensor combinations.

To help separate unlikely combinations from likely combinations, an initial guess procedure and a set of guess evaluation criteria were developed. This section describes the initial guess procedure itself and the following section discusses the evaluation criteria.

4.1 <u>Crossed-Bearing, Crossed-Frequency Initial Guess</u> <u>Procedure</u>

An initial guess procedure which used crossed-bearing and crossed-frequency information to generate

initial conditions for target tracks was described in last year's report (Reference 3). It was used to generate a reasonable guess of a target's position and velocity. This guess was then passed to the initializers of either Tracor's EKF or HTA so that the tracker could more quickly converge onto an acceptable set of initial conditions for the target. This procedure used the frequency and bearing measurements from two or more sensors in a least squares formulation to estimate the target's initial position and velocity. The estimates from this procedure were found to be reasonably good and greatly improved the speed and the initialization characteristics of Tracor's two trackers.

For a bearing estimate from sensor i, the crossed-bearing equation is:

$$\kappa \sin \beta_i - y \cos \beta_i = x_i \sin \beta_i - y_i \cos \beta_i$$

where

 β_i = the bearing estimate for sensor i, (x_i, y_i) = the position components for sensor i, (x, y) = the position components for the target.

Similar crossed-bearing equations can also be obtained for any other observing sensors. The crossed-frequency formulation uses the bearing and frequency estimates from a given sensor to generate target velocity estimates. The actual crossed-frequency equation for buoy i is:

$$\dot{\mathbf{x}}\cos\beta_{\mathbf{i}} + \dot{\mathbf{y}}\sin\beta_{\mathbf{i}} = c\left[\frac{\mathbf{f}_{o}}{\mathbf{f}_{\mathbf{i}}}\left(1 + \frac{\dot{\mathbf{x}}_{\mathbf{i}}\cos\beta_{\mathbf{i}} + \dot{\mathbf{y}}_{\mathbf{i}}\sin\beta_{\mathbf{i}}}{c}\right) - 1\right]$$

where

β ₁	= Bearing estimate from buoy i,
fi	 Doppler-shifted frequency estimate from buoy i,
fo	 Unshifted center frequency transmitted by the target,
с	= Speed of sound in the water,
(ż,ÿ)	= Velocity components for the target,
(\dot{x}_i, \dot{y}_i)	= Velocity components for the buoy.

Once again, similar crossed-frequency measurement equations can be generated for other sensors that observe the target.

A motion model is needed to correlate the target's position and velocity estimates and to provide a means of mapping all measurements back to an initial epoch so that a batch-type, least squares estimate can be generated for the target's initial position and velocity. Since only a small data stream which covers a short time span is used in the initial guess procedure, a linear motion model was selected to describe the target's trajectory. The linear motion model used is:

$$\mathbf{x} = \mathbf{x}_{0} + \dot{\mathbf{x}}_{0} \Delta \mathbf{t}$$
$$\mathbf{y} = \mathbf{y}_{0} + \dot{\mathbf{y}}_{0} \Delta \mathbf{t}.$$

Following Bard (Reference 15), the crossed-bearing and crossed-frequency equations can be combined with the motion model in a two-equation least squares model to estimate $(x_0, y_0, \dot{x}_0, \dot{y}_0)$.

4.2 <u>Measurement Weighting Equations</u>

Last year, least squares estimates for the target's initial state vector were generated by weighting all the measurements equally. This resulted in good position estimates but relatively poor velocity estimates. It was felt that weighting the measurements might improve the target's velocity estimates.

Associated with each frequency and bearing measurement is a standard deviation value which describes statistically the accuracy of the measurement. These standard deviations are computed as functions of the estimated signal-to-noise ratio for the received signal. Typically, the measurement standard deviations are used to compute a measurement weight which enables the data to be processed by a weighted least-squares tracking algorithm. With the crossed-bearing and crossed-frequency initial guess algorithm, the measurement equations are functions of the frequency and bearing estimates. Some method has to be used which computes standard of the crossed-bearing deviations and the crossed-frequency measurements as functions of the standard deviations computed for the frequency and bearing estimates. Following Young (Reference 16), a first order Taylor series approximation is used to compute the standard deviation for the modified measurement equation. If we use G to represent either the crossed-bearing or the crossed-frequency measurement function and f and β to represent the frequency and bearing estimates, respectively, the following equation is used to compute the measurement weights:

 $\sigma_{\mathbf{G}}^{2} = \left(\frac{\partial \mathbf{G}}{\partial \mathbf{f}}\right)^{2} \quad \sigma_{\mathbf{f}}^{2} + \left(\frac{\partial \mathbf{G}}{\partial \beta}\right)^{2} \sigma_{\beta}^{2} .$

N N N

Y

In the above formulation, the assumption has been made that the buoy positions and their velocities are perfectly known, so no buoy position or velocity errors are allowed to propagate through this standard deviation equation. Specifically for the bearing equation, the measurement equation, G, is:

 $G = x_i \sin \beta_i - y_i \cos \beta_i$.

Since no buoy position errors are encountered and since the frequency estimate does not contribute to this measurement equation, the crossed-bearing standard deviation becomes a function of the bearing estimate only

$$\sigma_{\mathbf{G}}^{2} = \left(\frac{\partial \mathbf{G}}{\partial \beta}\right)^{2} \sigma_{\beta}^{2}$$

 $\sigma_{G}^{2} = (x_{i} \cos \beta_{i} + y_{i} \sin \beta_{i})^{2} \sigma_{\beta_{i}}^{2}$

Conversely, the crossed-frequency measurement equation is a function of both the bearing and frequency estimates

$$G = c \left[\frac{f_o}{f_i} \left(1 + \frac{\dot{x}_i \cos \beta_i + \dot{y}_i \sin \beta_i}{c} \right) - 1 \right]$$

To obtain the crossed-frequency standard deviation, it is assumed that no errors are encountered in obtaining buoy state values so that they do not contribute to the sources for the crossed frequency measurements:

> c - speed of sound in the water, (\dot{x}_i, \dot{y}_i) - velocity components of buoy i, f_o - unshifted center frequency value for the transmitted target signal.

. . . -1 ۰⁻ ۲ M

Tracor Applied Sciences

The crossed-frequency equation for sigma then reduces to:

$$\sigma_{\mathbf{G}}^{2} = \left(\frac{\partial \mathbf{G}}{\partial \beta}\right) \sigma_{\beta}^{2} + \left(\frac{\partial \mathbf{G}}{\partial \mathbf{f}_{\mathbf{i}}}\right)^{2} \sigma_{\mathbf{f}_{\mathbf{i}}}^{2}$$
$$\sigma_{\mathbf{G}}^{2} = \left[\frac{\mathbf{f}_{\mathbf{0}}}{\mathbf{f}_{\mathbf{i}}} \left(-\dot{\mathbf{x}}_{\mathbf{i}} \sin \beta_{\mathbf{i}} + \dot{\mathbf{y}}_{\mathbf{i}} \cos \beta_{\mathbf{i}}\right)\right]^{2} \sigma_{\beta}_{\mathbf{i}}^{2}$$
$$+ \left[\frac{-c\mathbf{f}_{\mathbf{0}} \left(1 + \frac{\dot{\mathbf{x}}_{\mathbf{i}} \cos \beta_{\mathbf{i}} + \dot{\mathbf{y}}_{\mathbf{i}} \sin \beta_{\mathbf{i}}}{\mathbf{f}_{\mathbf{i}}^{2}}\right)\right]^{2} \sigma_{\mathbf{f}}_{\mathbf{i}}^{2}$$

Using the weights generated by these equations in the estimation procedure has greatly improved the velocity estimates without affecting the position estimates from the MTTA's IG algorithm. The IG algorithm now in use in the MTTA, crossed-bearing, therefore, is а weighted least-squares crossed-frequency algorithm.

5.0

PHYSICAL AND STATISTICAL CONSTRAINTS ON THE INITIAL GUESS ESTIMATES

After initial guess estimates are generated for each of the possible intersensor cluster combinations, these estimates must be examined to determine which ones provide which are plausible initial conditions in real world encounters. As stated previously, there are twenty-seven possible two-cluster intersensor combinations, and twenty-seven, three-cluster intersensor combinations for the three-target, three-sensor scenario shown in Figure 7.1. Figure 5.1 shows all the possible bearing line-of-sight intersections that would result from combining bearings from two sensors in the crossed-bearings procedure. Notice the large number of intersections that result from this simple example. Although it is not as easy to illustrate, a large number of possible intersections also result from using the crossed-bearing procedure for all possible three-cluster, intersensor combinations. This section examines ways to use physical and statistical constraints to eliminate the unlikely combinations produced by the initial guess procedure.

5.1

Physical Constraints Test

The first step in eliminating implausible initial guess values is to examine the estimates in terms of sonobuoy detection limits and maximum submerged target performance levels. These limits are referred to as the physical constraints on the problem. These constraints have been left as user defined inputs so that they may be varied for the different classes of sonobuoys or the different types of targets that might be encountered in real engagements.



2

Ø

The first physical constraint fixes a maximum detection range for the sonobuoys to eliminate candidate initial guess estimates that fall well outside the detection range of the sonobuoy pattern. In real applications, an operator uses parameters such as the class of sonobuoys used, the minimum detectable signal level (MDL) for these sonobuoys, ambient conditions such as sea state, and other factors to set a maximum range for the day for the sonobuoys. For our simulation, we used a transmitted signal-to-noise ratio of 80 dB in a 1 Hz band at a distance 1 yard from the source to simulate each of the target's signal sources. Furthermore, the MDL used for our simulation was set to a reasonable value of 0 dB in a 1 Hz band (this gives +10 dB in a 0.1 Hz band). Given this 80 dB source level, a 0 dB MDL, and assuming a 20 log R signal propagation loss through the water, the maximum detectable range for each of the sonobuoys would be 10,000 yds. To insure a sizable safety margin, a 50% error tolerance was added to this 10,000 yd range, so the maximum allowable range was set at 15,000 yds. Any initial guess estimate whose position values would cause the target to fall outside this 15,000 yd range for any of the observing sensors was then rejected and was never considered again in the MTTA.

The second physical constraint restricted the target's speed to some maximum value which once again was set by the program user. For actual applications, operators usually know what the maximum submerged speeds are for certain classes of targets. If not, the operator could set the upper limit to the maximum known value for any class of targets. For this simulation, the maximum target speed was set to 10 m/sec, which corresponds to roughly 20 knots. Once again, a 50% error tolerance was added to this value so that the maximum allowed speed used by the program was 15 m/sec. This tolerance was

. ــــ ۲.2 •.• ي. تر -. ′**.** , ----

Tracor Applied Sciences

much larger than the speed of the fastest target in the scenario which was 9 m/sec, so this upper limit was considered to be reasonable for our experiments.

With these physical constraints, several of the guess initial estimates could be rejected. so their corresponding intersensor combinations could also be eliminated from further consideration. However, many more combinations passed these two tests than were rejected, so many of the combinations were still under active consideration at this point. If one looks at Figure 5.1, one sees that most of the two-cluster intersensor combinations fall well within the 15,000 yd range of their participating sonobuoys. While the target speed constraint helped reject a few of the remaining combinations, most combinations were still left intact after this test also. Since so many combinations were left, it was decided to perform some statistical tests on the remaining cluster combinations to examine the quality of the initial guess estimates, and to then eliminate all estimates that were found to fit the data poorly in a statistical sense.

5.2

Two Sensor Statistical Compatability Test

After a particular two sensor initial guess estimate has passed the physical constraints test, it is then subjected to a statistical test that measures the consistency of the information contained in the contributing clusters. This test is similar to the test used to determine cluster consistency and is based on the use of indicator variables.

From section 4.2, it will be recalled that the initial guess procedure uses a weighted least squares method to

-

Ē

estimate the target's initial state vector $(x_0, y_0, \dot{x}_0, \dot{y}_0)$. In doing this, it fits a model of the form

$$M_{i} = B_{1}Z_{i1} + B_{2}Z_{i2} + B_{3}Z_{i3} + B_{4}Z_{i4} + \epsilon_{i}$$

where (using the notation of section 4.1):

 $M_{i} = \begin{cases} x_{1} \sin\beta_{1} - y_{1} \cos\beta_{1} \text{ for sensor } 1 \text{ 's bearing data} \\ x_{2} \sin\beta_{2} - y_{2} \cos\beta_{2} \text{ for sensor } 2 \text{ 's bearing data} \\ c \left[\frac{f_{0}}{f_{1}} \left(1 + \frac{\dot{x}_{1} \cos\beta_{1} + \dot{y}_{1} \sin\beta_{1}}{c} \right) - 1 \right] & \text{For sensor } 1 \text{ frequency} \\ data \\ c \left[\frac{f_{0}}{f_{2}} \left(1 + \frac{\dot{x}_{2} \cos\beta_{2} + \dot{y}_{2} \sin\beta_{2}}{c} \right) - 1 \right] & \text{For sensor } 2 \text{ frequency} \\ data \end{cases}$

 $(B_1, B_2, B_3, B_4) = (x_0, y_0, \dot{x}_0, \dot{y}_0)$

 $(Z_{i_1}, Z_{i_2}, Z_{i_3}, Z_{i_4}) = \begin{cases} (\sin\beta_1, -\cos\beta_1, \Delta t \ \sin\beta_1, -\Delta t \ \cos\beta_1) \\ \text{for sensor 1's bearing data} \\ (\sin\beta_2, -\cos\beta_2, \Delta t \ \sin\beta_2, -\Delta t \ \cos\beta_2) \\ \text{for sensor 2's bearing data} \\ (0, 0 \ \cos\beta_1, \ \sin\beta_1) & \text{for sensor 1's} \\ \text{frequency data} \\ (0, 0 \ \cos\beta_2, \ \sin\beta_2) & \text{for sensor 2's} \\ \text{frequency data} \end{cases}$

 ϵ_i - the measurement error associated with the ith observation.

When using two sensor data to estimate the initial state vector of a particular target, there are only two possibilities:

- 1) The two clusters contain observations which generate estimates for an actual target, or
- 2) The two clusters contain information which leads to estimates for a spurious target.

The problem is to separate spurious target estimates from real ones. Figure 5.1 illustrates the potential number of bearing intersections for even a simple example. Because a linear least squares procedure is used, the state vector estimated by the initial guess procedure is a weighted average of all the measurements from both clusters. This will result in producing a great number of the spurious intersections shown in Figure 5.1. The question of separating real from spurious estimates can then be viewed as one of determining whether the two clusters are estimating the same target or not.

One way to do this is to use indicator variables, that is, to formulate the model as above with an additional term Z_{15} ;

 $M_{i} = B_{1}Z_{i1} + B_{2}Z_{i2} + B_{3}Z_{i3} + B_{4}Z_{i4} + B_{5}Z_{i5} + \epsilon_{i}$

where $Z_{i5} = \begin{cases} 1 & \text{if the observation comes from cluster l} \\ 0 & \text{if the observation comes from cluster 2.} \end{cases}$
à

a deserver a

accontenta 1 - NAV

CANARASAN CARARASAN PERSENTE ARAGAMANAN I

ci

Ē

The response model associated with this is

 $E(M_{i}) = B_{1}Z_{1} + B_{2}Z_{2} + B_{3}Z_{3} + B_{4}Z_{4} + B_{5}$ for observations from cluster 1,

 $E(M_i) = B_1Z_1 + B_2Z_2 + B_3Z_3 + B_4Z_4$ for observations from cluster 2.

If B_5 is significant, then there is a significant difference in the mean response of the system to the data from the two sensors. In essence, B_5 is a measure of the degree of averaging required between the two data sets to generate an initial state vector estimate. A large amount of averaging implies two different targets are producing the estimate, a small amount of averaging implies only one target produced the estimate.

Thus, the two sensor consistency test reduces to a standard linear model test of the significance of the indicator variable Z_5 . This corresponds to the hypothesis test

 $\begin{array}{ll} H_{o}: & B_{5} = 0 & \text{versus} \\ H_{1}: & B_{5} \neq 0 \end{array}$

with test statistic

$$F^{*} = \frac{SSR (Z_{5} | Z_{1}, Z_{2}, Z_{3}, Z_{4})}{1} \div \frac{SSE (Z_{1}, Z_{2}, Z_{3}, Z_{4}, Z_{5})}{n_{1} + n_{2} - 5}$$

where

 $SSR(Z_5|Z_1,Z_2,Z_3,Z_4,)$ is the reduction in regression sum of squares caused by including Z_5 ,

 $SSE(Z_1, Z_2, Z_3, Z_4, Z_5)$ is the error sum of squares for the full model,

 n_1 , n_2 - number of points in clusters 1 and 2, and

F* - has an F distribution.

The decision rule is:

if $F^* \leq F_c$ (1, $n_1 + n_2 - 5$) accept H_o if $F^* > F_c$ (1, $n_1 + n_2 - 5$) accept H_1

where

 F_c (1, $n_1 + n_2 - 5$) is the desired critical point of the F-distribution with 1 and $n_1 + n_2 - 5$ degrees of freedom.

Again, because the cost of a type I error is much higher than that of a type II error, the alpha level for this test was set at .005.

5.3 Three Sensor Statistical Compatability Test

Once all two-sensor initial guess estimates have been examined, attention is turned to potential three-sensor initial guess estimates. As with the two-sensor estimates, the three-sensor state estimates are made up of weighted averages

<u>고</u> :4

.

R

9

Lever excel

a second

of observations taken from all three clusters. Thus, spurious target estimates which satisfy the physical constraints of the problem often occur.

Figure 5.2 contains a flowchart of the three statistical Note that for sensor test. а particular three-sensor combination to be examined, at least two of the associated two-sensor estimates must have passed the two-sensor test. Once a three-sensor initial guess has been calculated, a nonlinear least squares hypothesis test developed by Gallant (Reference 7) is used to decide whether this estimate is statistically equal to each of the previously computed It is this test which forms the heart of two-sensor estimates. the three-sensor compatability test. If at least two of the associated two-sensor estimates are statistically equal to the three-sensor estimate, then the three-sensor estimate is passed on for further consideration.

The intent of the three-sensor test is to search for consistency among the already computed two-sensor estimates and the current three-sensor estimate. Requiring agreement with at least two of the possible two-sensor estimates prevents one two-sensor estimate from dominating the three-sensor estimate.

The Gallant nonlinear regression test (Reference 7) considers the hypothesis

$$H_0: \vec{X} = \vec{X}_0 \quad \text{versus}$$
$$H_1: \vec{X} \neq \vec{X}_0$$



٠,

5 5 5 5 5 F

Figure 5.2. Initial Guess Procedure's 3-Sensor Combination Tests

· . • . •

at the α -level of significance where

x₀

represents the initial guess, state vector estimate under consideration.

The data, M_i , are measurements assumed to be responses to the input vector $\vec{Z_i}$ linked by the non-linear regression model

$$M_{i} = f(\vec{z}_{i}, \vec{x}_{o}') + e_{i}.$$

Under the null hypothesis the quantities

 $M_i - f(\bar{Z}_i, \bar{X}_o') = e_i$

are called the measurement residuals and are assumed to be independent, normal, random variables with zero mean. If each measurement has an associated positive weight W_i , then the following sum

 $S(\vec{x}_{o}) = \sum W_{i}e_{i}$

is the weighted residual sum of squares for \vec{X}_{a} .

Then, for the initial guess routine the regression function is:

 $M_{i} = Z \cdot [(x_{o} + \dot{x}_{o} \Delta t_{i}) \sin \beta_{is} - (y_{o} + \dot{y}_{o} \Delta t_{i}) \cos \beta_{is}] + (1-Z) \cdot [\dot{x}_{o} \cos \beta_{is} + \dot{y}_{o} \sin \beta_{is}]$

where

(x _o , y _o , x _o , y _o)	-	-	initial state vector of target,
^{Δt} i sin ^β is		-	t _i - t _o , sine of bearing estimate
cos ^β is		-	from sensor s, cosine of bearing estimate from sensor s.
Z =	1	-	when measurement is crossed-bearing,
	(0	-	when measurement is crossed-frequency.

In Gallant's notation, the input vector, \overline{Z} , for each measurement is:

$$\overline{Z} = (Z_1, Z_2, Z_3, Z_4) = (Z, \Delta t_i, \sin \beta_{is}, \cos \beta_{is})$$

, ,

-

4

and the parameter vector is:

$$\vec{x}_{0} = (x_{0}, y_{0}, \dot{x}_{0}, \dot{y}_{0}).$$

The test statistic used is

$$\mathbf{r}(\vec{\mathbf{x}_{o}}) = \frac{\mathbf{s} \quad (\vec{\mathbf{x}_{o}})}{\mathbf{s} \quad (\vec{\mathbf{x}_{o}})}$$

ومقادرة والمعادية í

Tracer Applied Sciences

where

- are values for the previously computed two sensor state vector estimates,

 \vec{X}_{0} - the three-sensor state vector estimate.

The critical point is given by

$$C^* = 1 + \frac{pF_{o}(p, n-p)}{n-p}$$

₹

where

		freedom.
		and n-p denominator degrees of
		F-distribution with p numerator
F _a (p, n-p)	-	α percentage point of the
n	-	number of data points
ä	-	number of parameters estimated

Lastly, the decision rule is:

if $T(\vec{X_{o}}) \leq c^{*}, \text{accept } H_{o}$ if $T(X_{o}) > c^{*}, \text{accept } H_{1}$.

The test procedure then is:

1) Get the three-sensor data and compute an initial state vector \vec{X}_{o} with the initial guess procedure,

2) Compute $S(\overline{X}_{o})$,

- 3) For each of the previously computed two-sensor estimates, \vec{X}_{o} , associated with the three sensors under examination, compute $S(\vec{X}_{o})$ (using the three sensor data),
- 4) Perform Gallant's test on each of the two-sensor estimates, and
- 5) If two of the two-sensor estimates pass, then pass the three-sensor estimate as good, otherwise fail it.

The test statistic T is derived by finding the likelihood ratio for the composite hypothesis

$$H_{0}: \vec{X} = \vec{X} \quad \text{versus}$$
$$H_{1}: X \neq \vec{X}_{0}'$$

The exact distribution of T is not known, but Gallant finds a random variable which is asymptotic to T and whose distribution is known. It is from this asymptotic distribution that C* is determined. Basically, the test examines the ratio of the residual sum of squares for the two-sensor and three-sensor estimates. By definition, the two-sensor estimates will always produce larger values of $S(\vec{X})$ than the three-sensor estimate, so Gallant's test provides a measure of what is too large.

6.0 INTEGER PROGRAMMING AND SCENARIO SELECTION

Up to this point, the MTTA has been concerned primarily with examining multi-sonobuoy data and organizing it into sets of observations that form potential tracks. These potential tracks must now be organized into likely track scenarios. That is, given the current set of observations, determine which set of potential tracks most likely describes the actual events taking place. With even a small number of potential tracks, this becomes a combinatorial problem of fairly large dimensions. To sift through the various potential scenarios to select the optimal set, Tracor has used a linear optimization technique known as integer programming (IP).

The most general integer programming formulation

is:

min c'x $= \sum_{i=1}^{n} c_i x_i$ such that $X \in S \subset z^n$

where

c' is an n-dimensional cost function
S is the constraint set
zⁿ is the set of all n-dimensional integer vectors.

Typically, integer programming problems are difficult to solve so a great deal of research has concentrated on identifying and finding efficient solution techniques for certain special classes of this problem. Fortunately, the scenario selection problem falls into a well known problem type called the set partitioning problem.

6.1 Set Partitioning Formulation

Garfinkel and Nemhauser (Reference 17) give the basic formulation of the set partitioning problem. Consider a set I = $\{1, 2, ..., m\}$ and let P be a set such that P = $\{P_1, P_2, ..., P_n\}$ where $P_j \subseteq$ I for the index set J = $\{1, ..., n\}$. Then a subset J* of J defines a partition of I if

and for j, $k \in J^*$, j $\neq k$ implies

 $P_j \cap P_k = \phi.$

Thus, a partition of I consists of a set of disjoint subsets of I such that their union is equal to I.

Let a cost, c_j , be associated with each $j \in J$ so that the total cost associated with a given partition is $\sum_{j \in J^*} c_j$. The set partitioning problem is to then find the

partition of I having the minimum cost. Written as an IP, this problem is:

 $\min \sum_{j=1}^{n} c_{j} x_{j}$ subject to $\sum_{j=1}^{n} a_{ij} x_{j} = 1, i = 1, ..., m$ $x_{i} = 0, 1; j = 1, ... n$

where

Ģ

 $x_j = \begin{cases} 1 & \text{if } j \text{ is in the cover} \\ 0 & \text{otherwise} \end{cases}$

 $a_{ij} = \begin{cases} 1 & \text{if } i \in P_j \\ 0 & \text{otherwise.} \end{cases}$

The optimal solution to this problem, if one exists, will yield an x-vector of zero's and 1's such that a 1 in the k^{th} element of x implies P_k is in the partition and a zero in the k^{th} position implies P_k is not in the partition. This problem can be rewritten in vector notation as

min c'x

subject to Ax = 1

where

1 is a column vector of 1's $c = (c_1, c_2, \dots, c_n)$ $x = (x_1, x_2, \dots, x_n) x_i$ as above $A = (a_{ij})_{mxn} a_{ij}$ as above.

Note that A has m rows an n columns. The k^{th} column of A corresponds to P_k . The mth row of column k has a l in it if m is in P_k and is zero otherwise.

6.2 <u>Applying Set Partitioning to the Scenario</u> <u>Selection - Constraints</u>

To see that the scenario selection problem can be cast as a 0-1 set partitioning problem let:

 $K = \{K_{1}, K_{2}, \dots, K_{n_{1}}, K_{n_{1}+1}, \dots, K_{n_{1}+n_{2}}, K_{n_{1}+n_{2}+1}, \dots, K_{n_{1}+n_{2}+1}, \dots$

where K₁,..., K_n represent the n₁ clusters associated with sensor 1



associated with sensor 2 represent the n_n clusters

associated with sensor n;

represent the n₂ clusters

Then a particular track scenario can be represented by

 $\mathbf{T^{*}} = \mathbf{T}_{\mathbf{j}_{1}} \quad \mathbf{U} \mathbf{T}_{\mathbf{j}_{2}} \quad \mathbf{U} \dots \quad \mathbf{U} \mathbf{T}_{\mathbf{j}_{k}}$

with cost

T_j

 $n_1+n_2+...+n_n$ $\sum_{j=1}^{n} c_j x_j = c_{j_1}+c_{j_2}+...+c_{j_k}$

where

×i

c i

: ; ; ;

à

6

F

a binary variable such that

 $x_{j} = \begin{cases} 1 & \text{if track j is in the scenario} \\ 0 & \text{if track j is not in the} \end{cases}$

scenario j

the cost of track j.

Forming the complement set $T_{j_{k+1}}$ as

 $T_{j_{1+1}} = K - T \star$

we can write K as the union of the T_{i} 's for any track scenario, i.e.

 $K = T_j U T_j U \dots U T_j_k U T_{j_{k+1}}.$

Lastly, if we allow a particular cluster to appear in one and only one of the $T_{j_{*}}$'s for a particular scenario we have that

 $j_i \neq j_k$ implies $T_{j_i} \cap T_{j_k} = \emptyset$.

Therefore, for any track scenario, the T_j 's form a partition of K, the set of all clusters. If each cluster in the complement set T_j is considered to be a potential track made up of a singleton cluster with zero cost, then the problem may be written as:

min c'x subject to

Ax + IS = 1

where

с	=	(c ₁ , c ₂ ,, c _n) cost vector
x	z	(x_1, x_2, \ldots, x_n)
×i	-	<pre>{1 if track T_i is in scenario 0 if track T_i is not in scenario</pre>
A	*	(a _{ij}) _{mxn}
^a ij	=	$\begin{cases} 1 \text{ if cluster } K_i \text{ is in track } T_j \\ 0 \text{ if cluster } K_i \text{ is not in track } T_j \end{cases}$
I	æ	m x m identity matrix
S	*	(s ₁ , s ₂ ,, s _m) vector of slack variables having zero cost
si	=	<pre>{1 if cluster K_i is in complement set {0 otherwise</pre>
1	æ	column vector of all ones.
	400	lying Set Partitioning to Scenario Selection -

6.3 <u>Applying Set Partitioning to Scenario Selection -</u> <u>Cost Function</u>

This section derives the cost function used in the IP formulation of the scenario selection problem. As explained in previous reports (References 2 and 3), Tracor's HTA uses an extended Kalman filter to sequentially track a given target. The filter is initialized by a batch, least-squares algorithm that uses a constant acceleration

motion model for target trajectories. Two measurement types, bearing and Doppler shifted frequency, are used by the measurement model.

The measurement model of the Kalman filter produces a residual stream as data are processed,

 $e_i = Y_i - Y_i$

where Y_i - observed measurement value

Y_i - predicted measurement value.

The e_i 's are assumed to be independent, normal random variables with zero mean and variance w_e . Thus for a particular track, the likelihood function for the measurement model residual stream can be computed. For track T_i possessing n_i observations, this function becomes:

$$L_{i} = \sqrt{\frac{e^{-\frac{1}{2}} (e_{1}/w_{1})^{2}}{\sqrt{2\pi w_{1}}}} \sqrt{\frac{e^{-\frac{1}{2}} (e_{2}/w_{2})^{2}}{\sqrt{2\pi w_{2}}}} \cdots \sqrt{\frac{e^{-\frac{1}{2}} (e_{n_{i}}/w_{n_{i}})^{2}}{\sqrt{2\pi w_{n_{i}}}}}$$
$$= \frac{e^{-\frac{1}{2}} (e_{1}/w_{1})^{2}}{(2\pi)^{n_{i}/2} (w_{1}, w_{2} \dots w_{n_{i}})^{\frac{1}{2}}} e^{-\frac{1}{2}} \sum_{i=1}^{n_{i}} (e_{i}/w_{i})^{2}} \cdots \sqrt{\frac{e^{-\frac{1}{2}} (e_{n_{i}}/w_{n_{i}})^{2}}{\sqrt{2\pi w_{n_{i}}}}}$$

Let $S_j = \{T_i, T_i, \dots, T_i\}$ be a track scenario, then the likelthood of S_j is defined as:

$$L(S_i) = L_{i_1} \quad L_{i_2} \cdots \quad L_{i_n}$$

where L_{i_k} is the likelihood function for the measurement

model residual stream corresponding to track T_i . If x_i is a variable such that

$$x_i = \begin{cases} 1 \text{ if } T_i \text{ is in scenario i} \\ 0 \text{ otherwise,} \end{cases}$$

then the above can be written as

$$L(s_1) = L(S) = L_1^{x_1} L_2^{x_2} \dots L_n^{x_n}$$

where $x_{i_1}=1$, $x_{i_2}=1$, ..., $x_{i_n}=1$ and all other $x_i=0$.

In general then,

$$L(S) = L_1^{x_1} * L_2^{x_2} * \dots * L_n^{x_n}$$

and a reasonable scenario selection candidate would be that set of x_i 's which maximizes L(S). This is equivalent to finding that set of x_i 's which minimizes

$$-2 \ln L(S) = x_1 (-2 \ln L_1) + x_2 (-2 \ln L_2) + \dots + x_n (-2 \ln L_n) .$$

Thus, for the scenario selection problem,

$$c_{i} = -2$$
 ln $L_{i} = n_{i}$ ln $2\pi + \sum_{i=1}^{n_{i}} W_{i} + \sum_{i=1}^{n_{i}} (e_{i}/W_{i})^{2}$

becomes the cost associated with track T_i , and the optimum scenario is the one that maximizes the scenario likelihood function.

6.4

à

Integer Programming Summary

In summary, it is important to keep in mind the following aspects of the MTTA integer programming formulation. For the given scenario under consideration the members of the component tracks, T₁, T_j, are determined T₁₂,..., by the initial guess routine, the members of the complement Tj_{k+} , are the clusters that are "left over." The set, cost c_i^{K+1} of a particular track T_i is determined by applying the HTA to the data contained in the clusters which comprise T_i.

The solution algorithm is a list based search procedure described in Garfinkel and Nemhauser (Reference 17) and, while there are potentially faster algorithms available, this formulation was felt to be sufficient at the present time. Larger problems will likely force consideration of faster solution algorithms.

Finally, it should be noted that Morefield (Reference 8) has also used integer programming to solve the intersensor correlation problem. He used a set packing algorithm which, with the addition of slack variables, is equivalent to the set partitioning algorithm. However, his rationale for casting the problem in this framework is substantially different, using a Baysian approach to track formation and his cost function is slightly different. More importantly, he uses data types which ignore the passive initialization problem and he works with substantially smaller data sets. Because of this Morefield used neither clustering techniques or initial guess procedures to prune the data, relying on the IP to handle these chores.

n vers

R

Ś

2.5

Ŋ

5

2

7.0 RESULTS

7.1 Introduction

This section presents the results of applying Tracor's MTTA to several different simulated multi-target scenarios. Topics discussed in this section include:

- 1) Generation of simulated multi-target data,
- A three target scenario with good geometry, strong signal, and a threshold of 0 dB in a l Hz band,
- 3) A three target scenario with poor geometry, strong signal, and a threshold of 0 dB in a 1 Hz band,
- 4) A three target scenario with good geometry, moderate signal, and no threshold, and
- 5) A two target scenario with good geometry strong signal, and a threshold of 0 dB in a 1 Hz band.

The section corresponding to topic (2) contains a detailed discussion that illustrates the performance of the MTTA. Figures and tables are presented which show the results of the data clustering, automatic cluster extraction, initial guess estimation, and integer programming modules.

All scenarios generated were single trial runs, thus the results presented are not Monte Carlo averages and

reflect more variation than do typical Monte Carlo results. All scenarios were of six minute duration and employed three sensors to gather data. Computing was performed on a UNIVAC 1100/61 where the run times per scenario ranged from seven to nine minutes depending on the scenario complexity and the received signal strength. This point should be noted, Tracor's MTTA is currently running in near real time. With only a minimal amount of parallel processing, i.e., separate processors to cluster each buoy's data and two processors to evaluate potential tracks for the IP module, the algorithm could easily be made to run in real time. Because the MTTA uses relatively little storage space (less than 64 K words for the clustering algorithm and less than 30 K words for all other modules) and requires only six to seven significant digits of numerical accuracy, current parallel processing options would be more than adequate for MTTA computations.

7.2 Scenario Data Generation

To study Tracor's MTTA tracking capabilities, multi-target data were needed to test the tracker. Since no real multiple target data sets were readily available, simulated multiple target data were generated to fill this need. Following is a brief discussion of the type of sonobuoys modeled, the type of trajectories used by the targets, and the target signal strength levels that were used to simulate the multi-target data.

7.2.1 Sensor Locations

For all the geometries investigated, three non-moving passive sensors were used to gather data from the targets. These sensors were deployed in an equilateral,

triangular arrangement that is known as a tri-tac pattern. The baseline distance between each of the sonobuoys was fixed at 7000 m. This arrangement was felt to be sufficient to insure that each of the targets could be observed by all the sensors.

7.2.2 Target Trajectories

Three different multi-target geometries were simulated for the current study. For each geometry, all the targets followed unique, constant velocity, constant heading trajectories for six minute durations. These trajectories were specifically designed to prevent track intersections during this six minute time span. The three geometries used are summarized in Tables 7.I-III. These scenarios were used to determine if certain geometric and dynamic combinations could be formed that might prevent the MTTA from tracking all the targets. As will be discussed, the scenarios described in Tables 7.I and 7.III proved to be favorable observation geometries whereas the scenario described in Table 7.II proved to be an unfavorable multi-target geometry with the MTTA failing to track one of the three targets.

7.2.3 Simulated DIFAR Multi-Target Data

After the three scenarios described above were created, simulated sonobuoy measurement data for the targets were needed so that the MTTA could attempt to reconstruct the tracks for all the targets. To generate the data for individual targets, a DIFAR simulator (References 2 and 3) was used to generate frequency and bearing measurements. This simulator used a peak picking method to estimate the frequency from a normalized frequency spectrum and an arctangent processor to estimate the bearing. For this simulation, a data

TABLE 7.I

DESCRIPTION OF THE 3 TARGET TRACKS FOR SCENARIO 1

Buoy Positions

Buoy #	X(m)	Y(m)	V(m/s)
1	-3500	1000	0
2	0	7062	0
3	3500	1000	0

Initial Target Tracks (t_o=0s.)

Target #	X _o (m)	Y _o (m)	V _o (m/s)	Course Heading (°)
1	-3500	-750	. 6	45
2	2500	500	9	90
3	-500	4500	4	300

Final Target Tracks (t_f=360s)

Target #	X _f (m)	Y _f (m)	V _f (m/s)	Course Heading (°)
1	-1473	777	6	45
2	2500	3740	9	90
3	220	3253	4	300

1

•

. .

З У

> с Эч

TABLE 7.II

DESCRIPTION OF THE 3 TARGET TRACKS FOR SCENARIO 2

Buoy Positions

Buoy #	X(m)	Y(m)	V(m/s)
1	-3500	1000	0
2	0	7062	0
3	3500	1000	0

Initial Target Tracks (t_o=0s.)

Target #	X _o (m)	Y _o (m)	V _o (m/s)	Course Heading (°)
1	-3000	-750	6	15
2	2500	500	9	90
3	-500	4500	4	300

Final Target Tracks (t_f=360s)

Target #	X _o (m)	Y _o (m)	V _o (m/s)	Course Heading (°)
1	-914	-191	6	15
2	2500	3740	9	90
3	220	3253	4	300

TABLE 7.III

DESCRIPTION OF THE 2 TARGET TRACK FOR SCENARIO 4

Buoy Positions

Buoy #	X(m)	Y(m)	V(m/s)
1	-3500	1000	0
2	0	7062	0
3	3500	1000	0

Initial Target Tracks (t_o=0s.)

Target #	X _o (m)	Y _o (m)	V _o (m/s)	Course Heading (°)
1	-1000	-500	6	0
2	-3000	7000	6	300

Final Target Tracks (t_f=360s)

Target 👭	X _f (m)	Y _f (m)	V _f (m/s)	Course Heading (°
1	1160	-500	6	0
2	-1920	5129	6	300

П

た合

Ì

Ē

update rate of 10 sec. was used to make the measurement, and since the accuracy of the frequency estimates is inversely proportional to the update rate, the frequency measurements were limited to a 0.1 Hz resolution. As described before (References 2 and 3), this data generation program is a fairly realistic model which outputs non-Gaussian measurements for the individual targets.

To form the multi-target data needed for the current investigation, individual target data sets were generated for each of the targets present in a given scenario. Then, the individual target data sets were merged for each of the sonobuoys to produce what was referred to last year as multiple linetracker data (Reference 3). Certain weaknesses were acknowledged previously in simulating the data in this fashion, but overall, this type of multiple target data was felt to be more than adequate for testing the MTTA.

For all but one case, each target transmitted a tone whose SNR level as measured 1 yard from the source was simulated as 80 dB in a 1 Hz band. For these cases, the DIFAR simulator used a threshold of 0 dB in a 1 Hz band that had to be met or exceeded before any measurement estimates were If one uses a 20 log R approximation for the output. propagation loss, where R is the magnitude of the distance from the source to the receiver, an 80 dB source level without any fluctuation could be heard 10,000 yards away and just meet the 0 dB threshold criterion. However, the DIFAR simulator introduces some random noise terms to model random fluctuations in both the target signal source level and the ambient noise level, so the absolute maximum observation range for the sonobuoys cannot be set.

One other case was simulated which used a lower SNR value for the target source level, but eliminated the thresholding criterion so that measurement estimates were made for all time updates. Usually, threshold levels are set so that most target measurements exceed the level but no, or almost no, random noise measurements are introduced into the Unfortunately, such thresholding techniques can data set. eliminate actual measurements in order to insure that no noise is included in the data. For this particular case, the threshold was dropped to see if the sensors could pick up a weaker signal than that used in the other simulations. The target source level was lowered by 3 dB to 77 dB in a 1 Hz This 3 dB loss effectively decreased by approximately band. 30% the range at which the signal could be heard. To compensate for this loss in range of the detection system, the threshold was eliminated so that measurement updates were always made. The clustering algorithms were then used to pick the true measurements from the random noise found in the data. The results from eliminating the measurement threshold and then clustering the data to separate true target data from random noise are presented in a subsequent subsection.

7.3 Three Target Scenario, Good Geometry, Strong Signal

7.3.1 Introduction

This section contains a discussion of the results obtained by applying the MTTA to a three target scenario with good observation geometry in a low noise environment. It is very likely that no multi-target algorithm will be able to handle all possible observation geometries, so this report

contains examples of the MTTA's performance for both good and bad observation geometries. Furthermore, real ocean ambient noise levels and target source levels vary drastically from one situation to the next, so detected SNR's vary greatly for different encounters. This variation in detected SNR's seriously affects the quality of the measurements which in turn greatly affects tracking accuracy. By running a three target scenario under various degrees of signal degradation, it was hoped that a better idea of algorithm performance could be obtained.

Figure 7.1 contains a plot of this scenario. The plus signs represent the buoys, the solid lines the actual trajectories, and the dashed lines MTTA's estimated trajectories. Detailed results of the MTTA's solutions from each module are presented next.

7.3.2 Data Clustering Results - Buoys 1 and 3

Table 7.IV contains the frequency, bearing sine, and bearing cosine measurements generated by buoy 1, while Table 7.V contains the measurements generated by buoy 3 of the scenario. The column headed 'TARGET' indicates whether the measurement came from target 1, target 2, or target 3.

Tables 7.VI and 7.VII present the clusters generated by Ling's algorithm (for buoys 1 and 3, respectively), the associated cluster level or node number (see section 3.2), and the object numbers for the members of each cluster. Examination of these tables illustrates several points:

HD-A1	37 564	APP PRO	LICATI GRAMMI	ONS OF Ng to	CLUS	TER AN Ple ta	ALYSES RG (L	AND	INTEGE COR IN	R C AUST	IN TX	2/2	2
UNCLAS	SIFIED) TRA	COR-T8	GIENCE 3-AU-9	-5 GRU 9552-U	N0001	4-82-0	сет ні с-0044	L. 21	F/G 1	2/1	NL	
							E.						
							Situed Seer DTIC						
							<u>س</u>						
													1.44



1. 1. 1. 1. S. 1.



88

COLORNACIONI (COLORNACIONI)

Sector.

1. 32. CO

Sec. Sec.

و المحمد من من من

A Contractor

EK

÷

સુ

ì

хяс В

53

100

Ş

TABLE 7.IV

SIMULATED MULTI-TARGET DIFAR DATA FOR

BUOY 1 OF SCENARIO 1

SAMPLE NO.	TIME	FREQUENCY	COS(β)	$SIN(\beta)$	TARGET NO.
1	15.0000	150.000	.308343	951275	1
. 2	15.0000	150.100	•655623	.755089	3
. 3	25.0000	149.900	.348433	937334	1
4	25.0000	150.100	.642F6C	.765954	3
5	35.0000	149.900	.391800	920050	1
6	35.0000	150.000	•97DF26	239784	2
7	35.0000	150.100	.778779	•627298	3
8	45.0000	149.900	.370085	928999	1
9	55.0000	149.900	•328874	944374	1
10	55.0000	150.000	•996433	643933-001	2
11	55.0000	150.100	.661244	.759171	3
12	65.0000	149.900	•44723C	894419	1
13	65.0000	150.100	•638200	•769871	3
14	75.0000	150.107	• 55 65 8 4	+830792	3
15	85.0000	149.800	. 56j186	228367	1
16	85.3030	150.900	.999073	430587-001	2
17 '	85.3000	150.100	•741068	.671430	5
18	95.0302	149.860	•529447	648343	Ţ
19	95.0000	150.000	•997932	642841-001	2
20	95.0000	150+100	.698291	.715814	2
21	105.000	149.900	·989346	-147611	2
22	115.000	149.800	.756462	654037	. 1
23	115.000	149.900	•999826	196446-001	2
24	115.000	159.100	•712585	.701586	5
25	125.000	149.800	+743048	669238	1
26	125.000	149.900	•975569	.219694	2
27	125.000	150.100	.672624	•739984	5
28	135.000	149.900	. 617491	786578	L L
29	135.330	149.900	.975051	.221983	2
30	135.000	152.100	.799014	.601313	5
31	145.000	149.800	•652499	730587	L L
32	145.000	149.900	.993101	.117262	Z
22	165.000	160 700	.471678	●_78039Z	L 1

(CONTINUED)

1220000 (N. 5.5.5)

24.4

ريمار عربة محمول محمد

A Second Second

1.22

TABLE 7. IV (CONCLUDED)

•

100

1. Z . L

ŝ

and a second second second second

SAMPLE NO.	TIME	FREQUENCY	COS(B)	SIN(B)	TARGET NO.
34	155-000	149.900	908245	.418439	2
35	165.300	149.700	-610771	791807	·ī
36	165.000	149.900	.999421	.340218-001	2
37	175.000	149.730	.771946	712233	1
38	175.000	149.900	.987672	.156538	2
39	175.000	150.100	.819604	•572931	3
43	185.000	149.700	•689152	724617	1
41	185.000	150.100	.777887	.628405	3
42	195.000	149.900	.999920	126506-001	2
43	195.000	159.100	.808951	.587876	3
44	205.000	149.700	.745327	666699	1
45	205.000	149.800	.973447	•228911	2
46	205.030	150.100	.797952	•602721	3
47	215.330	149.700	.761986	647593	1
48	215.000	149.800	.963391	•268399	2
49	215.000	150.100	.790408	•612580	3
50	225.000	149.700	.777883	628409	1
51	235.000	149.700	•792367	610045	1
52	235.000	149.500	.935300	.353854	2
53	235.000	150.100	.759726	•650244	3
54	245.000	149.600	•61772C	575616	1
5,5	245.000	150.000	.696281	•717769	3
56	Z55.000	149.600	.845389	534151	1
57	255.000	150.000	.813719	.581259	3
58	Z65.000	149.600	.81762 C	575758	1
59	265.000	152.000	.775536	.671704	3
60	275.000	149.600	.856447	495783	1
61	295.303	149.632	.863743	503933	1
6Z	295.000	149.700	•917841	.396947	2
63	295.000	150.000	.79643 <u>c</u>	.634730	3
64	305.303	149.600	.654648	466260	1
65	315.300	149.607	.872470	488667	Ţ
66	325.200	149.600	.841562	543163	1
67	323.JUU	149.700	•948587	•316532	2
68	325.000	150.000	.821891	.569646	3
6 7	333.000	147.600	.894065	447937	12
70	3330UUU 776 300	149.700	•756681	• 2 3 3 7 8 2	2
71	3330JUU	120.000	.771498	•030231	3
12	3436JUU 745 330	149.600	.978669	41/515	± _
13	37 30JUU 768 000	190.000	• 911#35	• 4]] 4 4 4	5
74	333.300	147.500	.919773	392950	⊥
75	222.002	193•306	.812330	•283198	3

27.73

73

19 m

202

MACH

A Constant

و و مرد المالي

TABLE 7.V

SIMULATED MULTI-TARGET DIFAR DATA FOR BUOY 3 OF SCENARIO 1

SAMPLE NO.	TIME	FREQUENCY	COS(3)	SIN(_b)	TARGET NO.
1	5.00000	150.600	957011	290051	1
2	5.00000	158.400	881346	472472	2
3	15.0000	150.600	991340	131323	1
4	15.0000	153.300	936695	350145	2
5	25.0000	150.600	996409	.846665-001	1
6	25.0000	150.200	969837	243756	2
7	25.0000	150.300	790688	.612219	3
8	35.0000	150.600	988933	148360	1
9.	35.0000	150.100	988085	153907	2
16	45.0000	153.607	938766	344555	1
11	45.6000	150.100	996927	783859-001	2
12	45.0000	150.300	707389	.706824	3
13	55.0000	150.000	999433	336628-001	2
14	55.0000	150.300	744212	.667267	3
15	65.0000	150.000	998722	.505450-001	2
16	65.0000	150.300	720066	•693885	3
17	75.3030	149.903	998110	.153750	2
18	75.3000	157.300	797583	•61620°	3
19	85.0000	150.600	934025	357209	1
20	85.0000	149.800	969759	.244063	2
21	95.0000	149.800	959518	.281646	2
22	105.300	150.600	985134	171788	1
23	105.000	149.600	922599	.385761	2
24	115.000	149.600	866289	•499543	2
25	115.000	157.300	815628	•578576	3
26	125.000	150+600	820882	571098	1
27	125.000	149.600	840743	.541434	2
28	135.000	150.600	956779	255615	1
29	135.000	149.500	792301	•612719	2
30	145.000	153.300	669386	.742915	3
31	155.000	153.600	896914	442408	1
32	155.030	149.500	77783	•628533	2
33	165.000	160.700	731440	-681906	2

(CONTINUED)

SAMPLE

NQ.

TIME

TABLE 7.V (CONTINUED)

TARGET $SIN(\beta)$ $COS(\beta)$ FREQUENCY -.969606 -.244673 150.600 165.300 -.727551 .686053 149.400 72600

ЗĊ. Ξ. . ----

NO.

			_		•
7.	165.000	150.600	969606	244673	1
34	165.000	149.400	727551	•686 <u>0</u> 53	2
22	165.000	150.300	797280	•603609	3
30	175 000	153.600	915253	402879	1
57	175.000	149.800	738751	•673978	2
28	175-000	150.300	645465	•763794	3
39	1/5.000	159 400	961503	274796	1
40	185.000	100 000	671320	.799008	2
41	105.000	1449400	718734	•695316	3
42	165.000	100.200	631385	.775470	2
43	195.330	149.300	712429	.701744	3
44 .	195.000	150.300		.797404	2
45	265.000	147.300	847868	-530207	3
46	205.000	120+200	959561	- 281502	ī
47	215.000	153.600		.797110	2
48	215.000	149.300	- 546741		2
49	225.000	149.300	- 077047	• • • • • • • • • • • • • • • • • • •	3
50	225.000	150.300	- 076779	- 216070	ĩ
51	235.000	150.600	- 577599	- 445777	2
52	235.000	149.300	- 706754	700000	2
53	235.300	159.300		- 10000m	1
54	245.000	150.600			2
55	245.330	150.307	-+70014-	•037074	2
56	255.000	149.300	492151	•870510	2
57	255.300	150.300	735882	•679277	2
58	265.000	150.600	948240	-+31/555	1
50	265.030	157.300	776°10	•627136	2
<u> </u>	275.330	149.300	453732	•855976	2
41	265.000	149.300	404698	.914451	Z
4.2	285-000	150.300	741406	•623521	3
94 47	295-000	149.300	469664	•882845	2
03	265.000	150.300	750253	•661151	3
04	365.000	150.600	996470	•5529 <u>07</u> -001	1
03	305-000	149.200	396199	• 420731	2
00	366.330	150.300	761725	•6479DC	3
0/	315.000	149.200	389568	•920995	2
66	315-3300	150.300	798016	•706196	3
69	3120000	150.607	992243	187611	1
76	775 836	157.300	862539	.505479	.3
71	36303UU Tie 830	150.600	966°61	254923	1
72	333+34V 776 838	140.200	379682	.925117	2
73	332.000	150.200	777935	•628344	3
74	335.730	190 - 300 130 - 300	435891	•900000	2
75	345.000	1479240	976276	+ 4 2 2 4 7 3	3
76	345.007	120 • 200			-

(CONTINUED)

N.

C.

141

3

ţ,

ALC:

Į

TABLE 7.V (CONCLUDED)

SAMPLE NO.	TIME	FREQUENCY	COS(B)	SIN(B)	TARGET NO.
77 78	355.000 355.000	158.600 149.200	976356 265759	216167	12
79	355.000	150.300	621143	.570723	3

a data da da d



- -

STATES STREETS STREETS STATES STREETS STREETS

Š

Š

K

60 1

1

Constant And

TABLE 7.VI (CONTINUED)

CLUSTER	NODE 20	SIZE	0,72	SJEC 28C	TS	กาว	STE	KED				
21	21	~	ا ک	18								
52	72	P	M	ŗ.	4							
23	33	~	3 7 8	9								
24	26	Ś	35	7	5	4	5 \$					
\$2	21	•.	-	S	4	œ						
2 6	- F2	•••	54	56	9 9							
21	1,8	2	64	75								
28	32	~	33	37	0 #							
53	35	~	.09	61	* 9	65	69	72	74			
8	34	*	~	\$	11	13						
31	39	10	54	56	9 9	60	61	ħ 9	65	69	72	74
32	. 14	Ņ	20	24	23							
33		ت	39	14	M Ø	9 #	6 #	53				
34	9 #	~	59	63		•						
35	61	11	5	56	8 2	60	19	64	65	66	69	12
36	5 û	2	32	36								
37	19	*	11	20	5 4	27						
36	22	m	96	16	1 9							
39	54	~	57	59	63							
04	55	~	67	10								
14	57	5	15	59	F 9	4	75					
		(CON	TINU	(ED)								

95

and the second state of the second stat
	1917, E. L					1. Ú.										••••	•••	••••	•	••••	•••		
Tracer Ap	plied S	Scier	ices																				
																				53			
																				64			
																				9			
																				* 3			15
																				1			50
																				39			1
																				30			*
																	27			27			-
																27	24			24		_	37
								15	27							2	50		_	20		52 5	35
6	-			53		51		50	2		_			28		20	[]	_	54 1	-			m
INUE	ZRED		75	6 #		50			20		30			%		1		8	2			6	
LTNO	USTE		1	4	12			;	1 17		36	-		9 33		1	-	2 2	30	-	6		Ň
Ö	CL	-	5			1	93	~	1		ii B		~	ŭ. e	3	1 1	1	2	9	1 2	9	9	2
IV 2	ECTS	2			-	÷.	Ň G		-	-	4	2 3	4 1 1	2				8	5		0	53	
ABLI	ILEO		1 5	ñ		in m	S I	5	~	m	5	2		1 2	9	~	~	15	5	~	4	55	15
H		~	w î	10		-	~			N			-		•				-				
									•		•												
	SIZE	m	•	~	s	~	ŝ	••	40	~	•	•	m	~	m	•	10	•	~	LŤ	-	•	47
	ODE	 4	 	68	12	73	74	15		8	45	16	95	68	00	52	103	113	130	134	168	198	217
	N N	-			• •			• • • • • •							-	-	-	-	-	. -	~		
	USTE	Ş	7	;	\$	4	24	4	3. #	50	51	25	53	5	55	56	57	Śå	59	9	61	62	ļ
	3																				•		

كالمناهدة

TABLE 7.VI (CONCLUDED)

1. 1.

Tracer Applied Sciences

OBJECTS CLUSTERED CLUSTER NODE SIZE

									_	
	*9		59				19		32 68	20 80
	61		57				50	70	30 67	19 39 59
	θĽ		55				2.4	61	29	16 33 58
	58		53				:	62	27 62	17 37
	8		6- #			52	40	52	26 , 55	9 20 20 20
	*5		9 #			9 #	37	8	24	15 15 15 15 15 15 15 15 15 15 15 15 15 1
	15		8 3			S #	35	\$ \$	33	1
	50		1		2 #	2 4	M M	8	2 F:	
	;		39	38	38	38	1£ 4	38	20	
	*	38	30	36	36	36	28	36	8 •	1000
		36	21	m	ŧ	7	25	*	°,	9 9 7 5 7
	37	32	24	32	32	32	6 23	32	5 1 6	0 2 V 2 0 2 0
	35	29	20	29	29	29	5 Å	58		8 ^N # 6 8
	33	26	17	26	26	26	15	26	13 13	••• •••
12	16 42	23	15	23	23	23	11 6	23		ស ហ័ ហ័ ១ ហ្វូ ឆ្នូ ឆ្
G •	28	21	13	21	21	21	•	21	01 0 1 0	5 7 7 9
•	- 52 F 9	39	11	19	61	19	4	19	- 9 M	*
	66	3	68	16	16	16	-0- -0-	16	9 7 7 8 7	M
	65 18		۴ ۲9	9	at	97	m	9	* · · · · · · · · · · · · · · · · · · ·	
	15	3	~	•		.	-	4	2	-
•	52		55	12	13	91	7	19	*	- V1
221	- 53 -	522	55 6	230	534	248	369	265	455	8 C 4
5	• •	49	67	8	6.9	10	11	12	22	74

Ň LARS STREET

TABLE 7.VII

CLUSTERING OUTPUT FOR BUOY 3

OF SCENARIO 1

OBJECTS CLUSTERE	66 68	44 74	35 38	64 67	49 52	51 54	E9 62	6u 63	f.6 68 73	55 59 62	5. 53	12 16	29 32	aj 45 48	12 14 16	2u 21	34 40	55 59 62 64 67
2 SIZE	2	2	2	~	~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~	2	2	2	£	£	2	2	2	м	5	N	2	ŝ
NODE	~	m	\$		-0	4	•	œ	10	11	12	13	14	15	16	17	14	19
LUSTER	~ ~	~		S	ه. ۲	ţ	. 1 0	0	10	11	12	13	14	15	16	E 1	18	61

1 - - J

.

K. L.

-

(CONTINUED)

Tracer Applied Sciences

ī **V** 20 K.C.C. EX. ŝ કુ ,¥ S.

TABLE 7.VII (CONTINUED)

Tracor Applied Sciences

																	~			\$			
																	61			61			
																	*		-j	*			
											-						•		÷	ت د			
											-						62		61	9			
										61	67						59		[g	5			
G										•9	9					63	57		9 j	57		54	
TER										62	62		63			61	55		52	55		15	
LUS'										65	56		60		c/	6.0	53		5- 4-	53	16	-	
ច	23		0			•				1	19		56		2	3	7		9	*	3	S	~
E E	-	-	-	_	-	~	_	~	_			~	-	•	-		3 0.	-	3		_	3	ED
Э Г	9	*	ñ	N	-	*	5	5	m	5	ŝ		6	ñ	ę	5	3	4	3	4	-	ň	INU
BO	56	34	28	54	12	33	53	9 #	31	53	5	71	3	36	60	7	33	1.	ر ه	55	12	2.	UNT
	•.																						<u>ಲ</u>
2E	m	m	æ	2	~	m	~	~	~	~	8	m	ŝ	2	*	÷	11	~	•	12		•	
SI																							
ы																							
NOD	20	25	23	25	27	28	6	3.0	31	32	33	35	30	37	30	0 #	4	42	43	.) \$	54	56	
TER																							
USJ	20	21	52	23	54	25	26	21	28	29	30	ţ	54 19	33	¥ M	. 5	36	57	35	39	3	:	
5																							

*

State I was

Tracor Applied Sciences

TABLE 7.VII (CONTINUED)

LUSTER	NODE	SIZE	0	IL B(SCTS	U S	TSU,	EREI	۵													
42	57	~	*	11																		
43	54	13	3.5	2	37 #	53	55	51	59	62	19	61	63	74	79							
**	. 19	tu:	12	*	16	18	25	••														
•	- - -0	æ	-	12	*	16	18	25														
ی. ج	61	~	13	15																		
[+	11	ñ	-	10	19																	
•	12	51	30	33	39	1 2	-	23	55	15	59	62	¥ 9	67	69	74	61					
64	14	-	~	12	14	lo	18	25	36													
20	75	~	28	\$ M	•	4.7	51	* 5	58													
15	74	9 1	30	33	39	42	3 4	53	55	57	59	79	† 9	67	5 9	11	74	79				
52	83	81	۲ ۲	33	51	42	3	9#	50	53	55	57	59	79	9	67	69	11	74 7	ø		
53	9 S		22	28	34	04	24	51	5	58												
\$. S	6	19	30	33	39	N #	2 7	9	50	53	55	15	59	62	4 9	67	69	1	74 7	6 7	•	
SS	2 7	26	ہ م ق	12	14	۱ ^ا	18	25 6 7	0£ 6	33	36	39	24	7 7		Su	53		57 5	9 6	5 5	
56	101	10	~	13	22	28	34	04		15	54	58										
57	117	L	ę Y	68	73	75	76															
Ĵ.	121	m	23	24	27																	
59	127	13	-	10	22	24	34	() 1	24	15	54	58	50	12	"							
6.0	133	16	-	m	90	0	19	22	28	đ.	3	47	51	54	5 e	52	72	"				
61	135	18	-	m	æ	10	19	22	9 2	31	34	37	د #	4 7	51	54	2 0	02	12 1	~		
62	171	. 61	-	m	39	10	19	22	26	28	31	34	37	7	7	51	54	28	1 1	2	~	

150.

ATTENDED AND いいとうというと R 33

> . Ş

Marcheneric

TABLE 7.VII (CONCLUDED)

Tracor Applied Sciences

ч IIСФЕР	NODE	CT 7F	aC	ر 11	U L		CTF C	U a a														
1001EN	195	1710	ų č	38.0	27																	
9	25 4	12	35	36		5	5	80 #	6 4	52	56	60	61	63								
65	255	11	35	38	7	n #	9	8	6. Ŧ	52	56	60	61	63	99	64	13	75	76			
9	264	•	3	11	۲ 1	15																
67	270	61	53	32	35	38	1	# 3	5 #	88	64	52	56	6.0	61	63	99	68	73	75	78	
2	281	50	-	м	ŝ	8	10	19	22	26	28	31	*	37	4	~ *	51	54	56	70	12	
69	284	52	23 75	24	27	29	32	35	38	1.	5 A	10 #	2 #	6 4	52	56	D D	61	63	66	68	
5.	167	¥î	•	7	11	13.	15															
11	298	~	11	20	51																	
77	310	•	٩	•	11	13	15	17	21	11												
11	317	•	3	٥	o	11	13	15	1.7	20	21											
7	345	10	2	8	٩	9	11	13	15	11	20	21										
75	356	77	ل 1	m	ŝ	8	10	15	22	26	26	31	34	37	3#	[#	51	24	5.6	e e	01	_
9	587	32	~	*	ي و	2 7 7	و 11 و	13 0 6	15	17 3 6	é 6	21 8 7.	23	24 5 71	5.3	56	32	35	36	1 #	* *	*
12	6 () 6	53	27 28 58	6 5 F	~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~	4 ¹⁰ 9	ور س م	وي بي م	ۍ سر ۲۵	6 <u>5</u> 6	10	2	~*~ ~~~~	15	~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~	19	20	21 21	22	23	24	~ _
9. 1.	784	61	51 51 7 7 7 7 7	0 10 10 0 2 2 0 0	M N # 4	8 8 8 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9 9	ມ ຊ ເງ ພ ຊ ຊ ຊ	୍ର ଜାଇ-ଜ ଜାଇ-ଜ	~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~		0 ° * * 0 1 0 0 0	0 0 0 0 0 0 0 0 0 0	- 10 m	122		3 10 m	15 55 75	16 36 75 97	12	9 ° ° ° ° ° ° ° ° ° ° ° ° ° ° ° ° ° ° °	6 5 6	· · · · · ·
			>	5	, ,	1	•	•	•		2	•	•								•	

101

- 1) The hierarchical nature of the data organization imposed by the algorithm;
- The varying lengths of time that clusters remain isolated before becoming subsets of other larger clusters; and
- 3) All clusters are eventually merged to form a cluster containing the whole set of data.

Note, that for buoy 3, cluster 75 contains exclusively all the points generated by target 1, cluster 76 does the same for target 2, and cluster 55 contains all the points associated with target 3. Similarly for buoy 1, cluster 71 contains only target 1 data, cluster 72 contains target 2 data, and cluster 67 contains all the data from target 3. Thus, the clustering algorithm has clearly separated the target data, now the MTTA must seek to automatically identify those clusters which are most likely to contain the desired individual target data.

Tables 7.VIII and 7.IX contain the isolation indices and survival function values for all of the clusters generated (see sections 3.2 and 3.3). Keeping only those clusters whose survival function values are less than 10^{-4} , it can be seen that not only do all the candidate clusters pass the isolation test, but so do several others. Thus, additional cluster extraction techniques are needed to pick out the appropriate data. This is discussed in the next section.

ALLENS ALLENS

大大学がたい

المراجع ليراجع

がいため

L.

12.24

3

5

3

3

TABLE 7.VIII

SURVIVAL FUNCTION AND ISOLATION INDEX VALUES FOR CLUSTERS FROM BUOY 1 OF SCENARIO 1

Ι	S	0	LA	T	Ŀ	10	Ň	IŅ	IDEX	
_	_							-		

CLUSTER	NQ.	:	1	SUPVIVAL	FUNCTION	=	-1830669-001	73
CLUSTER	N0.	=	2	SURVIVAL	FUNCTION	Ξ	.8501675	3
CLUSTER	NC.	=	3	SURVIVAL	FUNCTION	=	.1487049	35
CLUSTER	N0.	=	4	SUPVIVAL	FUNCTION	:	.2180120	28
CLUSTER	NO.	=	Ę	SUPVIVAL	FUNCTION	:	•3200111	14
CLUSTER	NO.	=	6	SURVIVAL	FUNCTION	=	•9302080-002	85
CLUSTER	NO.	=	7	SURVIVAL	FUNCTION	:	•8555628-001	45
CLUSTER	NO.	:	8	SUPVIVAL	FUNCTION	=	• 8972366	2
CLUSTER	NO.	=	9	SURVIVAL	FUNCTION	=	•4933486	13
CLUSTER	NC.	=	10	SURVIVAL	FUNCTION	=	•9218807	1
CLUSTER	N0 •	:	11	SURVIVAL	FUNCTION	=	•1112133-002	62
CLUSTER	NO.	=	12	SURVIVAL	FUNCTION	:	•7217808	6
CLUSTER	NO.	:	13	SUPVIVAL	FUNCTION	=	.8046436	4
CLUSTER	NJ.	:	14	SUPVIVAL	FUNCTION	:	.4417269	15
CLUSTER	NO.	=	15	SURVIVAL	FUNCTION	:	•2850254	23
CLUSTER	N0.	=	16	SUPVIVAL	FUNCTION	:	•8969457	2
CLUSTER	NO.	=	17	SURVIVAL	FUNCTION	=	•2293015	18
CLUSTER	NO.	=	18	SURVIVAL	FUNCTION	=	•1566442	17
CLUSTER	NO.	:	19	SURVIVAL	FUNCTION	=	.4666636	7
CLUSTER	NO.	=	20	SUPVIVAL	FUNCTION	=	•3173511	21
CLUSTER	NC.	=	Z 1	SURVIVAL	FUNCTION	:	•6106839-D02	92

(CONTINUED)

A CONTRACTOR OF A CONTRACT OF A CONTRACT

TABLE 7.VIII (CONTINUED)

ISOLATION INDEX

CLUSTER NO. =	22	SURVIVAL FUNCTION = .6644110	5
CLUSTER NO. =	23	SURVIVAL FUNCTION = .1872693-001	72
CLUSTER NO. =	24	SURVIVAL FUNCTION = .9803829-001	17
CLUSTER NO. =	25	SURVIVAL FUNCTION = .7076964-002	45
CLUSTER NO. =	26	SUPVIVAL FUNCTION = .4401400	10
CLUSTER NO. =	27	SURVIVAL FUNCTION = .2269791	27
CLUSTER NO. =	28	SURVIVAL FUNCTION = .3375559-031	41
CLUSTER NO. =	20	SUPVIVAL FUNCTION = .4659114	4
CLUSTER NO. =	30	SURVIVAL FUNCTION = .1351269-001	.39
CLUSTER NO. =	31	SURVIVAL FUNCTION = .6603140-001	10
CLUSTER NO. =	32	SUPVIVAL FUNCTION = .4384878	10
CLUSTER NO. =	33	SURVIVAL FUNCTION = .1611562-001	25
CLUSTER NO. =	34	SURVIVAL FUNCTION = .6437461	8
CLUSTER NO. =	35	SURVIVAL FUNCTION = .2384186-035	173
CLUSTER NO. =	36	SURVIVAL FUNCTION = .4368393	15
CLUSTER NO. =	37	SURVIVAL FUNCTION = .5632228-001	26
CLUSTER NO. =	38	SUPVIVAL FUNCTION = .5559623-034	116
CLUSTER NO. =	39	SUPVIVAL FUNCTION = .7801826	3
CLUSTER NO. =	4C	SURVIVAL FUNCTION = .8179311-001	45
CLUSTER NO. =	41	SURVIVAL FUNCTION = .2886278	9
CLUSTER NO. =	42	SUPVIVAL FUNCTION = .4729810	9
CLUSTER NO. =	42	SUPVIVAL FUNCTION = .2165139-004	64

(CONTINUED)

SECTOR WEARANT AND A CONTRACTOR OF A CONTRACT

Ч Ч

Ş

TABLE 7.VIII (CONTINUED)

									ISOLATION	INDEX
•	CLUSTER	NC.	=	44	SURVIVAL	FUNCTION	=	.2652407-005	- 66	• • •
	CLUSTER	NO.	=	45	SUPVIVAL	FUNCTION	:	.4619360-006	149	
×	CLUSTER	NO.		46	SURVIVAL	FUNCTION	=	•6786491	2	
	CLUSTER	NO.	:	47	SURVIVAL	FUNCTION	=	•2166149	11	
	CLUSTER	NO.	:	48	SURVIVAL	FUNCTION	:	.2687209-006	142	
	CLUSTER	N0 .	:	49	SURVIVAL	FUNCTION	=	.3830850~002	25	
2	CLUSTER	NO.	Ξ	50	SURVIVAL	FUNCTION	=	•5722709	10	
3	CLUSTER	NO.	=	51	SUPVIVAL	FUNCTION	=	•1132821	13	
Ş	. CLUSTER	NO.	=	52	SURVIVAL	FUNCTION	:	•8452478- <u>0</u> 01	22	
	CLUSTER	NO.	:	53	SURVIVAL	FUNCTION	=	•2264977-025	153	
	CLUSTER	N0.	=	54	SURVIVAL	FUNCTION	=	.3129244-306	127	
	CLUSTER	NO.	=	. 55	SURVIVAL	FUNCTION	=	.1102686-005	165	
	CLUSTER	NO.	2	56	SUPVIVAL	FUNCTION	:	.777778	1	
Ś	CLUSTER	NO.	=	57	SUPVIVAL	FUNCTION	=	.1671453-003	31	
	CLUSTER	N0.	=	58	SURVIVAL	FUNCTION	=	• 4 17 2 3 2 5 - 0 0 6	104	
10	CLUSTER	NO.	=	59	SURVIVAL	FUNCTION	=	.1415610-005	68	
	CLUSTER	N0.	:	60	SURVIVAL	FUNCTION	:	.1192093-006	92	
% ·	CLUSTER	NO•	=	61	SUPVIVAL	FUNCTION	=	.1293644-032	57	
	CLUSTER	N0.	=	62	SURVIVAL	FUNCTION	:	.1404643-302	28	
	CLUSTER	NØ.	:	63	SUPVIVAL	FUNCTION	=	•1309161	5	
	CLUSTER	NO.	:	64	SUPVIVAL	FUNCTION	=	•4649013-003	43	
-							-			

(CONTINUED)

105

the second works

1.1.1.17

TABLE 7.VIII (CONCLUDED)

ι.,

								ISOLATION	INDEX
CLUSTER	NO.	:	65	SURVIVAL	FUNCTION	=	.1341105-006	42	
CLUSTER	N0.	:	66	SURVIVAL	FUNCTION	:	.1985213	5	
CLUSTER	NO.	=	67	SURVIVAL	FUNCTION	:	.8940697-007	229	
CLUSTER	NO.	=	68	SURVIVAL	FUNCTION	:	.2439254	4	
CLUSTER	NO.	=	69	SURVIVAL	FUNCTION	:	.4708424-002	14	
CLUSTER	NO.	:	70	SURVIVAL	FUNCTION	=	.3411472-073	17	
CLUSTER	NO.	:	71	SURVIVAL	FUNCTION	:	.1843081-806	600	
CLUSTER	NO.	:	72	SURVIVAL	FUNCTION	:	.1043091-006	190	
		•	78	CHEVTVAL	FILMETTON	-	-1043081-006	409	
CLUSIEX	NV +	-	13	304 4 T A V C	1 0110 1 104	-			

<u> Antonio de Constante de Const</u>

4

832 1

-24242424242

3

2

 इन्द्रम् म् न

. . . .

TABLE 7.IX

SURVIVAL FUNCTION AND ISOLATION INDEX VALUES FOR CLUSTERS FROM BUOY 3 OF SCENARIO 1

								ISOLATION	INDEX
CLUSTER	NO.	Ξ	1	SUFVIVAL	FUNCTION	:	•4869150	14	
CLUSTER	NO.	=	2	SURVIVAL	FUNCTION	Ξ	•6630117	8	
CLUSTER	NO+	=	_ 3	SURVIVAL	FUNCTION	=	.2757257	25	
CLUSTER	NO.	1	4	SUPVIVAL	FUNCTION	=	•4042685 - 094	191	
CLUSTER	N0.	=	5	SUPVIVAL	FUNCTION	=	.4864466	14	
CLUSTER	NO.	=	6	SURVIVAL	FUNCTION	=	.2124872	30	
CLUSTER	N0.	=	7	SUPVIVAL	FUNCTION	=	.7896106-001	49	
CLUSTER	NO.	=	P	SUPVIVAL	FUNCTION	Ξ	.8570225	3	
CLUSTER	NO.	=	9	SURVIVAL	FUNCTION	:	.5674055	11	
CLUSTER	N0.	=	10	SUPVIVAL	FUNCTION	Ξ	.1141836	28	
CLUSTER	NO.	3	11	SUPVIVAL	FUNCTION	:	.5389733	8	
CLUSTER	N0.	:	12	SURVIVAL	FUNCTION	Ξ	.6948277- 092	95	
CLUSTER	NO.	:	13	SUPVIVAL	FUNCTION	=	.8568013	3	
CLUSTER	N0.	=	14	SURVIVAL	FUNCTION	=	•1609 325- 005	262	
CLUSTER	N0+	2	15	SUPVIVAL	FUNCTION	:	.1137623	28	

(CONTINUED)

107

5

B

TABLE 7. LX (CONTINUED)

						ISOLATION	INDEX
CLUSTER NO. =	16	SURVIVAL	FUNCTION	:	•5202636-071	38	
CLUSTER NO. =	17	SURVIVAL	FUNCTION	=	•1236796-005	281	
CLUSTER NO. =	18	SURVIVAL	FUNCTION	=	·813¤705	4	
CLUSTER NO. =	19	SUPVIVAL	FUNCTION	=	-1868028	13	
CLUSTER NO. =	20	SUPVIVAL	FUNCTION	=	•2989062	16	
CLUSTER NO. =	21	SURVIVAL	FUNCTION	Ξ	•9254659	1	
CLUSTER NO. =	22	SUPVIVAL	FUNCTION	=	•3250551 - 001	33	
CLUSTER NO. =	23	SUPVIVAL	FUNCTION	=	.6443247-302	96	
CLUSTER NO. =	24	SURVIVAL	FUNCTION	=	•6607238	8	
CLUSTER NO. =	25	SURVIVAL	FUNCTION	=	•3638257	13	
CLUSTER NO. =	26	SUPVIVAL	FUNCTION	=	.556C889	3	
CLUSTER NO. =	27	SURVIVAL	FUNCTION	=	•6270218-001	53	
CLUSTER NO. =	28	SUPVIVAL	FUNCTION	=	•4153132-002	104	
CLUSTER NO. =	2 <u>9</u>	SURVIVAL	FUNCTION	=	•8346999	1	
CLUSTER NO. =	30	SURVIVAL	FUNCTION	=	•1916807	8	
CLUSTER NO. =	31	SURVIVAL	FUNCTION	=	.6954670-303	92	
CLUSTER NO. =	32	SUPVIVAL	FUNCTION	=	.5954256	4	
CLUSTER NO. =	33	SURVIVAL	FUNCTION	=	.1608063	35	
CLUSTER NO. =	34	SUPVIVAL	FUNCTION	=	-2457201-003	79	
CLUSTER NO. =	3 <u>=</u>	SUPVIVAL	FUNCTION	=	•6277485	3	
CLUSTER NO. =	36	SURVIVAL	FUNCTION	=	.3229120	4	
CLUSTER NC. =	37	SURVIVAL	FUNCTION	:	.2197462	29	

(CONTINUED)

and the second

S. C. S.

2

ļ

L.T.L

TABLE 7.IX (CONTINUED)

ž										
									VARIABLES	CLUSTERED
	CLUSTER	N0 .	=	38	SUFVIVAL	FUNCTION.	=	•298 <u>0</u> 232-006	20	7
	CLUSTER	NO.	=	39	SUPVIVAL	FUNCTION	=	.1332875-001	14	4
	CLUSTER	NO.	Ξ	40	SURVIVAL	FUNCTION	=	.4812570		7
]	CLUSTER	NO.	=	41	SURVIVAL	FUNCTION	:	.5071954-001	19	9
Ŭ	CLUSTER	N0.	=	42	SURVIVAL	FUNCTION	:	.1750886-004	20	3
3	CLUSTER	NO.	Ŧ	43	SURVIVAL	FUNCTION	:	.1288421-001	1	3
~	CLUSTER	N0.	=	44	SURVIVAL	FUNCTION	=	.6755485	:	3
Ŕ	CLUSTER	N0.	=	45	SUPVIVAL	FUNCTION	=	.2078058	1	D
Я	CLUSTER	NO.	=	46	SURVIVAL	FUNCTION	=	.2908707-004	19.	3
러	CLUSTER	N0.	=	47	SURVIVAL	FUNCTION	=	.7183716-092	6	2
	CLUSTER	NO.	=	48	SURVIVAL	FUNCTION	=	.9947246-001		6
	CLUSTER	NO.	=	49	SURVIVAL	FUNCTION	=	.3642718-001	1	8
5	CIUSTER	NO.	=	50	SURVIVAL	FUNCTION	:	.1591002	1	o [:]
	CLUSTER	NO.	Ξ	51	SUPVIVAL	FUNCTION	=	.1291833		5
Д,	CLUSTER	NO.	=	52	SURVIVAL	FUNCTION	=	.2543985		3
	CLUSTER	NO.	=	57	SUPVIVAL	FUNCTION	=	.9631157-002	2	2
13	CLUSTER	NO.	=	54	SUPVIVAL	FUNCTION	=	•5627866-001	•	6
3	CINSTER	NO.	Ξ	55	SUFVIVAL	FUNCTION	=	.1341105-006	69	2
3	CI UCTED		=	54	SUPVIVAL	FUNCTION	:	.49966*7-002	2	0
	ULU3 I E K		-						• •	C
7	CLUSTER	N0.	=	57	SUPVIVAL	FUNCTION	=	.5066395-006	, 13	0
31	CLUSTER	NO.	:	58	SUPVIVAL	FUNCTION	:	.1639178-035	5 16	7

(CONTINUED)

109

Alexandre alex

TABLE 7.IX (CONCLUDED)

								ISOLATION	INDEX
CLUSTER	NO.	Ξ	59	SUPVIVAL	FUNCTION	:	.1273446	6	
CLUSTER	NO.	=	60	SURVIVAL	FUNCTION	=	•4329841	2	
CLUSTER	NO.	=	61	SURVIVAL	FUNCTION	:	.1490116-006	36	
CLUSTER	NO.	=	62	SURVIVAL	FUNCTION	=	.1341105-006	110	
CLUSTER	N0.	=	63	SUPVIVAL	FUNCTION	=	.1034689-001	55	
CLUSTER	NO.	:	64	SUPVIVAL	FUNCTION	Ξ	•3669095	3	
CLUSTER	NO.	=	65	SURVIVAL	FUNCTION	=	•2089143 - 004	23	
CLUSTER	NO.	:	66	SURVIVAL	FUNCTION	:	•1516947-001	37	
CLUSTER	NO.	=	67	SURVIVAL	FUNCTION	=	•1882628-002	12	
CLUSTER	N0.	=	68	SUPVIVAL	FUNCTION	=	.1192093-006	75	
CLUSTER	NO.	=	6 ¢ '	SURVIVAL	FUNCTION	=	·1043081-006	299	
CLUSTER	NO.	=	70	SURVIVAL	FUNCTION	=	+1559601	13	
CLUSTER	NO.	=	71	SUBAIAVE	FUNCTION	=	•3577756	12	
CLUSTER	NC.	=	72	SURVIVAL	FUNCTION	Ξ	-2037570	7	
CLUSTER	N0.	:	72	SURVIVAL	FUNCTION	=	•6867349-073	28	
CLUSTER	NO.	:	74	SURVIVAL	FUNCTION	=	•2533197-Ché	242	
CLUSTER	NG.	=	7£	SUPVIVAL	FUNCTION	=	.1043081-006	250	
CLUSTER	NO.	=	76	SURVIVAL	FUNCTION	:	.8940697-007	19	
CLUSTER	NO.	=	77	SURVIVAL	FUNCTION	=	.8940697-007	178	

110

7.3.3

Ę

3

3

a

<u>Cluster Extraction Results -- Buoy 3 and Buoy 1</u>

A cut-off point of 10^{-4} was used for the survival function to determine which clusters were passed to the extraction module for further processing. For buoy 1, the following cluster numbers survived this test: 35, 38, 43, 44, 45, 48, 53, 54, 55, 58, 59, 60, 65, 67, 71, 72, and 73. For buoy 3, these cluster numbers were chosen: 4, 14, 17, 38, 42, 46, 55, 57, 58, 61, 62, 65, 68, 59, 74, 75, 76, and 77.

The polarization test (see section 3.4) eliminated from further consideration clusters 38, 53, and 55 from buoy 1 and clusters 4, 14, 17, 42, 46, 58, and 74 from buoy 3. The regression test and the exclusion inclusion rules resulted in cluster 67, 71, and 72 being extracted for buoy 1 and clusters 55, 69, and 75 being extracted for buoy 2. Thus for buoy 1, the clustering and extraction processes were able to completely separate the data into three clusters containing only the measurements belonging to a given target. For buoy 3 the same processing techniques produced two of the required data clusters but not the third, selecting cluster 69 rather than 76. From Table 7.VII, it can be seen that cluster 76 contains clusters 69 and 74 as subsets. The polarization test rejected cluster 74 as a potential candidate and the regression test rejected the consistency hypothesis of 69 and 74. Thus, only the last two-thirds of the data for target 2 which was contained in cluster 69 was selected for further consideration.

7.3.4 Initial Guess Results

Table 7.X gives a listing of the clusters extracted for each buoy along with an identification of which

and the second the second descent language

TABLE 7.X

BUOY, CLUSTER, AND TARGET NUMBER CORRESPONDENCE TABLE

Buoy Number	Cluster Number	Target Number
1	1	3
	2	1
	3	2
2	1	. 3
	2	1
	3	2
3	1	3
	2	2
	3	1

Tracor Applied Sciences

target is represented by each cluster. Allowing a track to be composed of both two-cluster and three-cluster intersensor combinations, there are then

(3*3 + 3*3 + 3*3) + (3*3*3) = 54

potential target tracks. Only twelve of these potential inter-sensor combinations correspond to actual tracks. Table 7.XI lists the results from the initial guess algorithm. From this table it can be seen that:

- Of the twelve cluster combinations corresponding to actual tracks, only 1 was rejected (marked by the asterisk),
- 2) Of the 42 cluster combinations corresponding to pseudo tracks, 29 (69%) were rejected,
- 3) Five of the three-cluster combinations were not even considered because at least two of the associated two-cluster combinations had been rejected, and
- 4) From Figure 5.1, several of the pseudo estimates can be seen to correspond quite closely to the illustrated line-of-sight crossed-bearing intersections.

Once the potential track set has been pruned by the initial guess procedure, the resultant state vector estimates and their associated cluster combinations are then passed to the integer programming module. TABLE 7.XI

INITIAL GUESS AND CONSTRAINTS RESULTS FOR SCENARIO 1

		hue	u UU		99																				
<u>ې</u>	(E)	-3.6	4.07	6.0	-2.5	1.17	3.72	-3.24	3.24	9.62	-2.87	1.57	7.42	-2.05	-4.68	1.45	-4.43	8.68	11.7	-3.5	-3.24	-4.45	1.25	4.6	.92
×	E	2.05	-5.51	-7.98	-3.2	5.53	1.53	52	.52	.74	1.6	-3.3	7.33	.13	-6.72	6.42	.15	.36	3.1	1.2	54	7.29	5.4	-2.62	6.5
X		4386.	1715.	5194.	-3021.	-571.	599.	1139.	749.3	20.	4638.	3523.	-30.	.9066	6538.	-197.	1230.	762.	115.	4760	7180.	368.	3835.	1916.	57.
×	: (II)	-598.	-2613.	423.	-1999.	-2992.	-3356.	-927.	-2557.	2668.	-521.	-1086.	-4835.	-6789	-5540.	-3186.	3382.	2638.	1980.	-392.	23.	-1232.	-137.	1630.	-1337.
Accepted	Rejected	A	R	A	A	A	R	A	A	R	A	A	R	A	A	A	R	Α	R	A	Я	A	A	A	A
Buoy 3 (Cluster/	Target)	;	1	8 1	8 1	1 (1	ţ	;	1	1/3	2/2	3/1	1/3	2/2	3/1	1/3	2/2	3/1	1/3	2/2	3/1	1/3	2/2	3/1
Buoy 2 (Cluster/	Target)	1/3	2/1	3/2	1/3	2/1	3/2	1/3	2/1	3/2	!	1	1 1	1	1	1 1	1	1	1	1/3	1/3	1/3	. 2/1	2/1	2/1
Buoy l (Cluster/	Target)	1/3	1/3	1/3	2/1	2/1	2/1	3/2	3/2	*3/2	1/3	1/3	1/3	2/1	2/1	2/1	3/2	3/2	3/2	1	1	!	ł	8	!

(CONTINUED)

Fracor Applied Sciences

Ì KAS - ALLABARAT LANCOCAMANUHIKA , caf

Tracor Applied Sciences

3/2 1/3 R 1890. 1415. 12.6 3/2 2/2 A 2407. 730. 46 9.15 3/2 3/1 R 2407. 730. 46 9.15 1/3 1/3 2/2 R -347. 4626 1.59 -3.43 1/3 3/1 R -598. 4386. 2.05 -3.6 -1.43 2/1 1/3 3/1 R -598. 4386. 2.05 -3.43 2/1 3/1 R -598. 4386. 2.01 -1.43 2/1 3/1 R -432. 4635. 5.01 -1.43 3/2 2/1 3/1 R -4243. -1.162. -2.19 3/2 3/1 R -2.899. -1.162. -2.19 0.0 -2.49 3/2 3/1 R -2.613. 1162. -2.13 2.14 -0.75 3/2 3/1 R -2.693. 1715. -2.13 2.143 0.164 0.1 3/2 <th>Buoy 2 (Cluste: Target</th> <th>r/ Buoy 3 (Cluster/ Target)</th> <th>Accepted or <u>Rejected</u></th> <th>X (III)</th> <th>ү (ш)</th> <th>X.</th> <th>$\frac{Y}{(m)}$</th>	Buoy 2 (Cluste: Target	r/ Buoy 3 (Cluster/ Target)	Accepted or <u>Rejected</u>	X (III)	ү (ш)	X.	$\frac{Y}{(m)}$
3/2 $2/2$ A 2407 . 730 . 46 9.15 $3/1$ $1/3$ $1/3$ $1/3$ $1/3$ $2/2$ 1.59 -3.43 10.93 $1/3$ $2/2$ R -347 . 4626 1.59 -3.43 10.93 $1/3$ $3/1$ R -3487 . 903 . 6.34 -5.25 $2/1$ $1/3$ R -2432 . 4635 . 2.01 -1.43 $2/1$ $3/1$ R -2613 . 1715 . -5.51 4.07 $2/1$ $3/1$ R -289 . 4535 . 2.01 -1.43 $2/1$ $3/1$ R -289 . 4535 . -1.35 2.18 $3/2$ $2/2$ R -289 . 4535 . -1.93 -2.649 $1/3$ $1/3$ R -2899 . 4552 . -1.35 2.19 $3/2$ $2/2$ R -2899 . 4530 . 07 -2.5 $1/3$ $1/3$ R -2899 . 4530 . 07 -2.649 $1/3$ $1/3$ R -2899 . 4700 . -7.98 6.00 $3/2$ $1/3$ 1.162 . -2.39 6.68 1.26 $1/3$ $1/3$ R -19999 . -752 . -1.93 $2/1$ $1/3$ R -1999 . -752 . 2.19 $1/3$ $1/3$ $1/3$ -2.54 -1.93 $2/1$ $1/3$ $1/3$ -1999 -2.54 -1.93 $1/3$ $2/2$ $1/3$ 1.26 <td>3/2</td> <td>1/3</td> <td>R</td> <td>1890.</td> <td>1415.</td> <td>12.8</td> <td>12.6</td>	3/2	1/3	R	1890.	1415.	12.8	12.6
3/2 $3/1$ R $24/23$. 708 . 4.35 10.93 $1/3$ $1/3$ $2/2$ R -347 . 4626 1.59 -3.43 $1/3$ $2/2$ R -558 . 4386 . 2.05 -3.6 $1/3$ $1/3$ R -347 . 903 . 6.34 -5.25 $2/1$ $1/3$ R -432 . 4635 . 2.01 -1.43 $2/1$ $3/1$ R -432 . 4635 . 2.01 -1.43 $2/1$ $3/1$ R -2613 . 1715 . -5.51 4.07 $2/1$ $3/1$ R -269 . 4552 . -1.35 2.18 $3/2$ $1/3$ R -269 . 4552 . -1.35 2.18 $3/2$ $1/3$ R -289 . 4552 . -1.35 2.18 $3/2$ $1/3$ R -289 . 4552 . -1.35 2.18 $3/2$ $1/3$ R -289 . 4552 . -1.35 2.19 $1/3$ $1/3$ R -289 . 4552 . -1.35 1.26 $1/3$ $1/3$ R -289 . 4552 . -1.93 2.49 $1/3$ $1/3$ R -289 . 4552 . -1.93 2.64 $1/3$ $1/3$ R -1909 . -106 -2.52 -1.93 $2/1$ $1/3$ R -2899 . -184 . -2.52 -1.93 $2/1$ $1/3$ $1/3$ -2.52 -1.93 -2.54 $2/1$ $1/3$ -2.52	3/2	2/2	А	2407.	730.	46	9.15
1/3 $1/3$ $1/3$ $1/3$ -347 4626 1.59 -3.43 $1/3$ $2/2$ R -598 4386 2.05 -3.6 $2/1$ $1/3$ R -3487 903 6.34 -5.25 $2/1$ $1/3$ R -432 4635 2.01 -1.43 $2/1$ $3/1$ R -2613 1715 -5.51 4.07 $2/1$ $3/1$ R -2613 1715 -5.51 4.07 $2/1$ $3/1$ R -289 4552 -1.35 2.18 $3/2$ $3/1$ R -289 4552 -1.35 2.18 $3/2$ $3/1$ R -289 4552 -1.35 2.18 $3/2$ $3/1$ R -289 4552 -1.35 2.18 $1/3$ $1/3$ R -289 4552 -1.35 2.18 $1/3$ $1/3$ R -289 4552 -1.35 2.18 $1/3$ $1/3$ R -4243 -1162 -2.54 1.93 $1/3$ $1/3$ R -4243 -1162 -2.49 1.03 $1/3$ $1/3$ R -4243 -1162 -2.54 1.93 $1/3$ $1/3$ R -2899 -663 0.07 -2.5 $1/3$ $1/3$ R -2899 -2894 -2.49 $1/3$ $1/3$ $1/3$ $1/3$ -2894 1.93 $2/1$ $1/3$ $1/3$ $1/3$ <	3/2	3/1	R	2423.	708.	4.35	10.93
1/3 $2/2$ R -598 . 4386 . 2.05 -3.6 $1/3$ $3/1$ R -432 . 4635 . 2.01 -1.43 $2/1$ $1/3$ R -432 . 4635 . 2.01 -1.43 $2/1$ $2/2$ R -2613 . 1715 . -5.51 4.07 $2/1$ $3/1$ R -2613 . 1715 . -5.51 4.07 $2/1$ $3/1$ R -2613 . 1715 . -5.51 4.07 $3/2$ $1/3$ R -289 . 4552 . -1.43 -1.43 $3/2$ $2/2$ R -289 . 4552 . -1.35 2.18 $3/2$ $2/2$ R -289 . 4552 . -1.35 2.18 $3/2$ $2/2$ R -289 . 4552 . -1.35 2.18 $1/3$ $1/3$ R -2899 . 4552 . -1.35 2.193 $1/3$ $1/3$ R -4243 . -1162 . -2.39 6.68 $1/3$ $1/3$ R -4243 . -1162 . -2.249 $1/3$ $1/3$ R -2899 . -752 . -1.93 $2/1$ $1/3$ R -3140 . -190 . -752 . -1.93 $2/1$ $1/3$ R -2879 . -798 . -7.98 -2.249 $1/3$ $2/2$ $1/3$ $1/3$ -1920 . -1920 . -2.549 $2/1$ $1/3$ $1/3$ -1920 . -1920 . -1920 . -1931 $2/1$ $2/2$ $1/3$ $-2/2$ <	1/3	1/3	А	-347.	4626	1.59	-3.43
1/3 $3/1$ R -3487 . 903 . 6.34 -5.25 $2/1$ $1/3$ R -432 . 4635 . 2.01 -1.43 $2/1$ $2/2$ R -2613 . 1715 . -5.51 4.07 $2/1$ $3/1$ R -289 . 4552 . -1.35 2.18 $3/2$ $1/3$ R -289 . 4552 . -1.35 2.18 $3/2$ $3/1$ R -289 . 4552 . -1.35 2.18 $3/2$ $3/1$ R -289 . 4552 . -1.35 2.18 $3/2$ $3/1$ R -289 . 4552 . -1.35 2.18 $1/3$ $1/3$ R -289 . 4100 . -7.29 6.6 $1/3$ $3/1$ R -4243 . -1162 . -2.39 6.68 $1/3$ $3/1$ R -2899 . -3021 . -3.22 -1.93 $2/1$ $1/3$ R -2879 . -752 . 2.193 1.26 $2/1$ $1/3$ R -7 -7 -7 -7 $2/1$ $1/3$ R -7927 . -1990 . 2.52 -1.93 $3/2$ $2/1$ R -3180 . -198 . 6.33 1.26 $1/3$ $1/3$ R -792 . -752 . 2.58 1.26 $2/1$ $1/3$ R -792 . 1199 . -522 . -752 $3/2$ $2/2$ $1/3$ $1/36$ -1937 . 1.26 -103 $2/1$ 1	1/3	2/2	R	-598.	4386.	2.05	-3.6
2/1 $1/3$ R -432 . 4635 . 2.01 -1.43 $2/1$ $2/2$ R -2613 . 1715 . -5.51 4.07 $2/1$ $3/1$ R $$ $$ $$ $$ $3/2$ $1/3$ R -289 . 4552 . -1.35 2.18 $3/2$ $1/3$ R -289 . 4552 . -1.35 2.18 $3/2$ $2/2$ R -289 . 4552 . -1.35 2.18 $3/2$ $2/2$ R -289 . 423 . 5194 . -7.98 6.0 $3/2$ $2/2$ R -2899 . 4230 . 0.7 -2.5 $1/3$ $2/2$ R -1999 . -1162 . -2.39 6.68 $1/3$ $3/1$ R -1999 . -1162 . -2.49 1.08 $2/1$ $1/3$ R -2879 . -1162 . -2.54 1.26 $2/1$ $1/3$ R -2879 . -198 . 6.33 1.26 $2/1$ $1/3$ R -2.52 -1.93 2.72 1.03 $2/1$ $1/3$ R -2.52 1.26 1.26 $3/2$ $1/3$ R -2.52 1.26 1.26 $2/1$ $3/1$ R -2.52 1.23 1.26 $1/3$ $2/2$ R -2.52 1.23 1.26 $2/1$ $3/1$ R -2.52 1.26 1.26 $3/2$ $2/2$ $1/3$ $1/3$ 1.24 1.26 <tr< td=""><td>1/3</td><td>3/1</td><td>R</td><td>-3487.</td><td>903.</td><td>6.34</td><td>-5.25</td></tr<>	1/3	3/1	R	-3487.	903.	6.34	-5.25
2/1 $2/2$ R -2613 . 1715 . -5.51 4.07 $2/1$ $3/1$ R $$ $$ $$ $$ $$ $3/2$ $1/3$ A -289 . 4552 . -1.35 2.18 $3/2$ $2/2$ R -289 . 4552 . -1.35 2.18 $3/2$ $2/2$ R -4243 . -1162 . -2.39 6.68 $1/3$ $1/3$ R -5198 . 6430 . 07 -2.5 $1/3$ $2/2$ R -1999 . -3021 . -3.2 -2.49 $1/3$ $3/1$ R -5198 . 6430 . 07 -2.5 $1/3$ $2/2$ R -1999 . -3021 . -3.2 -2.49 $1/3$ $2/2$ R -3410 . -180 . 5.52 -1.93 $2/1$ $1/3$ R -5198 . 6.33 1.26 $2/1$ $1/3$ R -3140 . -180 . 5.52 -1.93 $2/1$ $3/1$ R -3370 . -198 . 6.33 1.26 $3/2$ $2/2$ R -3366 . 599 . 1.63 3.72 $3/2$ $2/2$ R -103 . 4184 . -1.52 -1.93 $3/2$ $2/2$ R -103 . 4184 . -1.52 -1.92 $1/3$ $1/3$ R -103 . 4184 . -5.28 -3.26 $1/3$ $1/3$ R -103 . -103 . -5.28 -3.26 $1/3$ $1/3$ R -103 .	2/1	1/3	R	-432.	4635.	2.01	-1.43
	2/1	2/2	R	-2613.	1715.	-5.51	4.07
3/2 $1/3$ A -289 . 4552 . -1.35 2.13 $3/2$ $2/2$ R 423 . 5194 . -7.98 6.0 $3/2$ $3/1$ R -4243 . -1162 . -2.39 6.68 $1/3$ $1/3$ $1/3$ R -5198 . 6430 . 0.7 -2.5 $1/3$ $2/2$ R -5198 . 6430 . 0.7 -2.5 $1/3$ $2/2$ R -1999 . -3021 . -3.2 -2.49 $1/3$ $2/2$ R -3410 . -180 . 5.52 -1.93 $2/1$ $1/3$ R -6352 . 9184 . 5.19 1.08 $2/1$ $1/3$ R -6352 . 9184 . 5.19 1.08 $2/1$ $2/2$ R -3210 . -180 . 5.52 1.93 $2/1$ $3/1$ R -6352 . 9184 . 5.19 1.08 $2/1$ $3/1$ R -6352 . 9184 . 5.19 1.08 $2/1$ $3/1$ R -6357 . 9184 . 5.19 1.08 $3/2$ $2/1$ R -6357 . 9184 . 5.19 1.08 $3/2$ $2/1$ R -7310 . -752 . 2.58 1.26 $3/2$ $1/3$ R -7 -7 -7 -7 $3/2$ $2/2$ $1/3$ $1/3$ $1/3$ 1.19 $3/2$ $2/2$ $1/3$ $1/3$ $1/3$ -7 -7 $3/2$ $2/2$ $1/3$ <	2/1	3/1	R	1	ł	1	1 1
3/2 $2/2$ R $423.$ $5194.$ -7.98 6.0 $3/2$ $3/1$ R $-4243.$ $-1162.$ -2.39 6.68 $1/3$ $1/3$ $1/3$ R $-5198.$ $6430.$ 07 -2.5 $1/3$ $2/2$ R $-1999.$ $-3021.$ -3.2 -2.49 $1/3$ $3/1$ R $-1999.$ $-3021.$ -3.2 -2.49 $1/3$ $2/2$ R $-1999.$ $-3021.$ -3.2 -2.49 $2/1$ $1/3$ R $-3021.$ -3.22 -1.93 $2/1$ $1/3$ R $-3021.$ -3.22 -1.93 $2/1$ $2/2$ R $-30410.$ $-180.$ 5.52 11.08 $2/1$ $3/1$ R $-6352.$ $9184.$ 5.19 1.08 $2/1$ $3/1$ R $-2879.$ $-752.$ 2.58 1.26 $3/2$ $1/3$ R -7 -7 -7 -7 $3/2$ $2/1$ $1/3$ R -7 -7 -7 $3/2$ $3/1$ R -7 -7 -7 -7 $1/3$ $2/2$ $1/3$ $1/36$ $1/39.$ $-5/29$ -3.26 $1/3$ $2/2$ $1/3$ $1/36$ $1/363.$ $-5/29.$ </td <td>3/2</td> <td>1/3</td> <td>A</td> <td>-289.</td> <td>4552.</td> <td>-1.35</td> <td>2.18</td>	3/2	1/3	A	-289.	4552.	-1.35	2.18
3/2 $3/1$ R -4243 . -1162 . -2.39 6.68 $1/3$ $1/3$ $1/3$ $1/3$ R -5198 . 6430 . 07 -2.5 $1/3$ $2/2$ R -1999 . -3021 . -3.2 -2.49 $1/3$ $3/1$ R -3410 . -180 . 5.52 -1.93 $2/1$ $1/3$ R -6352 . 9184 . 5.19 1.08 $2/1$ $3/1$ A -3180 . -198 . 6.33 1.26 $2/1$ $3/1$ A -3180 . -198 . 6.33 1.26 $2/1$ $3/1$ A -3180 . -198 . 6.33 1.26 $2/1$ $3/1$ A -3180 . -198 . 6.33 1.26 $3/2$ $1/3$ R $$ $$ $$ $$ $$ $3/2$ $3/1$ R -3356 . 599 . 1.53 3.72 $3/2$ $1/3$ R $$ $$ $$ $$ $$ $1/3$ $1/3$ R -103 . 4184 . $.3$ -3.56 $1/3$ $2/2$ R -103 . 4184 . $.3$ -3.56 $1/3$ $3/1$ R -103 . $2/2$ -103 . -522 -3.24 $1/3$ $3/1$ R -103 . -103 . -522 -3.24 $1/3$ $3/1$ R -103 . -1139 . -528 -3.56 $1/3$ $1/3$ $1/3$ $1/3$ -1139 . -522 -3.24 <td< td=""><td>3/2</td><td>2/2</td><td>R</td><td>423.</td><td>5194.</td><td>-7.98</td><td>6.0</td></td<>	3/2	2/2	R	423.	5194.	-7.98	6.0
1/3 $1/3$ $1/3$ $1/3$ $2/2$ R -5198 . 6430 . 07 $2.5.5$ $1/3$ $2/2$ R -1999 . -3021 . -3.2 -2.49 $1/3$ $3/1$ R -3410 . -180 . 5.52 -1.93 $2/1$ $1/3$ R -6352 . 9184 . 5.19 1.08 $2/1$ $2/2$ R -6352 . 9184 . 5.19 1.08 $2/1$ $3/1$ A -3180 . -752 . 2.58 1.26 $2/1$ $3/1$ A -3180 . -198 . 6.33 1.24 $3/2$ $1/3$ R -2 -7 -7 -7 $3/2$ $1/3$ R -2356 . 599 . 1.53 3.72 $3/2$ $3/1$ R -7 -7 -7 -7 $3/2$ $3/1$ R -7 -7 -7 -7 $1/3$ $1/3$ R -103 . 4184 . $.3$ -3.56 $1/3$ $2/2$ R -103 . 4184 . $.3$ -7 -7 $1/3$ $2/2$ R -103 . 4184 . $.3$ -7 -7 $1/3$ $2/2$ R -103 . 6.31 . 1.53 -3.56 $1/3$ $1/3$ R -103 . -7 -7 -7 $1/3$ $2/2$ R -103 . -103 . -522 -3.26 $1/3$ $1/3$ $1/3$ $1/3$. -1185 . -528 -3.65 <tr< td=""><td>3/2</td><td>3/1</td><td>R</td><td>-4243.</td><td>-1162.</td><td>-2.39</td><td>6.68</td></tr<>	3/2	3/1	R	-4243.	-1162.	-2.39	6.68
1/3 $2/2$ R $-1999.$ $-3021.$ -3.2 2.49 $1/3$ $3/1$ R $-3410.$ $-180.$ 5.52 -1.93 $2/1$ $1/3$ R $-6352.$ $9184.$ 5.19 11.08 $2/1$ $2/1$ $2/2$ R $-2879.$ $-752.$ 2.58 11.26 $2/1$ $3/1$ A $-3180.$ $-198.$ 6.33 11.34 $2/1$ $3/1$ A $-3180.$ $-198.$ 6.33 11.34 $3/2$ $1/3$ R $-7.$ $-7.$ $-7.$ $-7.$ $3/2$ $2/1$ $3/1$ R $-7.$ $-7.$ $-7.$ $-7.$ $3/2$ $3/1$ R $-7.$ $-7.$ $-7.$ $-7.$ $-7.$ $3/2$ $3/1$ R $-7.$ $-7.$ $-7.$ $-7.$ $-7.$ $-7.$ $1/3$ $1/3$ R $-103.$ $4184.$ $.3.76$ -7.52 -3.26 $1/3$ $2/2$ R $-103.$ $4184.$ $.3.76$ -7.52 -3.26 $1/3$ $2/2$ R $-103.$ $4184.$ $.3.65$ -7.52 -3.26 $1/3$ $3/1$ R $-103.$ $2431.$ 2.528 -3.65 -3.65 $2/1$ $1/3$ R $-1185.$ $363.$ 5.28 -3.65 -3.65	1/3	1/3	R	-5198.	6430.	.07	-2.5
1/3 $3/1$ R $-3410.$ $-180.$ 5.52 -1.93 $2/1$ $1/3$ R $-6352.$ $9184.$ 5.19 1.08 $2/1$ $2/2$ R $-2879.$ $-752.$ 2.58 1.26 $2/1$ $3/1$ A $-3180.$ $-198.$ 6.33 1.34 $2/1$ $3/1$ R $$ $$ $$ $$ $3/2$ $1/3$ R $$ $$ $$ $$ $3/2$ $3/1$ R $$ $$ $$ $$ $3/2$ $3/1$ R $-103.$ $4184.$ $.3$ -3.56 $3/2$ $3/1$ R $-103.$ $4184.$ $.3$ -3.56 $1/3$ $1/3$ R $-103.$ $4184.$ $.3$ $$ $1/3$ $2/2$ R $-927.$ $1139.$ 522 -3.24 $1/3$ $2/2$ R $-1185.$ $363.$ 5.28 -3.65 $1/3$ $1/3$ R $-100.$ $2431.$ 2.5 -3.24	1/3	2/2	R	-1999.	-3021.	-3.2	-2.49
	1/3	3/1	R	-3410.	-180.	5.52	-1.93
2/1 $2/2$ R -2879 . -752 . 2.58 1.26 $2/1$ $3/1$ A -3180 . -198 . 6.33 1.26 $3/2$ $1/3$ R $$ $ $ $$ $3/2$ $1/3$ R $$ $$ $$ $$ $3/2$ $2/2$ R -3356 . 599 . 1.53 3.72 $3/2$ $3/1$ R $$ $$ $$ $$ $$ $1/3$ $1/3$ R -103 . 4184 . $.3$ -3.56 $1/3$ $2/2$ R -927 . 1139 . 52 -3.24 $1/3$ $3/1$ R -103 . 4184 . $.3$ -522 -3.26 $1/3$ $3/1$ R -103 . 4184 . $.3$ -522 -3.26 $1/3$ $3/1$ R -103 . $2/27$. 1139 . 522 -3.26 $1/3$ $3/1$ R -1185 . 363 . 5.28 -3.65 $2/1$ $1/3$ R 1700 . 2431 . 2.5 -3.26	2/1	1/3	R	-6352.	9184.	5.19	1.08
	2/1	2/2	R	-2879.	-752.	2.58	1.26
3/21/3R3/22/2R-3356.599.1.533.723/23/1R1/31/3R-103.41843-3.561/32/2R-927.113952-3.241/33/1R-1185.363.5.28-3.652/11/3R1700.2431.2.5.13	2/1	3/1	A	-3180.	-198.	6.33	1.34
3/22/2R-3356.599.1.533.723/23/1R1/31/3R-103.41843-3.561/32/2R-927.113952-3.241/33/1R-1185.363.5.28-3.652/11/3R1700.2431.2.5.13	3/2	1/3	R	f	1	ł	1
3/2 3/1 R 3 3 3 3	3/2	2/2	R	-3356.	599.	1.53	3.72
1/3 1/3 R -103. 4184. .3 -3.56 1/3 2/2 R -927. 1139. 52 -3.24 1/3 3/1 R -1185. 363. 5.28 -3.65 2/1 1/3 R -1185. 363. 5.28 -3.65 2/1 1/3 R 1700. 2431. 2.5 .13	3/2	3/1	R	1	t	!	1
1/3 2/2 R -927. 1139. 52 -3.24 1/3 3/1 R -1185. 363. 5.28 -3.65 2/1 1/3 R 1700. 2431. 2.5 .13	1/3	1/3	R	-103.	4184.	.	-3.56
1/3 3/1 R -1185. 363. 5.28 -3.65 2/1 1/3 R 1700. 2431. 2.5 .13	1/3	2/2	ጸ	-927.	1139.	52	-3.24
2/1 1/3 R 1700. 2431. 2.5 .13	1/3	3/1	ጸ	-1185.	363.	5.28	-3.65
	2/1	1/3	R	1700.	2431.	2.5	.13

Tracor Ap	plic	l Sci	onc	BS	
بې (۱۱)	4.29	3.62	;	9.62	;
× (II)	18	3.97	:	.74	;
<u>т</u>	1453.	171.	;	20.	:
X (X)	1817.	-422.	1	2668.	!
Accepted or Rejected	A	A	R	A	R
Buoy 3 (Cluster/ Target)	2/2	3/1	1/3	2/2	3/1
Buoy 2 (Cluster/ Target	2/1	2/1	3/2	3/2	3/2
Buoy l (Cluster/ Target)	3/2	3/2	3/2	3/2	3/2

•

Z

Ŷ

.

Ş

7.3.5 Integer Programming Results

Figure 7.2 gives the integer programming problem to be solved for this scenario (see Section 6). Note that the number of potential tracks represented in the objective function (i.e., the number of columns) is fewer than the number passed by the initial guess module. This happens because those tracks with positive costs, as computed by the HTA, need not be considered since they would never be part of the solution to this minimization problem. This can significantly reduce the amount of time spent finding an optimum since the solution algorithm spends a lot of time arranging and sorting the columns.

The solution found by the algorithm is

 $\begin{array}{c} x_{13} = x_{14} = x_{16} = 1 \\ x_1 = x_2 = \dots x_{12} = x_{15} = s_1 = s_2 = \dots s_4 = 0. \end{array}$

Using Figure 7.2 and Table 7.X this solution is seen to correctly correspond to:

 X_{13} - three sensor data for target 3 (1, 1, 1), X_{14} - three sensor data for target 1 (2, 2, 3), X_{16} - three sensor data for target 2 (3, 3, 2).

7.3.6 <u>Conclusion</u>

The MTTA correctly sorted the raw sensor data, correctly eliminated about 70% of the possible pseudo-tracks, and then identified the actual tracks. For this scenario, Figure 7.1 illustrates the estimated tracks generated by

Salarda 2221/22 Statistic Salary

122222



-Integer Programming Equation for Scenario Figure 7.2.

Ľ

AN ALACE

Contraction of

CONTRACTOR 1 CARACTERIA

C. C. L. L. L.

1

222

Ś

3

i.

the MTTA and the actual trajectories followed by the targets. The "tic" marks in each true trajectory line represent two minute intervals. All the estimated trajectories have come fairly close to the actual tracks before the first tic mark. The early severe fluctuations are a direct result of the batch initializer's updating scheme. After an initial state vector estimate is obtained for t=0, a Kalman filter is used to update the state to the current time. The filter is applied with a large initial covariance matrix which allows it to search over a wide area to update the state estimate. Typically, early in the track, only one buoy supplies the first three or four measurements, and this limited observation geometry coupled with the large covariance matrix results in state updates which are substantially different from the actual ones. In all cases, the initial state vector estimates at t=0, and the updated estimates passed to the sequential filter after initialization were quite close to the true values. In any of the tracking accuracy is result HTA event, a characteristics, not MTTA performance, given that MTTA has correctly solved the intersensor data matching problem.

The large fluctuations observed for target 2 in Figure 7.1 are due to the fact that for buoy 3, the cluster extraction algorithm did not put the first third of the T_2 (target 2) data set with the last two-thirds. Thus, buoy 2 by itself supplied the first few measurements for target 2. This failure of the extraction algorithm can be traced to the linearity assumption of the testing mechanism; target 2 went through CPA relative to buoy 3 resulting in severe nonlinear frequency and bearing changes. The first part of the T_2 data set contained the Doppler compression data up through CPA, the last part consisted of the Doppler expansion data for T_2 that was found after CPA. The cluster extraction algorithm rightly viewed these two data streams as different processes, and concluded that they did not go together. It is felt that this

problem can be remedied by including quadratic terms in the extraction's regression model to compensate for the nonlinear frequency and bearing changes encountered during CPA.

7.4

Three Target Scenario with Bad Geometry Low Noise, 0 dB threshold in 1 Hz Band

Figure 7.3. shows the scenario discussed in this section. Table 7.II lists the true initial and final conditions for each of the targets in this scenario. The trajectories simulated are the same as those modeled for the scenario discussed in section 7.3 except that target 1 has a forty-five degree course heading instead of the fifteen degree heading that was used previously.

From Figure 7.3, it can be seen that the MTTA provides reasonable tracks for targets 1 and 2, but no track at all is generated for target 3. Tables 7.XII and 7.XIII list the data and the clusters, respectively, selected for buoy 1. while Tables 7.XIV and 7.XV list the same information for buoy 3. Buoy 2 sees all targets perfectly. From Tables 7.XII and 7.XIII, it can be seen that buoy 1 sees target 3 perfectly, however, it gets targets 1 and 2 confused and splices the last ten points from target 1 onto the target 2 observation set. A similar phenomena occurs for buoy 3 when the first 10 observations from target 2 are interspersed among those of target 1. Buoy 3 also sees target 3 perfectly. Looking at the data, it is clear that the measurements become quite similar during these times and the clustering algorithm simply cannot distinguish between the right and wrong data sets. However, when the power levels of the various signals are examined, a clear difference is exposed. The choice of attribute variables used in this study ignores signal power and consequently



121

1.02

TABLE 7.XII

SIMULATED MULTI-TARGET DIFAR DATA FOR BUOY 1 OF SCENARIO 2

SAMPLE NO.	TIME	FREQUENCY	COS(B)	SIN (B)	TARGET NO.
1	5.0 0000	157.700	.247612	759957	1
Z	15.0007	120.702	· _ P 7 P 17	75 75 85	1
2	15.40055	1.0.1.0	.138427	. 7 % = ? ? ?	3
<u>_</u>	. .		• • • • 1 12	??	ĩ
	• • • • • • •	127.107	• .42757	.745014	3
		1.7.0.7	• 37 - 747	-, -, -, -, -, -, -, -, -, -, -, -, -, -	ī
7	35.1720	1:5.707	• > 7 6 7 24		2
2	75・0707	1.7.1.2	•779779		3
4	43.47.5	100.200	. + 70159	313938	ī
۰.	5 • 2 [*] 2 [*]	1.7.70	•47,117		ī
11	10. J. J. T. T.		.,76437	=43037-731	2
• · · · · · · · · · · · · · · · · · · ·		117.130	• 251744	• 7 / •	3
17	65.0707	150.200	. + * 5 ? ? *	755511	ī
14	50.00 Cut	127.122	• 0 Te 2 Th	.74.771	3
<u>ن</u> ا	75.3730	127.160	• _ 7 _ 6 - 7	564471	ĩ
1	· · · · ·	· · · · · ·	• 17 6F, 4	• 7 7 7	3
•	• • • •	1.57.1.57	• 4 1 6 7 9	-,:? <u>;</u> ; <u>,</u> ,,,	ĩ
_ :	· • · · · ·	1	. 777	-,4 ⁻ -, ⁻ -,1	2
14	[™] 3 • 2 • 2 [™]	107.100	.741769	.07:437	3
Ë.	1000 J. 100 J	157.107	.3:6700		1
.1			.377037	641-011	2
. .	•	1. 7.1	• • • • • • • • •	.7	3
•	• •		ء 1 ل 1 ل.	12797	1
_ !	· · · · ·	1-0.0.5	• • • • • • • • • • • •	.14.7 * 1*	2
<u>.</u>		127.127	?_7:4	- ,	ī
_ :	115.00	147.700	• 9 7 9 7 5 5	105446-701	2
_7	1.7.7.27	157.140	•712535	. 75 15 9 K	3
-	• • •	· · · · · · · ·	• E = . 79 1	75.2.4	1
•	• •	147 • <u>15</u>	.,75550	1.6.74	2
د.	1	· · · · · ·	14774 <u>24</u>		3
۵.	137.7 .0	197.767	.01.490	7:57.46	1
52	137.0	1-7-55	• • 7 • 7 • 7 • 1		2
13	125.700	157.100	.729714	• 5 7 1 7 1 7	3
54	147.307	1.7.567	4 2 1 6 4	749337	ĭ
ن.	• • •	140 J. T. T.	• / • •		2
•	• • •	100.000	s e se en en	• • • • • • • • •	ĩ
- '	• •	1	• • • • •	. 4 1 4 30	2
1 1	105.115	100.000	.7434.1	••• <u>•</u>	ī
د د	10	147,017	• • • • • • • •	• 74 ; 7 [? = ^. 1	2
◀.	17: . 27		· · · · · · · · · · · · · · · · · · ·		1

(CONTINUED)

-

•

TABLE 7.XII (CONCLUDED)

TARGET

SAMPLE

<u>ک</u> ار

-

Ņ

3

j

ŝ

NO,	TIME	FREQUENCY	COS(β)	SIN(B)	TARGET NO
	• • •	147.547	. 747-	·1:05:00	2
4 L	• • •	107.107	• . 1 + C . 4	.572931	2
4 ⁷			. 77 . 774	65-057	1
44	107.07	123.165	.777927	•6*20 mm	3
4.	1.7.1	149.025	74 777	5+2755	1
44 ·	197 . 1 97	142.802	.9999427	170604-001	. 2
47	· · · · · · · ·	127.167	· 7:511	. 5 7 7 7 6	2
4 4	• • •	140.00	• 24 1777		1
+	• •	242420	· · · · · · · · · · · · · · · · · · ·		2
5		157.100	·107080	.6.2721	2
٦.	71.01	140.000	• 245111	- <u>.</u>	
	7 . 7•1d7	147.800	.763791	262090	2
	•	1_1.107	.7014.0	• 61.500	2
ہے ا	• •	149.20	 ↓↓ E 9 € 4 E 	-,51,001	3 1
	•	1424227	• • • • • •	4 -00	1
j _	· · · · · · · · · · · · · · · · · · ·	149,200	• \$ 7 7 7 7 7	.35.3954	2
37	0.07 • Tu N	107.105	.75572€	.67.244	2
i l		149.000	.950275	3 34 -7	3 1
3	4	1	• >**€ ** 1	.717750	7
	•		7 7 6 5 7	3730_0	1
	•	1 J 1 • 1 J 7	1 . 7 . 0	• 5 C 2 C 5 C	2
<u>د</u> .		157.707	.775736	.571774	ž
	27	140.500	. /F 75 15	- <u>-</u>	1
04	72 7. 727	142.740	. : 71025	1-5ean	1
· ·;	• •	147	• • • • • •	23CE37	1
.+	•	1 - 7 - 7 - 7	. ,**⊽:41	.31 947	2
ວີ			. 1 2 . 10 2 -	. 814737	2
C .		140.707	• <u>)</u>		1
. .	**************************************	142.707		2-3-40	1
7		140.70-	.,40502	31-530	2
71	· · · · ·	1:0.500		4 L	2
7.	•	1 - 2 - 7 - 7	,	-1-40-5	1
7:	• •	143.202	• 1 2 4 4 1		2
7+		100.000	. 17 420	•5 Jun •1	2
75	747.0	149.70	. ; ? 4 ~ 5 E		ວ 1
15	747.	1.0.00	· · · · · ·		3
17	3-1	147.7.7	, 24 - 1 -	- 17 - 7 19	ن 1
1	•				43
		· -			5

CLUSTERING OUTPUT FOR BUOY 1

TABLE 7.XIII

1.1.1.1

OF SCENARIO 1

CLUSTI	IR NO	DE	SIZ	E	~	OBJ	ECTS	5 5	SNT	TERE	Ľ																
-	-	•	~			\$	121					-	يوم. منهر	14		÷÷ ≓æ			2			を					i
~?	~		2		~	ет Ф	32			•																	
т. Ж		i	.			S	1	5) 1	a		ба 3				-	2		1			÷.,	F		-			1
		•	N						: • •		ו							ang Ngg Lang					-			-50	**
S	ŝ		m		~	2	75]																				
4		h	2	ç	f.	- 	\$\$	•	5. К. – С ДС	i i					-	1.1				F				ř			-
~	•	•	•••			0	ğ		-			5	•••	-• r .				4	2								
8	Ð		2			m	ŝ																				
•	6		3			1. A. A.	201	ň			į.	7		•			1			1						÷.,,	• :
10	10		N			•		•	•					•			4			7.5						•	•
11	11		~		N	0	23	5																			
12) S	~					F.;;							5			i			- -	1.1		ŕ 4.			
13	13		**	44 43		•	51% 21%	″. 	-5			j.	на. 										•		-	•	•
. 4	14		N		v	60	34																				
15	ST	r Z	N	•		54		•		•	••	1 . 				•	n F F 2	7	· · ·					÷.		•	
16	11.		a		.	ي. ح	- #-	21	.						•		: •	**				•		,	· .	• -	~
11	18		æ		4	~	5	Ľ,	53																		
18	, 19		r 2		in L	N	12	•			•	:				7 -	•		i,	•	•	•	•		5	4	\$
61	20		- #		•		10	13				× .				•	;	· .	•	.:	•	.		•	.		-
										9	TNO	UNI.	ED)														

1

3

Ì

3

đ

1.1.1

S. 16.

AND SAVA

TABLE 7.XIII (CONTINUED)

A CONTRACTOR AND A

				ç		m ≱	an a				*-	ł	15-1-2 5
						Ĭ							
	L'ANTER S			L									
	1	•		• •									
	Mer. 8.			•• •									
				4 2 2									
					ана стана 1977 година 1977 годи 1977 годи 1977 годи 1977 година 1977 година 1977 година 1	•							e view
				k k									
	े ्र राज												1 2 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1
ERFD		1						1 57	F				-
LUSTI						4 11		50 53	OI D				
TS C			545		N	12	72	24	-		21	21	10 51
BJEC	26.60	2		63		S	21	2 4		04	22	18	8 8
0		-		90 141	2		8 9 8	4 2		36	5 S	11	9
6-1													
SIZE	N	~		~	N M	4		9	N O	~	N	m	
NODE	23.1.	5 4	26 . 13 29 . 29 29	30		35	36	5.		46		64	52
R	 		5		E .								:
CLUSTE	20	21		5 #	58 58	27	58 54	30	32	33	1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	36	37

(CONTINUED)

· · · · · ·

TABLE 7.XIII (CONTINUED)

OBJECTS CLUSTERED

SIZE

NODE

CLUSTER

	53	70 73		ł	: 				
1 1.5 6 6.1 6.2 7.1 7.4 7.8 1 1.6 1.7 1.2 1.4 1.5 1.5 1.5 1.5 1 1.6 1.7 1.8 1.6 0 3.5 1.5 1.5 1.5 1.5 1.5 1.5 1.5 1.5 1.5 1.5 1.5 1.7 1.6 1.5 1.5 1.7 1.6 1.5 1.5 1.7 1.6 1.7 1.6 1.7 1.6 1.7 1.6 1.7 1.7 1.7 1.6 1.7 1.7 1.7 1.7 1.6 1.7 1.7 1.7 1.7 1.7 1.7 1.7 1.7 1.7 1.7 1.7 1.7 1.7 1.7 1.7 1.7 1.7 1.7 1.7 1.7 1.7 1.7 1.7 1.7 1.7 1.7 1.7 1.7 1.7 1.7 1.7 1.7 1.7 1.7 1.7 1.7 1.7 1.7 1.7 1.7 1.7 1.7 1.7 1.7 1.7	777			F .	28				
1 1 5 1 5 1 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 1 1 5 5 5 1 5 5 1 1 5 5 1 1 5 5 5 1 1 5	n #	J 65 6 61 62	61	11	74	78			
06 69 5 0.3 05 15 15 15 15 15 15 15 15 15 16 10 13 07 17 8 15 17 20 23 25 15 17 20 23 29 11 20 23 29 11 20 21 21 21 21 21 21 21 21 21 21 21 21 21 21 21 21 21 21 21 21 21 21 21 21 21 21 21 21 21 21 21 21 21 21 21 21 21 21 21 21 21 21 21 21 21 21 21 21 21 21 21 21 21 21 21 21 21 21 21 21 21 21 21 21 21 21 21 21 21 21 21 21 21 21 21 <td< td=""><th></th><td></td><td>4 0 4 9</td><td>- M - Q</td><td>26</td><td>53 51</td><td></td><td></td><td>1</td></td<>			4 0 4 9	- M - Q	26	53 51			1
11 11 12 13 11 12 13 11 20 23 25 15 11 20 23 25 10 17 15 15 17 20 23 25 23 25 11 13 14 15 17 20 23 25 29 12 24 29 12 24 29 12 24 29 14 25 29 11 24 29 11 25 19 11 25 19 11 25 19 11 25 19 11 24 24 26 12 24 24 26 14 25 10 11 25 10 11 27 27 26 11 27 27 27 26 11 27 27 26 11 27 27 20 11 27 27 20 11 27 27 20 11 27 27 20 11 27 27 20 11 <	9 8	5 69 5 63 45	61) 47	51	54				
49 75 5 15 17 20 23 25 51 83 3 36 10 14 14 22 27 30 51 83 3 36 10 14 14 22 27 30 52 64 6 24 29 12 14 14 22 27 30 53 90 7 94 52 55 19 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1			27 S	9 Q					
50 717 8 12 14 15 27 30 51 83 5 36 70 22 24 29 25 39 41 52 64 6 24 29 25 39 41 53 90 7 80 55 90 12 14 19 22 21 10 11 53 93 3 60 70 73 35 40 1 14 22 21 10 14 22 21 10 14 22 21 10 14 22 21 10 14 22 21 10 14 22 21 10 14 20 21 10 22 21 10 14 20 21 10 22 21 10 12 14 14 20 21 10 22 21 10 14 20 21 20 21 20 21 20 21 21 14 14	5	75 5 15 17	20	23	25				
52 64 6 24 29 72 35 39 11 53 70 7 60 72 35 39 11 54 72 3 66 70 73 35 39 11 55 93 3 66 70 73 35 39 11 56 95 71 74 74 74 74 74 76 57 96 10 3 66 70 73 22 21 30 16 58 15 16 12 74 74 76 22 21 30 59 115 7 59 12 74 74 76 22 21 30 16 17 74 70 50 31 50 51 50 51 50 51 50 51 50 51 50 51 50 51 50 51 50 51 50 51 50 51 50 <	51 5		12.	*	b	22 27 30		and the second secon	
53 93 3 66 70 73 35 19 01 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	25	? B4 6 24 29	12	35	39	"]			
55 93 3 66 70 73 56 95 95 3 66 70 73 57 96 10 3 5 10 12 10 12 10 12 11 74 78 58 115 7 59 61 62 71 74 78 59 115 7 59 61 62 67 71 74 78 50 116 17 74 78 61 11 74 78 50 116 17 59 61 62 67 71 74 78 50 116 17 74 78 36 16 17 74 78 50 116 17 53 51 16 16 17 74 78 50 119 15 26 28 38 61 17 72 27 21 20 31 87 57 51 51 51	5 2 2 3 3		5 6 2 9	N. 1	32				
58 95 96 10 3 5 16 19 22 21 30 36 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 <td< td=""><th>55</th><td>5 93 3 66 70</td><td>13</td><td></td><td></td><td></td><td></td><td></td><td></td></td<>	55	5 93 3 66 70	13						
	56		49 . 53	~ ~	* *	19 22 21			
	58	3 115 7 59 6 1	62	61	11	74 78			
	5.		*	36	80 M				
	60		6 0	3		16 19 22	27-130-151-42	51 51	

1

1

L.L.A.

. رب. ج ļ

TABLE 7.XIII (CONCLUDED)

Tracor Applied Sciences

5

20

69

TABLE 7.XIV

SIMULATED MULTI-TARGET DIFAR DATA FOR BUOY 3 OF SCENARIO 2

SAMPLE NO.	TIME	FREQUENCY	COS(B)	SIN(B)	TARGET NO.
1	5.Ju%u0	157.407		- 472472	2
đ	15. 3900	1:7.500	979414	-,76 2040	1
2	15.3750	157.763	36695	- .? ^r .14 ^r	2
-	سري ميني سامي الانتخاب	167.000		14-7.7	1
•	i Line and in the	120.767	 ■ ↓ < < < 27 	+.2427.4	2
-		127+700	705f+5	•t12217	3
7	35.JCUA	107.100	- •≠°3r3F	+.153RuT	2
÷	45.0000	157.500	9731 30		1
÷	45.000	157.198	-,,26923	-,7°3°u9+101	2
<u>م</u> -	···· · · · · · · · · · · · · · · · · ·		 777:29	• 7 4	3
÷ •	ັບ າ - ີ - ີ	157.Fur	- .975014	→ , 20 / 5 , 4	1
۰ ۲	11.17.2	110.00	- .909407	➡,7™\≤?? > +`∪ <u>1</u>	2
17	52.0733	107.207	744=1r	.567767	3
¥ =	65+4720	107.507	-•>^5767	+.JT3570	1
41	45.0150	300.000	-,,93727	.373497-731	2
•			-,12.2.4	•	3
•		107.527	- , 94 <u>-</u> , 1	111-1	1
-		142.0.0	- • y ^ 111	• 1 - 2 - 7 - 7	2
<u> </u>	76.0305	157.757	7°7537	• • 1 • 2 7 9	3
8		149.00	- • 76 97 56	• 2 • • 7 • 7	2
	- 4 ≎ • 2 [™] 2 [™]	14745 1477	■ • > ⁵ 7 ⁵ ¹ ¹	• _ * 1 = 4 =	2
. .			•••		1
-	· · · · ·		- • · · · · · · · · · · · · · · · · · · ·	• 3 . 7 5 1	2
<u> </u>	1				1
4		147.00	■ 1 2 4 3 13 1	• • • · · · • • • • · · ·	2
⊷ ¹ .					3
• 1					1
-			- بالار	• 54 24 54 • 7 7 4 5 4	2
-			• y - 4 - y - 5 		1
• ···	• • • •			• • • • • • • • •	2
ب ا للہ ا	- 4 ¹ 1 4 − 4 - 5 − 4 − 4 − 4	100 e 100 1000 E 17			3
4 K. 2 m	1 E #	101+100			1
ب ن بر د	1.5.0	1971700			2
24					2
-	• • • • • •		- · · · · · ·	• • • •	2
₩ I 	4 y 5				э 1
-		1	• / / / · ·		2
•	17		••••••••••••••••••••••••••••••••••••••		2
- ·	1		- 37-42 C		
H	a () € a	• •	• · · · · · · · ·	• • • • • • •	T

(CONTINUED)

•

...1

į

TABLE 7.XIV (CONCLUDED)

SAMPLE NO.	TIME	FREQUENCY	COS(β)	SIN(B)	TARGET NO.
-4 <u>-</u>	1 - ² - ² - ² - ²	147.407	*. : <u>`.</u> `.`	.70,000	2
41	1	100.700	710774	•57531f	3
4 .	1 - 7 • 1 - 1	1-7.7	- •) 7 <u>-</u> 7 <u>-</u> 5	.775477	2
44	195.333	1:0.300	712429	.7~:744	3
45	205.100	127.460	;91736	10,040	ī
46	205.128	142.313	573445	•707434	2
47	205.707	100.700	147565	• 7 _ 7	3
4 ÷	7 <u>-</u> 1	142.54	− .J13725	•777117	2
-	1	127.407		344737	ī
30	215.100	140.100	<u>546741</u>	* 6 7 5 0 Q X	2
±1	227.707	150.700	- •k^23967	• 5+ 6£ 2°	3
<u>ت</u> ت	253.037	149.200	373599	•34375T	2
J .	23*.127	1.7.7_7	- .//u/ju	•7 * . 4 , 7	3
54	247. 27	120.400	- .,+279€		1
	<u></u>	100.700	74 .440	· 575677	3
5 E	265.00	149.750	422151		2
57	255.722	108.707	7722.2	. 57 . 77	3
2 6	265.100	157.467			1
-	5	127.727		71 . 4	3
τ.	511.12	107.407	-,,2474r	17.4.1*	1
•	· · · · · · · · · · · · · · · · · · ·	140,700	- + 6 37 37	. 25 . 574	2
	185.0 <u>5</u> 1	149.707	- 474490	914451	2
	2 - 5 - 5 - 5 - 5	1.0.707	7-16	600500	2
54		1.5.4.5		173073	5
•					2
					2
, 1		1 1 4 5			5 1
C		140,200	-, ?		1
0	7,5,5,5,5	1 1 7 0	- 7617-5		2
7		1-7-450	-, 2770 ⁻⁷	- 1 ° K P • 7	3
					1
	· • • • · ·	1 7 7 7			2
	· • • •			$\begin{array}{c} \bullet & \bullet \\ \bullet & \bullet \\$	3
7.4	······································		+ U & U - U 1127		3
74					1
7:	22.00 .00 .00			• 7 · 2 · A	2
i () 7 7	745 313				<u></u> ታ
÷ 1 2					1 2
1	··· · · · · · · · · · · · · · · · · ·	• ™ • • • • •		• 7 •	2
1		• • • •			3
-					1
	22110			● * * * · · · · · · · · · · · · · · · ·	2

TABLE 7.XV

FPUT FOR BUOY 3 ENARIO 2	の「「「「「「「「「「「」」」」」」「「「「「」」」」」」「「「」」」」」」」」													
USTERING OUT OF SCEI	CLUSTERED						£					1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1		1
5	OBJECTS	68 71	69 9ÿ		59 63	111 119 593 19	55 59 6	24.24	Ju 33		20 21			
	E SIZE	N	61	NN	~		m		2		2			
	ER NOD	~	ŝ		60		11	13	* -		11	89	42	24
	CLUST	د م	ري	10 P	Ŀ	29 C) 14	11	12 13	14	91 	11	61	20	5

1

130

iin La

•

1.1.1

.

1977

9

に開

2 H			₩0 	** ******														
' \$₹	54				19.00	•												C (
26	30		-	S	5	29	.9.	5.0	i e te									
27	11		ac	51	55 5	7 59	63 6	699	76									
28	.			04 0404	1. 4 4 S. 1	1 57	3.		5. 5. 1. 1.				r	L. Lean				
50		_	975 10	9	52.5										، بر بر			
20	35	-	*	22	24 2	1 32								-				
31	95		X	Land of	6 C.		ет н 11 настра 19		n in second								5 K	
32	11	•			71 71	5	****		•	•		a q						-
33	39		2	8 4	67													
34				05	. 25 . 21	19	62 6		€. + ; ~	10 - 8 2 -				÷	2			7
35		-			45.41	. 23	22	45 6	63	6	76		•	: ب بسر	· ·	-	i i i	•
36	4 2		3	60	11 11	1 17												
37	÷.		•			350	s. 23		29	9				•	14	, 1		
38	54	· ·	12		1 3	1 53	55	7 59	63 (12	76	• •	• •	•	•		•••
35	53		3	10	13 1(5 19												
90	3 0 454 5 7	•		85.	61.7	11 0	•	<u>ب</u> ،					•	•	:	•	•	•
[]	5 5 5	· •			0	3 - -	•	t A 4	ید اید ۲	ء بر ية				•	2	с. Э	•	
	• ອະ • ມາ		1 3				ີ ຍາ ດີ ຍາ ເ	65 L	93	9	12	ء مور مر	<u> </u>		•		` `	

•
(CONTINUED)

1

. . .

(CONTINUED)	- 「「」」「「「「「「「「」」」」「「「「「」」」」」「「「」」」」」」」」」		57 59 63 66 69 72 76 82					153 155 157 159 163 168 697 72 13 76 142 142 15 15 15 15 15 15 15 15 15 15 15 15 15					74 77 80	
TABLE 7.XV ((LUSTERED	11 11 11 11 11 11 11 11 11 11 11 11 11		42 44 53 55	16 - 19 - 28 - 36 - 4 76 - 77		42 446 53 55 29 32 40	11 14 17		29 32 37 40	18 19 26 31 76 79 82 60 67 70 74	58 60 67 70		58 60 67 70	1. 14 1. 14
OBJECTS C	1. 0 1. 0 1. 0 1. 0 1. 0 1. 0 1. 0 1. 0		71 34 39	13 10 13 10 13 10 13 10 13 10 10 13 10 10 10 10 10 10 10 10 10 10 10 10 10	29 ti 0	12 42 22 34 34	2		22 24 21	() 10 13 10 13 10 13 10 13 10 13 10 13 10 10 10 10 10 10 10 10 10 10 10 10 10		11 12 14 12 14 14 14 14 14 14 14 14 14 14 14 14 14	#5 #9 5#	
R NODE SIZE	- 94	69 	73 15		79 2		8 8 8		95 7	96 .	120 9	127 5 127 5 14 14 15 15 15 15 15 15 15 15 15 15 15 15 15	145 10	Marsh St. Links 641
CLUSTE			5.	50 51	52	5 53	55	56	8 5	65 .	61	9 9 1 1 9	• •	

Tracer Applied Sciences

-1

.1

Ì

Ŋ

22

23

2

сб.н.

TABLE 7.XV (CONCLUDED)

the second

CLUSTER	ION	Œ	SIZE	OBJ	ECTS	ដ	USTI	RED	_														
67	25	ŝ	12	35	38		5	.	8		56	Ţ	9	65									
	56					12	1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1			S		9					2	24	7				
70	117	60	19	30	33	35	38	•	ē m	¥ 	1 50	52	ŝ	6	62	65	68	71	75	78	81		
K			23	23	35	10 - 10 30 - N								5				3			71		
2 2 2		6 . . .		2 2 3 3 3 4 3 3 4 3 4 3 4 3 4 3 4 3 4 3		34.0	12 11 10		N		Ĩ.			'n								9	F. ,
76	31	0						5 N				and a second s										й. С.,	
a 61		1			2 70		8			2	60 11 60	21	5	10.1				5	2.5	60 15 	60	5 N.	·
.	25	י גיז	- - 9	1 1 1 2 1 2 1	2 4 5 2 4 5 7 5 2 4 5 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	3.24	18≉02 12≉73 - 18	5 27	60 60 60	~	6 % # 6 % #	34 30	11 20 20	37 23	10 13 10	40 72	15	9 7 7 7 7	17	1 1 8	6 6 6 6 6 6	6 2 7 2 7 2	M'N
3	й Ф 	ະ ເວ.					555 0 8 M 5	67 8 93 67 8 93 67 68	000 000 00 00 00 00	~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~	0 01 80 80 10 # 00	20 20 20 20 20 20 20 20 20 20 20 20 20 2		12	32 32	122	194 194 194						- 000

treats all measurements as if they were received at a common power level. However, the propagation losses for a near target should be much smaller than those for a distant target, so clearly the received signal powers should vary according to the proximity of the target to the receiver. Other studies currently under way at Tracor indicate the power levels can be clustered by applying an intensity index to the power level and then clustering with this index. It is felt that application of a similar methodology to DIFAR data could improve the ability of the algorithm to separate data from two targets that are close in bearing and frequency, but differ in power levels.

7.5 <u>Three Target Scenario with Good Geometry</u>, <u>Moderate Noise, No Threshold</u>

Geometrically, this scenario is the same as the one discussed in section 7.3. The difference between the two scenarios is in the simulated source power levels. For the first scenario, there was an 80 dB source level, but for this scenario, the source level was reduced to 77 dB. Figure 7.4 provides graphical comparison of the target true tracks versus the MTTA's estimates of these tracks.

This particular scenario was simulated to test the following two data processing capabilities of the MTTA:

- The ability of the clustering and automatic cluster extraction algorithms to separate signals from random, background noise, and
- 2) The ability of the MTTA's HTA and IP to pick the correct targets and to adequately track them with noisy data gathered from moderate strength target signals.



As stated previously, the 3 dB drop in the simulated target signal strength resulted in an approximate 30% decrease in the range over which the target could be readily detected. For this scenario, each of the targets could be well detected by one (i.e, the closest) sonobuoy, each could be fairly well detected by a second sonobuoy, but the targets could just be marginally detected by the sonobuoy farthest from the signal Particularly if a 0 dB threshold was used source. for detection by the sonobuoys, each target would be nearly unobservable for one sonobuoy in the pattern. To compensate for any possible loss of data, it was decided to drop the threshold criterion and to let the clustering algorithm sort the true signals from the random noise.

Overall, the MTTA performed very well in processing the noisy data. For the most part, random noise was eliminated from the true signals by the clustering and automatic extraction processes. Most of the signal clusters contained a couple of noise points that were similar to the true measurement set, but that did not really belong with the true data. However, the DIFAR simulator computes a measurement standard deviation that is a function of the estimated SNR of the received signal. For the spurious noise samples, the estimated SNR values were very small (i.e., less than 0 dB), so a large variance was computed for the frequency and bearing estimates. The HTA uses the measurement variance to weight the measurement before generating a least squares update of the large variances track. When are provided with target measurement data, the tracker essentially weights the data so

1

à

R

CANANY IN ANY IN

that they can cause little or no change in the track update. The net effect is that when only a few poor measurements are provided to the HTA with variances that indicate the data should carry minimal weight, the HTA can continue to track the target without seriously degrading the track updates. The results from this scenario, presented in Figure 7.4, illustrate this point well.

From the figure, it can be seen that after some fluctuations during the track initialization phase, excellent track estimates were obtained for targets 2 and 3. For target 1, somewhat poorer estimates of the trajectory were obtained. Part of the problem with this estimate is that target 1 is approaching buoy 3 nearly head-on from a very long distance. Because the target is heading nearly directly at the buoy, very little bearing change would be observed even for a well received signal. Since the signal is only of moderate strength is fairly distant from the sonobuoy receiver, and the propagation losses are substantial enough to cause the signal to be poorly received. Since the signal is so poorly detected, serious fluctuations result in the bearing measurements which may in turn appear to the tracker to be a significant bearing rate. Hence, even though buoy 3 does observe target 1 and the clustering and cluster extraction processes adequately sort the data, the measurements from buoy 3 are still of such poor quality that they adversely affect the tracking solutions from the MTTA for target 1. Nonetheless, these tracking solutions for all the targets would probably be adequate for real world encounters.

In conclusion, the MTTA seems to have passed the two tests that this scenario was designed to examine. First,

it did an adequate job of separating noise from true signals. Second, the MTTA did an adequate job of tracking the targets with the poorer quality data generated for this examination. However, once again, bad observational geometry adversely affected the MTTA's estimates, perhaps even more so than the degraded quality of the data.

7.6 <u>Two Target Scenario with Good Geometry, Low Noise,</u> and 0 dB in a 1 Hz Band Threshold

The last case considered by this study was a relatively simple, two target scenario. The same tri-tac buoy pattern was used to observe the targets, and favorable geometries and low background noise levels were used to simulate this scenario. Figure 7.5 contains a plot of the true target trajectories versus the MTTA's estimates of the tracks for this two-target scenario. While the tracking solutions presented in the plots show good results, one interesting point was encountered in generating these results that needs further For target 2, the MTTA picked only a 2-cluster, discussion. intersensor combination to track the target rather than the correct 3-cluster combination that would have been preferred. The remaining third cluster was left unused, and the MTTA did a good job of tracking the target without the data from this One must question why the third cluster describing cluster. target 2 was excluded, however, and so the results were more closely examined to find the cause of this exclusion.

For this scenario, each sonobuoy should have generated two different data clusters, one cluster to represent



each of the two targets. The clustering and extraction algorithms did a perfect job of sorting the data into individual target data sets for each of the sonobuoys. Next, the data clusters were passed to the initial guess (IG) algorithm to eliminate unlikely cluster combinations and to keep the rest. Since the data were simulated for this scenario, it was known that the correct cluster combination for target 1 should have been (1, 2, 1), meaning the first cluster from sensor 1, the second cluster from sensor 2, and the first cluster from sensor 3 should be combined to track target 1. For target 2, the correct three-cluster combination should have been (2, 1, 2). The IG correctly passed all of the correct 2-cluster combinations and the single 3-cluster combination for target 1. It also passed all of the correct 2-cluster combinations for target 2, but the IG rejected the one 3-cluster combination that should have been passed to represent target 2. Apparently, the 3-cluster statistical test found too much statistical difference between this estimate and the 2-sensor combinations that make-up this cluster triplet. Because of this rejection by the IG, the preferred 3-cluster combination was rejected before the MTTA's IP could even consider it. Without this cluster triplet, the IP could only pick the single 2-sensor combination that minimized the cost function and leave the remaining cluster unaccounted for. This proved to be exactly what happened when the MTTA picked opt nal solutions of (1, 2, 1) for target 1 and (2, 1, 0) for target 2 and left the second cluster for sensor 3 unused.

In conclusion, the MTTA once again performed well in tracking the two targets in this scenario as can be seen in Figure 7.5. Unfortunately, the MTTA's IG module prematurely eliminated the cluster combination that would have been preferred to track target 2. This one case is the sole No.

A

F

2022

example in all the multi-target scenarios studied where the IG failed to pass the preferred 3-sensor solution. It should be remembered, however, that any statistically based test is going to make a wrong decision part of the time, even though hypothesis tests are constructed to minimize type I errors. The result is that, occasionally, good initial guess estimates are going to be eliminated purely by chance.

8.0 SUMMARY AND RECOMMENDATIONS

8.1

.

_

272222

a state and a state of the

Ę

ſ

÷

いたのでものでした。

Project Summary

The scenarios discussed in Section 7.0 have shown that in its present form the MTTA possesses the ability, with no required operator interventions, to properly sort passive sonobuoy data into individual target data sets, correlate those data sets across sensors, and select the correct target track scenario from the hundreds of potential tracks. Additionally, in its current configuration, the algorithm has performed in near real time and, with minimal processing alterations, could operate substantially faster than real time.

Simulation results showed the importance that scenario geometry played in correctly identifying appropriate tracks. Under certain circumstances, the clustering algorithm can become confused over target data and group together points from different targets having substantially different power levels. However, even in the worst case studied, two of the Results three possible tracks were estimated fairly well. based on good geometry but moderate to high background noise levels indicate that the MTTA is not as sensitive to noise levels as it is to bad geometry. This implies that the clustering and extraction algorithms are confused less by measurements corrupted by noise than by similar measurements coming from different targets. Remedies to these problems will be discussed in the next section.

8.2

Recommendations for Future Work

The recommendations for future work contained in this report fall into two natural categories: enhancements to

correct the deficiencies observed in the MTTA under the current contract, and extensions required to make the MTTA able to perform for long periods of time.

Under the current contract, the following deficiencies were detected in the MTTA:

- Confusion of targets under certain geometries, even when the power levels of the signals were different, and
- 2) Incorrect cluster extraction when targets were undergoing CPA.

To remedy the first problem, it is felt that the estimated power level can be converted to an intensity index, and this index can then be used as an added attribute to sort the data by clustering techniques. The reason for converting the power level to index form is that typically the raw power-level estimates vary too much to be useful, but indexing such as is done for visual displays has proven to be useful in other studies. The second problem can be corrected by adopting a curvilinear model instead of a linear one for the extraction regression equations. This should permit the model to more closely approximate the nonlinear measurement time histories cbserved during CPA.

Proposed extensions to the MTTA are essentially based on the nature of the multi-target tracking problem itself. Basically, the problem can be divided into two parts. The first part consists of recognizing new targets and initializing their tracks, and the second part consists of updating tracks for established targets. Tracor's MTTA, as it

presently stands, is essentially a solution to the first part of the problem. It can take raw data, automatically cluster to recognize targets, and then estimate the associated tracks for each of the targets.

Figure 8.1 represents Tracor's conception of the completed multitarget algorithm. The current MTTA is located in the "search for new targets..." box. The boxes surrounding it contain data management functions. They maintain sensor data pools, place elements in the current track table, and remove points forming a trajectory from the sensor data pools. There is no theoretical development associated with the implementation of these functions, they amount to the lines of code necessary to set up the appropriate data structures and module linkages.

The thrust of the theoretical development will be concentrated on determining if a given data point is associated with one or more current tracks and then performing the appropriate update. There are several potential candidate strategies for attacking this problem including:

- Nearest neighbor and gating approaches of Sea (Reference 18) and Singer and Sea (Reference 19).
- A posteriori analysis of track density by Sittler (Reference 20) and Stein and Blackman (Reference 21),
- 3) Probability data association scheme used by Bar-Shalom and Tse (Reference 22), and
- 4) Cluster-based data association schemes.

Ċ,

STATES STATES

and the second



Figure 8.1. Logical Flowchart of Tracor's Proposed Final MTTA

At this point, the thinking is that track determination and state update will be handled by a combination of (1) and (4) in conjunction with the existing sequential filter used by the HTA. Recent work (Reference 23) indicates that (1) is among the most widely used of all measurement classification schemes. Coupled with a gating mechanism at the clustering stage, it seems like a very promising approach. Additionally, to handle the problem of intersecting targets, it may very well be appropriate to allow given measurement to update more than one track but with a substantially reduced measurement weight for each track. This is similar in philosophy to use of the $B_{\rm b}$ overlapping cluster algorithm to sort data.

In summary, it is felt that the most difficult part of the multi-target tracking problem, target recognition and track initialization, has largely been solved. The literature contains a broad range of strategies for updating tracks once initialization has occurred, but up until now, little work done with the there has been problem of multi-target track initializations. It is believed that Tracor's MTTA provides the basis for solving this total problem, and it is recommended that its development be continued.

7

)

2

ŝ,

귉

j

9.0 REFERENCES

1) Reeder, Hugh, <u>Final Report: A Maximum Likelihood</u> <u>Procedure for Air ASW Programs (U)</u>, Contract N60921-79-C-0123, 15 May 1980.

Cardalan.

- 2) Corser, Glenn and Wilson, Thomas, <u>Final Report: Hybrid</u> <u>Passive Tracking Algorithms</u>, Contract N00014-78-C-0670, 31 October 1980.
- 3) Cooper, Don and Corser, Glenn and Wilson, Thomas, <u>Final</u> <u>Report: Hybrid Tracking Algorithm Improvements and</u> <u>Cluster Analysis Methods</u>, Contract N00014-78-C-0670, 26 February 1982.
- 4) Alam, Kursheed and Mitra, Amitava, "Polarization Test for the Multinomial Distribution," <u>J. Am. Stat. Assoc.</u> Vol 76, 1981.
- 5) Draper, N.R. and Smith, H., <u>Applied Regression Analysis</u>, John Wiley, New York, 1966.
- 6) Neter, John and Wasserman, William, <u>Applied Linear</u> <u>Statistical Models</u>, Richard D. Irwin, Homewood, Ill., 1974.
- 7) Gallant, A.R., "The Power of the Likelihood Ratio Test of Location in Nonlinear Regression Models," <u>J. Am.</u> <u>Stat. Assoc.</u> Vol. 70, 1975.
- 8) Morefield, Charles L., "Application of 0-1 Integer Programming to Multi-Target Tracking Problems," <u>IEEE</u> <u>Trans. on Automatic Control</u>, Vol. AC-22, No. 3, 1977.
- 9) Jardine, Nicholas and Sibson, Robin, <u>Mathematical</u> <u>Taxonomy</u>, John Wiley and Sons Ltd., London, 1971.
- 10) Marshall, Kim and Romesburg, Charles H., "CLUSTAR and CLUSTID-Programs for Hierarchical Cluster Analysis," <u>The</u> <u>American Statistician</u>, Vol. 34, No. 3, August 1980.
- 11) Ling, Robert F., "On the Theory and Construction of k-Clusters," <u>The Computer Journal</u>, Vol. 15, No. 4, November 1972.
- 12) Ling, Robert F., "A Probability Theory of Cluster Analysis," J. Am. Stat. Assoc., Vol. 68, No. 341, March 1973.

- 13) Suich, Ronald and Derringer, George, "Is the Regression Equation Adequate? -- One Criterion," <u>Technometrics</u>, Vol. 19, 1977.
- 14) Ellerton, Roger, "Is the Regression Equation Adequate? -- A Generalization," <u>Technometrics</u>, Vol. 20, 1978.
- 15) Bard, Yonathan, <u>Nonlinear Parameter Estimation</u>, Academic Press, New York, 1974.
- 16) Young, Hugh D., <u>Statistical Treatment of Experimental</u> <u>Data</u>, McGraw-Hill, New York, 1962.
- 17) Garfinkel, Robert S. and Nemhauser, George L., <u>Integer</u> <u>Programming</u>, John Wiley, New York, 1972.
- 18) Sea, R.G., "Optimal Correlation of Sensor Data with Tracks in Surveillance Systems," <u>Proc. 6th Hawaii Int'l</u> Conf. on Sys. Sci. IEEE, 1973.
- 19) Singer, R.A., and Sea, R.G., "A New Filter for Optimal Tracking in Dense Multi-Target Environments," Proc. 9th Ann. Conf. on Cir. and Sys. Th., 1971.
- 20) Sittler, R.W., "An Optimal Data Association Problem in Surveillance Theory," <u>IEEE Trans. Mil. Elec.</u>, Vol MIL-8, 1964.
- 21) Stein, J.J. and Blackman, S.S. "Generalized Correlation of Multi-Target Track Data," <u>IEEE Trans. Aero. and Elec.</u> <u>Sys</u>., Vol. AES-11, No. 6, 1975.
- 22) Bar-Shalom, Y. and Tse, E., "Tracking in a Cluttered Environment with Probabilistic Data Association," <u>Automatica</u>, Vol. II, 1975.
- 23) Goodman, Irwin R. and Wiener, Howard L. and Willman, Warren W., <u>Naval Ocean-Surveillance Correlation</u> <u>Handbook, 1979, (Draft)</u>, Naval Research Lab, Systems Research Branch, Washington.

